

Lincoln University Digital Thesis

Copyright Statement

The digital copy of this thesis is protected by the Copyright Act 1994 (New Zealand).

This thesis may be consulted by you, provided you comply with the provisions of the Act and the following conditions of use:

- you will use the copy only for the purposes of research or private study
- you will recognise the author's right to be identified as the author of the thesis and due acknowledgement will be made to the author where appropriate
- you will obtain the author's permission before publishing any material from the thesis.

A network analysis of how neonicotinoids have become embedded in New Zealand's agricultural practices

A thesis

submitted in partial fulfilment

of the requirements for the Degree of

Master of Natural Resources Management and Ecological Engineering

at

Lincoln University

by

Thomas Scott

Lincoln University 2020

Abstract of a thesis submitted in partial fulfilment of the requirements for the Degree of Master of Natural Resources Management and Ecological Engineering

A network analysis of how neonicotinoids have become embedded in New Zealand agricultural practices

by

Thomas Scott

This thesis uses tools from within the expansive area of scholarship known as Actor-Network Theory (ANT) to describe the material relations contributing to the on-going use -or discontinuation- of neonicotinoids in NZ's agricultural practices. All though ANT is named a theory, for this thesis, it is better understood as a disparate family of material-semiotic tools, sensibilities and methods. These tools are simultaneously semiotic because they are about relations that carry meaning, and material because they are about the physical stuff caught up in those relations.

Because ANT is so expansive, I have focussed on the specific tools from within its literature that I have applied to the case-study of neonicotinoids. Therefore, I have used Callon's key principles (agnosticism, generalised symmetry and free association) to remain free of *a priori* assumptions, describing actors and controversies as they emerge from the data and to look at the importance of non-humans without bias. I have also used Bruno Latour's CSSF to identify potential interview participants and to analyse this hazardous substance/ agricultural practice controversy. Subsequently, these two tools have enabled me to make sense of how neonicotinoids are embedded -and maintained- in NZ's agricultural practices.

Interview participants were identified by utilising the CSSF as a method for following the actors, subsequently resulting in 16 semi-structured interviews being performed. A thematic analysis was initially used to order the interview data and developed some scope of the case study. This was followed consecutively by an application of the CSSF, which supported an in-depth analysis of the case study and a representation of those material relations between growers, beekeepers, seed and chemical company representatives, international markets, consumers and regulators. Subsequently, the CSSF revealed that neonicotinoids are very much embedded in the agricultural practices of the

arable sector, while they have been simultaneously removed from the agricultural practices of the apples and pears sector. Latour stresses that there is no separation between the context and the issue itself and subsequently, his CSSF brings the scientific, environmental, political and historical contexts into the material relations which the CSSF represents. Therefore, it is these material relations that I have traced and shown to contribute to the on-going use -or discontinuation- of neonicotinoids in NZ. More specifically, this thesis demonstrates how differences in the public representation of neonicotinoids and the presence, or the absence of maximum residue levels and grass grub underwrite the opposing uses of neonicotinoids in NZ.

Thus, by applying the CSSF to a case-study of neonicotinoids, this thesis demonstrates how it might be used to understand agricultural practices. Furthermore, by applying a critical lens to the CSSF, I have critiqued the way I set boundaries in this thesis, the lack of practical direction in the literature, the imposition of theoretical lexicon and the CSSF for what it missed. Subsequently, this research offers guidelines for using the CSSF to identify research participants, follow the actors and make sense of hazardous substance or agricultural practice controversies.

Keywords: Neonicotinoids, Agricultural Practices, Actor-Network Theory, Circulatory System of Scientific Facts, Arable Sector, Apples and Pears Sector, New Zealand.

Acknowledgements

Completing this thesis sees me end a five-and-a-half-year stint at Lincoln University. I began my time here straight out of high school as an undergraduate student studying a Bachelor of Commerce (Agriculture). Both Lincoln staff and students have given me endless opportunities, and I like to think that I have taken full advantage of that. There are several people I would like to express my thanks to for supporting me through this strenuous task of completing an independent master's thesis.

Among these, I would like to thank:

My supervisors at Lincoln, Dr Sarah Edwards and Dr Suzanne Vallance, for their continued support and guidance. Without the pair of them, I think this thesis may have gone on forever, and I would have remained on the journey of wanting to take down big chem.

My supervisor at BOKU, Johann Zaller, from who I learnt a lot about the world of academia during my time in Vienna and who offered up his time to help me produce this document from the other side of the world.

The various individuals who I interviewed, without whom this thesis would have been impossible and who have offered me fantastic networks as I begin my career.

To the administration at both Lincoln and BOKU Universities, this degree has offered me an experience I will always remember, and one which I was lucky enough to enjoy. I thank you for all your help getting me to Austria and looking after me while I was there.

My flatmates who have put up with my constant jabbering about an insecticide for two years or maybe more.

My Mum and Dad, for feeding me when I was too busy to feed myself, for giving me strength throughout my entire education and offering a helping hand whenever I needed it.

Finally, my girlfriend Annabel for your love, passion and encouragement throughout this thesis. You made it a whole lot easier and allowed me the time and space to get work done when I needed it.

Table of Contents

Abstractii				
Acknowledgementsiv				
Table of Contentsv				
List of	f Figuresvii			
Acror	nymsvii			
•	ter 1 Introduction1			
1.1	Background and significance1			
1.2	Personal motivation2			
1.3	Methodology3			
1.4	Research objectives4			
1.5	Outline and structure4			
Chapt	ter 2 Background			
2.1	Introduction			
2.2	Agriculture in New Zealand			
2.3	Agrichemical industry			
2.4	Pollinators and the apiculture industry			
2.5	Hazardous substance regulation in New Zealand12			
2.6	Neonicotinoids in Europe			
2.7	Neonicotinoids in New Zealand			
2.8	Summary16			
Cham	ter 3 Literature Review			
3.1	Introduction			
3.2	Key actor-network theory principles			
3.2	Actor-networks and network representations			
3.3 3.4	The circulatory system of scientific facts			
5.4	3.4.1 Agricultural practices and case study applications of the CSSF			
	3.4.2 The CSSF as a research methodology Error! Bookmark not defined.			
3.5	Limitations of the key principles of ANT and the CSSF			
3.6	Summary34			
-	ter 4 Methodology			
4.1	Introduction			
4.2	Case study			
	4.2.1 Case study methods			
4.3	Organising the data			
4.4	Summary			
Chapt	ter 5 Results			

5.1	. Introduction				
5.2	Emergent themes				
	5.2.1	Theme 1: New Zealand is a small market for agricultural chemicals	51		
	5.2.2	Theme 2: Not enough tools in the toolbox	53		
	5.2.3	Theme 3: Market pressure			
	5.2.4	Theme 4: Disconnect between landowners and agricultural practices			
	5.2.5	Theme 5: Grass Grub			
	5.2.6	Theme 6: Relationship between beekeepers and landowners	70		
5.3	Summa	ary	73		
Chap	ter 6 Dis	cussion	75		
6.1	Introdu	lction	75		
6.2	The arable sector, the stable network76				
6.3	The ap	The apples and pears sector, the destabilized network			
6.4	What h	as the CSSF taught us about agricultural practices	95		
6.5	Overco	ming the limitations of the key principles of ANT and the CSSF	97		
	6.5.1	Responding to general critiques of ANT			
	6.5.2	Avoiding the objective and un-reflexive approach when following the actor			
	6.5.3	Inclusion and exclusion of actors	101		
6.6	Summa	ary	103		
Chap	ter 7 Coi	nclusion			
7.1	Summa	ary of research findings			
7.2	Contrib	outions of this study			
Refer	ences		110		
Δnne	ndix		132		
A.1		ew Schedules			
A.2		ch Information Sheet			
A.3		University Human Ethics Committee Approval			
A.4		o participants			
A.5	Consen	it Form	137		

List of Figures

Figure 1 The circulatory system of scientific facts, sourced from (Latour, 1999a)......24

Acronyms

ACVM	Agricultural Compounds and the Veterinary Medicines Act
ANT	Actor-Network Theory
CRI	Crown Research Institute
CSSF	Circulatory System of Scientific Facts
ERMA	Environmental Risk Management Authority
EU	European Union
FAO	Food and Agricultural Organization of the United Nations
HORTNZ	Horticulture New Zealand
HSNO	Hazardous Substances and New Organisms Act
MFE	Ministry for the Environment
MPI	Ministry for Primary Industries
MRL	Maximum Residue Level
NZEPA	New Zealand Environmental Protection Authority
NZ	New Zealand
OP	Organophosphate
p.a.	Per annum
STS	Science and Technology Studies

Chapter 1

Introduction

1.1 Background and significance

This thesis explores neonicotinoid use in New Zealand (NZ). More specifically, it explores how this class of insecticides has become - and remains - embedded in NZ's agricultural practices. Neonicotinoids are the most widely used insecticide in the world (Cressey, 2017). They resemble nicotine, and are systemic, meaning that they are taken up by the roots or leaves of a plant and translocated to all other plant parts (Clarke, 2018). Subsequently, global agricultural and chemical communities consider them to be highly effective and environmentally friendly (AGCARM, 2019; United States Environmental Protection Agency, 2019). Recently neonicotinoids have been gaining traction in the media and drawing the attention of many governments due to the potential harm they cause pollinators (Clarke, 2018). Subsequently, they are becoming restricted, even banned around the globe (Sanchez-Bayo, 2014; Magallanes, 2018). Nonetheless, NZ legislators are not following this trend and have left neonicotinoids out of their hazardous substance reassessment programme (Ardern et al., 2014; Clarke, 2018). Therefore, the primary objective of this research is to understand how neonicotinoids have become embedded -and maintained- in NZ's agricultural practices.

On the one hand, advocates state that neonicotinoids lead to economic gain, are a necessary treatment especially at the seedling stage of crop/pasture development, are highly targeted and environmentally friendly, do not harm pollinators, are already strictly regulated and ensure people have access to affordable and reliable food (AGCARM, 2013; Carrington, 2017; Chapman & Jackson, 2010; Kleinschmit & Lilliston, 2015; McSweeny, 2017; Piddock, 2018). On the other hand, critics state that neonicotinoids have adverse effects on human health, ecosystems and the environment, they cause pollinator decline and the regulation surrounding their use is outdated (Budge et al., 2015; Carrington, 2017; Hobbs, 2002; Kleinschmit & Lilliston, 2015; Ministry for the Environment, 2018; Walters, 2013). Due to the potential impacts which neonicotinoids have on food production and the environment, whether these be positive or negative, it has become a widespread debate. An important argument against the use of neonicotinoids is around the potential harm they cause honeybees and other beneficial insect populations (Ardern, 2014). In NZ, this is of great concern because apiculture contributes \$5.1 billion to the economy per annum (p.a.) (Ministry for Primary Industries, 2017). This contribution is primarily associated with the benefits provided by pollination services. Additionally, a new Manuka honey industry has emerged, leading to the highest

national beehive population on record. Manuka honey exports are valued at over \$329 million p.a. Subsequently, it has attracted a lot of media attention (Ministry for Primary Industries, 2017).

Bruno Latour's Circulatory System of Scientific Facts (CSSF) (1999a, pp. 98-108) has been applied to this research to facilitate an exploration of how neonicotinoids are embedded -and maintained- in NZ's agricultural practices. My research will also explore how the CSSF, a network representation, might facilitate an investigation into the relationships and interactions that surround the use of neonicotinoids in NZ. By using the CSSF, my research will show how relationships have formed and how neonicotinoids are being framed. Furthermore, this thesis explains why research opposing neonicotinoid use has been unable to remove them from NZ's agricultural practices. Subsequently, this case study has also enabled me to explore the potential of the CSSF as a suitable framework for investigating hazardous chemical controversies in the future.

1.2 Personal motivation

Both my academic and professional experiences motivated this research. Most importantly, my position at Springbank honey in Cust, NZ, drove my interest in neonicotinoids. After beginning my employment there, to support myself during my studies, I quickly became aware of several studies demonstrating the negative impacts which neonicotinoids cause honeybees. My personal experience of observing beehive deaths caused by the unlawful use of agricultural compounds (Environmental Protection Authority, 2019) intensified this interest. During the completion of this master's degree, I was lucky enough to spend some time in Austria, where I noted the ongoing European neonicotinoid debate. I was living in Austria at the time when three types of neonicotinoids were banned and therefore observed the significant bottom-up movement which drove this. Subsequently, I could not understand how neonicotinoids were still being used so extensively in NZ.

I had previously used tools from within ANT to try and understand why honeybee sites were being established on public conservation land in NZ despite a Department of Conservation recommendation not to do so (Beard, 2015). Having found some success performing this research, I decided to once again use tools from within ANT for my master's thesis. Having completed a Bachelor of Commerce (Agriculture) and having spent many years working on farms across Canterbury NZ, I have developed an interest in agricultural practices and how they are both implemented and chosen. Therefore, the main objective of this research, to understand how neonicotinoids have become embedded -and maintained- in NZ's agricultural practices, appeared to be a good fit. Initially, I wanted to know why neonicotinoids were still used in NZ. I was misled in thinking that there was a hidden corporation making sure it stayed around. Fortunately, ANT is a constructivist approach to social theory (Tonelli, Silva et al., 2010), it asks how something is made rather than why something has happened. Furthermore, Latour's CSSF (1999a) asks how science is done. So rather than assuming neonicotinoids are embedded -and maintained- because of the actions of a particularly powerful actor, adopting a CSSF approach made me re-evaluate how they remained embedded in agricultural practice. The reader needs to keep these personal motivations in mind as they make their way through this thesis. Admittedly there are many possibilities for me to have biases, considering my previous work, employment and the fact that I have beehives. There is not much more I can do to alleviate these biases, but I hope that by being self-reflexive and acknowledging this potential for bias at the outset I am allowing the reader to draw their own informed conclusions from my research.

1.3 Methodology

The methodology of this thesis focused on the analysis of in-depth interviews with individuals from across NZ's agricultural community, a document analysis of relevant publications and reflections of my own practices. The interview participants included landowners, beekeepers, seed and chemical company reps, product group reps, regulators and researchers, while the documents primarily consisted of journal articles, regulatory material, newspaper articles, conference papers, press releases and web pages. Finding suitable interview participants and the relevant documents proved quite challenging, due to there being some secrecy around hazardous substance supply chains in NZ. Furthermore, most of the NZ media coverage referenced the same spokespeople when discussing neonicotinoids. I utilised snowball sampling, Latour's CSSF (1999a), Callon's key principles (1986) and some previous CSSF case studies at every stage of this research. It was these theoretical resources and methodologies that enabled me to collect data, be selective about what data were relevant and to then analyse it and discuss all the pertinent phenomenon.

The CSSF (1999a) is a framework used during the data collection and analysis stages of this research and it has informed agricultural practice studies before. For example, Ingram (2007) used the CSSF to highlight how the participation of scientists and the pursuit of scientific research developed several agricultural technologies. More specifically, she used the CSSF to follow the making and maintenance of three separate agricultural methods for managing soil. Fitzsimmons (2004) builds on the work of Latour to demonstrate how ecologists have vascularised their work throughout American agricultural and political communities. She explained how "ecologists made a structure of engagements for translating knowledge into practical action and political power". Warner (2008; 2016) also used Latour's tool to understand science in general, but mainly those controversies stemming from the work of Rachel Carson's Silent Spring. Latour's CSSF played a significant role in directing me where to look for interview participants. I used his loops as a way to find relevant actors and their peers. Later on, I used the CSSF to represent the data in such a way that I could highlight phenomena such as agricultural practices relational and material nature, as well as answer those research questions presented below. I finally went on to critique the CSSF, attempting to reflect on my practices, by acknowledging the boundaries and limitations of both myself and this research.

1.4 Research objectives

The main aim of my study is to explore how neonicotinoids have become – and remain - embedded in NZ's agricultural practices. To do this, I have explored how relationships across NZ's agricultural, scientific, political and environmental communities have developed. I have also explained how the continued use of neonicotinoids is an outcome of these various relationships. Furthermore, I have discussed why the scientific knowledge that challenges the status quo, like the non-target effects of neonicotinoids, has been unsuccessful in removing them from NZ's agricultural practices. I have also been able to assess the suitability of specific tools taken from ANT literature for investigating hazardous chemical controversies in the future. Finally, I explored the idea that agri-environmental controversies are never just scientific, that policy, business and society all have a significant role to play. To achieve these objectives, I have answered the following research questions:

- 1. Who and what are the major actors that influence the scale and frequency of neonicotinoid use in NZ, and what have been their roles?
- 2. How have certain relationships been made and maintained between actors?
- 3. How are neonicotinoids as an agricultural practice embedded -and maintained- in NZ, and why has opposing knowledge been unsuccessful in removing it?
- 4. How can Latour's CSSF be used to investigate similar controversies in the future, in particular, those which involve hazardous chemicals used in NZ's agriculture?

1.5 Outline and structure

This thesis is organised into seven chapters. Chapter 1 has provided a broad overview of the controversy surrounding neonicotinoids, both on a global and domestic scale. It also provides this study's terms of reference. Chapter 2 provides a detailed account of the background information which is relevant to this case study. This includes a background investigation into the policy controlling agrichemicals, the key industry stakeholders (apiculture, agriculture and agrichemical), the international positions on neonicotinoids as well as the domestic ones. Chapter 3 provides a detailed review of Michel Callon's key ANT principles (1986) and Bruno Latour's CSSF (1999a), both scrutinising them through a literature review and considering their relevance to this case study. Chapter 4 details the specific methodology utilised during the data collection and analysis, which is followed by Chapter 5 that outlines the key findings from a series of in-depth interviews in a thematic format. Chapter 6 produces a review of the results while also discussing

some issues that occurred during the research process. Finally, Chapter 7 concludes the research by explaining what it has contributed to CSSF and hazardous substances literature.

Chapter 2

Background

2.1 Introduction

Before I can undertake an analysis of neonicotinoid use in NZ using the CSSF (1999a), it is necessary to consider the background information relevant to this case study.

Firstly, this chapter provides background information on who and what appears to be involved with neonicotinoids in NZ. These include regulators and the NZ agricultural, agrichemical and apicultural industries. There is a need to evaluate the relationships between these groups to answer the first and second research questions of this thesis.

Secondly, I will consider the role of policy in controlling agrichemical use in NZ, particularly the Hazardous Substances and New Organisms Act (HSNO) (Ministry for the Environment, 1996) and the decision-making body named the NZ Environmental Protection Authority (NZEPA). This provides some of the background information necessary for understanding the relationships between those actors which are contributing to the embedding -and maintenance- of neonicotinoids in NZ's agricultural practices.

Finally, I will discuss the changing global perspectives on neonicotinoids, which will focus on the European Union (EU) and their extensive bans on this class of insecticide (McGrath, 2014). This information is relevant to answering the third research question and understanding why knowledge opposing the use of neonicotinoids has been unsuccessful in removing it from NZ's agricultural practices.

This chapter will, therefore, demonstrate that neonicotinoids have become embedded -and maintained- in NZ's agricultural practices.

2.2 Agriculture in New Zealand

NZ split from the supercontinent Gondwana around 85 million years ago. Because of its isolation, NZ's flora and fauna have evolved unique characteristics and traits. This not only led to a lack of predatory land-dwelling species but has helped mould NZ's agricultural landscapes and practices into what they are today. The primary reason is that some endemic insect species adapted well to the modification of landscapes caused by human activities. This subsequently led to them thriving in those landscapes and now being classified as agricultural pests (Brockie, 2007; McLintock, 1966). Since colonisation, agriculture has been the primary source of income for NZ, and there has also only been a brief period in the 1860s when agriculture did not lead NZ's export earnings (Peden, 2008; Tourism Industry Association New Zealand, 2019). There has always been large scale economic activity occurring in NZ. However, the first settlements of Māori did not practise commercial market transactions but engaged in gift-giving and barter. They lived in subsistent, self-sufficient units known as hāpu and effectively utilised what the land provided them (Haggerty and Cambell, 2008; Easton, 2010). Cultural and ideological changes meant that Māori became expert resource managers and gardeners (The Reserve Bank of New Zealand, 2007; Easton 2010) and their diet consisted of birds, seafood and gathered fern-root, as well as cultivated crops which had been imported by their Pacific ancestors (Kingi, 2008).

In 1769, two European explorers, James Cook and Jean François Marie de Surville both came to NZ. They subsequently traded cabbages, turnips, potatoes, wheat, rice, peas and pigs with the Māori tribes of Tolaga and Doubtless Bays (Kingi, 2008, 2019). It did not take long for Māori to begin trading these European crops with one another, with some records suggesting that this occurred from as early as 1803 (Kingi, 2008).

Before the establishment of an organised European settlement in NZ, there was a heavy reliance on those resources that occurred naturally. This period saw the rise of export industries reliant on seals, whales, timber and flax (Russell Stephens, 1966; Peden, 2008). After many years and the development of an organised British settlement, the industrial revolution occurred. Subsequently, large scale agricultural expansion, the replacement of sail with steam and the invention of refrigeration allowed NZ to dive into the international market (Russell Stephens, 1966).

Since the 1850s, NZ's economic backbone mostly revolved around pastoral grazing and the subsequent commodities of wool, meat and dairy (Kingi, 2019). The great pastoral era began when the first flock of Merino sheep arrived in NZ and, within 30 years, there were over 3 million. This growth continued, and by 1901 there were over 20 million Merino sheep in NZ. The vast area of 'free' land available in the South Island, which was ideal for fine-wooled sheep breeds, helped this rapid expansion. This growth subsequently led to the development of a frozen meat export industry (Russell Stephens, 1966; Peden, 2008). The tussock lands of the South Island quickly required pasture renewal which allowed for vast areas of land to be ploughed and planted with wheat crops. Thus, by the 1880s, this wheat crop made up 20% of the NZ's income (Peden, 2008). At the same time, the dairy industry began in NZ with the first factory opening in 1881, and by 1890 there was within the region of 150 butter and cheese factories in the country (Peden, 2008). Through the early 20th century pastoral based agriculture remained the biggest earner in NZ and in 1920 it contributed 91% of NZ's export earnings.

The capacity of NZ's agricultural land was reached around this time, and the large-scale use of superphosphate help facilitate continued growth (McLintock, 2019). NZ gained the title of Britain's distant stock farm following WWI, and several scientists arrived from Britain to boost the productivity of NZ's agriculture. Subsequently, this initiated the founding of the Department of Scientific and Industrial Research and Massey Agricultural College. Lincoln Agricultural College was also upgraded (Brooking, 2006). However, farms struggled to survive between the wars, and many returned to native bush, which was exacerbated by the great depression and the collapse of wartime prices (Brooking, 2006).

NZ entered its golden age of farming following WWII. There was a record number of farms, and the nation's sheep flock reached 70 million, while the cattle herd reached 8 million (Brooking, 2006). Over the same period, cereal crops remained an essential part of NZ's agriculture, and a new agricultural commodity in the form of vegetables for processing commenced in the 1980s. When Britain joined the EU, NZ's agricultural production exceeded demand, and the country faced a severe fiscal crisis. Subsequently, the NZ government implemented a serious deregulation programme which saw the removal of farm subsidies and the end of NZ's agricultural golden age (Ross, 2019).

To remain competitive in the highly subsidised markets of the EU, NZ's farmers had to diversify their production and improve the efficiency of their operations. This move towards a value-added product space, in hand with greater access to irrigation and the opening of new markets in Asia, Europe and the Middle East, resulted in a dairy boom (Brooking, 2006; Peden, 2008). Since the 1980s, there has also been a spike in NZ's horticultural production. Apple export earnings grew from \$37.75 million in 1980 to over \$800 million in 1995. Kiwifruit exports increased from \$34.5 million in 1980 to over \$540 million by 1990 and total export earnings from wine were \$662.4 million in 2007 (Peden, 2008). As of 2016, 45.3% of NZ's total land area is being farmed. Between 2002 and 2016, total land area under dairy increased from 1.8 to 2.6 million hectares and subsequently, land area under sheep and beef decreased from 10.7 to 8.5 million hectares. Furthermore, the land area under grain crops increased by 101.4% over the same period (Stats NZ, 2018).

This section demonstrates that NZ has a heavy reliance on horticultural and agricultural production (McLintock, 2019). This is because of its economic benefits and the connection which much of the population has with the land. Therefore, many difficulties arise when agricultural and horticultural productivity is threatened. Nevertheless, social, political and environmental movements have had significant effects on NZ's agricultural practices. DDT was once heavily used in NZ. It was mixed with fertiliser to control endemic pasture pests. In 1989 this harmful chemical was banned in NZ on the back of Rachel Carson's Silent Spring (Wildblood-Crawford, 2006). However, as some things change, some do not. Neonicotinoids, a class of agricultural pesticides subject to substantial bans

internationally (see section 2.6), are still widely used in NZ (see section 2.7). NZ's farmers rely on neonicotinoids to protect their crops from endemic agricultural pest species at the seedling stage. Unlike DDT, there does not appear to be substantial movements backing a neonicotinoid ban (Morton, 2018; Twaddle and Sanderson, 2014).

2.3 Agrichemical industry

The agrichemical industry revolves around the importation, production and supply of those chemicals which improve agricultural production (Ministry for the Environment, 2019). Agrichemicals are managed in NZ under three acts of parliament; the Hazardous Substances and New Organisms Act (HSNO Act), the Ozone Layer Protection Act and the Agricultural Compounds and the Veterinary Medicines Act (ACVM Act) (Ministry for the Environment, 2019). There is a small number of key stakeholders in this industry, including six international brand owners that either operate within NZ or hold secured relationships with NZ based parties. They usually have teams in NZ that focus on building relationships with farmers, specialist farm retailers and service providers (Ministry for the Environment, 2019). Even more recently, these six major agrichemical companies have formed three entities. These entities now jointly control 70% of the global agrichemical and 60% of the global seed market (Crowe Horwath, 2017; Ministry for the Environment, 2019).

As well as the brand owners there are generic suppliers who have a physical company presence in NZ, source their products via imports and represent around 30% of the market. A generic agrichemical is one whose patent has run out and is now being manufactured, sold and marketed under various brand names, by multiple companies (Ministry for the Environment, 2019). The ability of these companies to produce and retail the same generic chemical under different brand names means that they are competing with one another on price alone, rather than chemical efficacy and ecotoxicity. Furthermore, modern telecommunication and social networks have enabled their sales teams, to interact frequently and directly with farmers. Farmers in NZ depend on agrichemicals for the success of their businesses, and they hold close relationships with their supplier. Typically, farmers only keep enough product on hand to maintain their farm for one year; this is likely a consequence of changing brands and the frequent release of newly improved formulations (Fitzpatrick, 2019). The combination of price competition and direct supplier to end-user relationships has meant that the generic supplier proportion of the industry is rapidly growing (Ministry for the Environment, 2019). Because of this growth, there are now anywhere between 15 and 20 generic chemical distributors in NZ who directly compete with research-based patented brands (Ministry for the Environment, 2019).

Because a number of widely used chemicals have recently come off patent, there has been a jump in the number of these generic brand options available, resulting in lower-priced compounds across NZ,

leading to higher application rates and less care taken when the applications occur (Manktelow, Stevens et al., 2005). Subsequently, agrichemical resistances have been observed across NZ (Manktelow, Stevens et al., 2005), resulting in various changes to the agricultural industry and its community. Farmers now have an increased reliance on specialist agrichemicals. There is also more public pressure to minimise application rates and several integrated pest management strategies have been developed (Martin and Workman, 1994; Science Media Centre, 2015; Bourdôt, Buddenhagen et al., 2019; Ministry for the Environment, 2019).

The Ministry for the Environment (MFE) has "not been able to identify information, collected and verified centrally, on which sectors in agriculture use agrichemicals and in what proportion" (Ministry for the Environment, 2019). Furthermore, the Environmental Protection Authority (EPA) has stated that "we really don't know what's out there" (Hancock, 2018). There are also no records of consistent data collection across the whole sector (Agrichemical Trespass Ministerial Advisory Committee, 2002; Ministry for the Environment, 2019). Because the agrichemical industry does not reveal any market share information (Ministry for the Environment, 2019), there is no way of determining the rates at which agricultural chemicals are used in NZ. This section outlines those existing relationships between the agrichemical sector, farmers, regulators, researchers, the media and more. It also indicates that further investigation of several documents and interview data will contribute to an understanding of the roles of and relationships between these individuals and organisations, and how they have contributed to the scale and frequency of neonicotinoid use in NZ. It will also help us understand how those relationships formed in the first place, and therefore suggest how neonicotinoids have been maintained and not removed from NZ's agricultural practices.

2.4 Pollinators and the apiculture industry

Pollination by animals is a crucial ecosystem service. It underpins New Zealand's agriculture-dependent economy yet has hitherto received little attention from a commercial perspective except where pollination clearly limits crop yield. (Newstrom-Lloyd, 2013)

Prompted by the works of those like Rachel Carson and a growing environmental movement, the use of pesticides and herbicides is more carefully considered by bodies like the NZEPA. Along with impacts on human health, the effects of agrichemicals on other species is also considered. The effect on pollinators is particularly important as according to the Food and Agricultural Organization of the United Nations (FAO), 35% of the world's agricultural land, which supports 87 of humanities leading food crops, are reliant on some form of animal pollination (Food and Agricultural Organization of the United Nations and Republic of Slovenia Ministry of Agriculture Forestry and Food, 2018). Because pollination in NZ has historically been performed by unmanaged *Apis mellifera* hives and non-*Apis* pollinators, there has not been much research into the contribution of pollinators to NZ's economic,

social and environmental well-being (Bennett et al., 2018). Nevertheless, NZ's managed honeybee hives do provide pollination for a wide range of crops and the use of bumblebees has been gaining traction.

Furthermore, NZ has several solitary bee species and other animal pollinators which all contribute to the pollination of crops, native and non-native flora. However, NZ is yet to harness the capabilities of and manage native pollinators, so it continues to rely heavily on non-natives for pollination services (Newstrom-Lloyd, 2013). This reliance makes NZ's pollination susceptible to risks like pests, disease, pesticides and declining floral resources (Newstrom-Lloyd, 2013). All of which are known to contribute to the endangerment of vertebrate and invertebrate pollinator populations (The White House Office of the Press Secretary, 2014). The accidental importation of honeybee queens carrying the parasitic mite, Varroa destructor, led to the death of thousands of honeybee hives over the last two decades and the ensuing rise of apiary operational costs. This caused the number of beekeeping enterprises to drop by 50% in 8 years and the destruction of all NZ's feral hives (Ministry for Primary Industries, 2017). Following the introduction of Varroa, those beekeepers who remained were rewarded with a price jump and an increase in demand for Manuka honey. The nectar produced by the Manuka (Leptospermum scoparium) flower has gained a reputation as having healing properties, fetching prices of up to \$300/kg (Beard, 2015). Subsequently, there is growing competition for bee sites with access to native bush and the value of NZ's honey exports reached \$329 million in 2016/17 (Ministry for Primary Industries, 2017). So, despite the pressure of Varroa, NZ's apiculture industry still contributes over \$5.1 billion to the economy p.a. (Ministry for Primary Industries, 2017).

Declining biodiversity of pollinators is linked to the intensity of agricultural practices and neonicotinoids have been ascribed as one of the leading causes. Habitat loss, parasites, disease, adverse weather and poor nutrition also contribute (Godfray et al., 2014; Beston, 2017). Subsequently, there is growing concern around the impacts which neonicotinoids have on hives. This begs the question, how does the relationship between NZ's beekeepers and landowners function, particularly around the use of agricultural chemicals, like neonicotinoids, and the potential harm they cause beekeepers' hives (Ardern et al., 2014).

This summary shows the importance of pollinators to the environment, people and the economy. It also demonstrates that it would be difficult to feed the world without pollinators (Klein et al., 2007). Representatives of pollinators frame neonicotinoids as being detrimental to not just humanity but all life on earth. Likely, neonicotinoid use in NZ would not have become controversial unless for the sequence of events involving the fall and rise of NZ's honeybee population. Beekeepers are probably the first individuals to observe the misuse of neonicotinoids and any of their adverse effects

(Woodcock et al., 2017). Which raises interesting questions about the relationship between farmers and beekeepers and the role they play in embedding neonicotinoids in NZ.

2.5 Hazardous substance regulation in New Zealand

As stated in section 2.3 of this chapter, pesticides, like neonicotinoids, are managed under 3 acts of parliament (Ministry for the Environment, 2019). For this thesis, I will only go into to detail about the HSNO act as it is the primary statute for assessing and approving pesticide use in NZ. I will also outline the function of the NZEPA as the decision-making body under the HSNO Act (Lorns, 2019).

In the early 20th century regulation of hazardous substances was created through market availability (Ministry for the Environment, 2002) and in 1959, formal law was established (Agricultural Chemicals Act, 1959). The Agricultural Chemicals Board was also developed and included users and producers of chemicals appointed by the Minister of Agriculture (General Assembly of New Zealand, 1959; Magallanes, 2018). Its primary function was to ensure the wellbeing of agriculture. Compulsory registration of agrichemicals was imposed and required an application showing the effectiveness of a chemical at eliminating pests (General Assembly of New Zealand, 1959; Mann, 1972; Magallanes, 2018). This raised questions about the extent to which people and the environment were considered in applications (MacIntyre, 1989; Magallanes, 2018). Which, together with public pressure, issues in the decision-making process and a lack of understanding about the effects of pesticides led to the act and board being reviewed (MacIntyre, 1989; Magallanes, 2018). What resulted was the formation of the Pesticides Act (1979) and the Pesticides Board, which focussed more on public safety and the environment. More power was given to annul registrations, and the board included a nominee of the Minister for the Environment. For the first time, pesticide registrations could be refused because of environmental risk (General Assembly of New Zealand, 1979).

Once again, public pressure and issues with the decision-making process of the Pesticides Board led to the establishment of a new Act during the 1990s. This was the HSNO Act which focused more on risk reduction, and it remains the primary statute for approving and regulating pesticides in NZ (Ministry for the Environment, 1996; Harris et al., 2004). Its purpose is to "protect the environment, and the health and safety of people and communities, by preventing or managing the adverse effects of hazardous substances and new organisms" (Section 4). Unlike previous regulations, the HSNO Act assesses hazardous substances based on their risk characteristics and allows for public participation (Section 2).

The Environmental Risk Management Authority (ERMA) was also established during the 1990s to make decisions under the HSNO Act. In 2011 another reform meant that ERMA was replaced with the NZEPA (Ministry for the Environment, 2011, Section 7 (1) (a) and 25). Unlike the Pesticides Board,

there is not a requirement for a member of the NZEPA's decision-making board to represent environmental interests (Magallanes, 2018). When deciding on a pesticide submission, the NZEPA must consider various principles (Ministry for the Environment, 1996) which safeguard the lifesupporting capacity of air, water, soil and ecosystems. They also maintain and enhance the ability of people and communities to provide their own economic social and cultural well-being (Section 5). Generally, the NZEPA approve hazardous substances if their benefits outweigh any adverse effects and vice versa (Section 29 (1) (c)). However, the NZEPA maintains total discretion over the approval of hazardous substances as long as they consider all the right factors (Section 29 (3)). Another significant regulatory body is the Ministry of Primary Industries (MPI) which holds some responsibility for enforcing the decisions made by the NZEPA. MPI began in 2012, and all the members of the NZEPA board are appointed on the recommendation of the Minister (Ministry for Primary Industries, 2019).

Following bans on three classes of neonicotinoids in Europe during 2018 (section 2.6) (Piddock, 2018), the NZEPA declared that "when new information is released, the EPA always takes a good look at the science, evaluating it to see if there is something we need to factor into our thinking here" (Environmental Protection Authority, 2018). In 2018, the NZEPA also declared to ramp up its chemical reassessment programme (Clarke, 2018). The reassessment programme was "designed to lay the foundations for a modern chemical management system; one supported by robust and up-todate evidence and data, and which aligns with the standards, knowledge and practices recognised by our regulatory partners globally" (Environmental Protection Authority, 2018). However, neonicotinoids, along with many other controversial chemicals, were not considered for reassessment (Cressey, 2017). Therefore, under the HSNO Act, the banning of a pesticide by other countries is not grounds for reassessment (Magallanes, 2018). The NZEPA stated that neonicotinoid's exclusion was because their hazard profile and their risk to human and environmental health does not meet priority criteria (Hutching, 2018). They followed this up by saying that the use of neonicotinoids in NZ is already heavily regulated. Interestingly, the NZEPA has recently called for information on neonicotinoids to understand how they are being used in NZ (Environmental Protection Authority, 2018). This description of the HSNO decision-making process, as well as the role of the NZEPA, raises several questions about; how the NZEPA gain data for hazardous substance applications, whether they are considering the adverse effects of neonicotinoids, what the benefits are which outweigh those adverse effects and who holds relationships with the NZEPA that benefit from the use of neonicotinoids.

2.6 Neonicotinoids in Europe

Following the first application of Gaucho (a neonicotinoid) in France, some of the local beekeepers noticed that their hives were dwindling. In response, the beekeepers approached the producer of Gaucho, (Bayer CropScience) and demanded a thorough investigation into whether this neonicotinoid affected honeybees. Several studies showed no adverse effects, but others carried out by the French government and independent researchers supported the beekeepers' claims. Subsequently, in 1999 the French government imposed a ban on Gaucho (Sanchez-Bayo, 2014; Magallanes, 2018). Following this, Europe has become increasingly concerned about the effects neonicotinoids have on pollinators. So, the European Food Safety Authority reviewed neonicotinoids, and in 2013 the EU imposed partial restrictions on three classes of neonicotinoid (AGCARM, 2013; Kothage, 2018), all of which are widely used in NZ.

These restrictions have sparked a debate over whether the benefits provided by neonicotinoids outweigh their detrimental effects (Woodcock et al., 2017). A meta-analysis published in the Proceedings of the Royal Society aimed to summarise all the natural science evidence relevant to neonicotinoids and insect pollinators (Godfray et al., 2014). It concluded that there is limited information to guide policymakers (Sanchez-Bayo, 2014) and that under current management strategies, neonicotinoids are toxic to insect pollinators. There are several ways pollinators can be exposed to neonicotinoids, including plant exudates, dust from machinery, foliar sprays and contamination of soil and water (Godfray et al., 2014). Estimates of neonicotinoid traces in pollen and nectar vary, but the dust emitted from drilling machines contains high levels, and its drift has been linked to events of mass honeybee death (Bonmatin, 2015). Neonicotinoids can persist in the environment for 15-300 days, and concentrations accumulate when treated crops are grown repeatedly. They have also been observed in weeds and untreated crops grown in rotation (Godfray et al., 2014). Because of these reasons, in 2018 the EU voted to extend the restrictions already in place by altogether banning the use of 3 neonicotinoids in open fields (Greenpeace European Unit, 2018; Environmental Protection Authority, 2019b). To comprehend how neonicotinoids have been made and maintained as an agricultural practice in NZ, I must understand how this became a controversy in the first place. The above description of Europe's perspective and the development of neonicotinoids as a globally recognised debate will support the discussion of and contrast to NZ's position on neonicotinoids.

2.7 Neonicotinoids in New Zealand

Neonicotinoids have been popular since their introduction to NZ in 1992 (Mellet, 2012). They are intended for arable crops, forage brassicas and pasture grasses and as of 2011, made up 28.5% of the global insecticide market. Which is primarily due to their correlation with increasing crop yields

(Godfray et al., 2014). Neonicotinoids are known for their ability to control insects while having low toxicity to mammals (AGCARM, 2013). There are several ways neonicotinoids can be applied, including sprays, drippers, granules and pour-on (Godfray et al., 2014). However, typically, the application is via a seed treatment which allows the insecticide to be taken up in the growing plant and be present in all the plant parts, including the nectar and pollen (Simon-Delso, 2015).

It is widely publicised that neonicotinoids are an environmentally friendly crop protection method that without, certain crops would be impossible to grow in NZ (AGCARM, 2013). The removal of seed treatment varieties of neonicotinoids would likely cost the NZ economy between \$800 million and \$1.2 billion (AGCARM, 2018), affect the employment of 5,300 full-time workers and reduce national crop yields by one third (AGCARM, 2018). This is because 62% of seeds planted in NZ are treated with some form of neonicotinoid, and the treatments alone contribute an additional \$368 million to the market value of those seeds (AGCARM, 2019b). This suggests that the economic benefits of neonicotinoids are helping to maintain them in NZ's agricultural practices. Subsequently, this opens a discussion into who is gaining those benefits. It also raises questions about how those beneficiaries are convincing the NZEPA that the benefits of neonicotinoids outweigh their detrimental effects.

Due to growing concern about the non-target effects of neonicotinoids, there have been several public calls to ban them (Clarke, 2018), but changes have not occurred. However, some retailers have decided to stop selling neonicotinoid products (Didovich, 2018), and the NZEPA has agreed to "keep an eye" on neonicotinoids (AGCARM, 2013). The NZEPA will not make regulatory changes despite their publications acknowledging neonicotinoid toxicity (Ardern et al., 2014; Clarke, 2018; Environmental Protection Authority, 2018b, 2019). The NZEPA and MPI, ensure that neonicotinoids are used in such a way that residues are minimised, and they also acknowledge a lack of NZ based studies supporting the adverse effects of neonicotinoids (Environmental Protection Authority, 2018b). The NZEPA maintains that NZ pollinators are not at risk from neonicotinoids because of tight regulations surrounding their use (Environmental Protection Authority, 2018b).

Furthermore, the Foundation for Arable Research (FAR) has stated that the risk to bees is not as significant in NZ because we do not tend to use neonicotinoids as a foliar spray (Piddock, 2018). The NZEPA also states that international attention on the effects of neonicotinoids mainly relates to colony collapse disorder (CCD), a phenomenon not recorded in NZ (AGCARM, 2013; Ardern et al., 2014). Although it is suggested that even in low concentrations neonicotinoids harm bees, the NZEPA maintains that NZ beehive problems are caused by Varroa mites (Magallanes, 2018). Nevertheless, the NZ House of Representatives has voiced concerns that the NZEPA is not assessing chemicals appropriately and is not monitoring volumes of pesticides imported or used in NZ (Ardern et al., 2014). Another concern is that the NZEPA's focus is on the use of neonicotinoids as a foliar spray,

even when most international interest is around pollinator exposure through the systemic effects of seed coatings. What this means is that seed treatment neonicotinoid applications are not controlled by any rules (Clarke, 2018). A final criticism is that the NZEPA ignores the unpredictable damage caused by neonicotinoids at population levels, the tipping points which then lead to the collapse of pollinator populations are only observable after they occur (Magallanes, 2018).

This section raises questions about the NZEPA's justification for not reassessing neonicotinoids. Is the NZEPA not considering the research which demonstrates the adverse effects of seed treatments or is there some difference between NZ's use of seed treatments and the rest of the world's?

2.8 Summary

This chapter outlined the background information needed for this thesis and provided a knowledge base for understanding Chapters 5 and 6. Thee descriptions of NZ agriculture, NZ's agrichemical industry and NZ's apicultural industry, present in this chapter, have provided me with some insight into how a point has been reached where neonicotinoids are a necessary agricultural practice in NZ. Furthermore, this chapter has provided an overview of the regulating bodies and statutes that manage neonicotinoids in NZ. Lastly, I have outlined the banning of neonicotinoids in Europe which highlights that agricultural practices are not universal. I have also described how neonicotinoids are seen and managed in NZ. The inconsistency between these cases shows how important it is to understand under what conditions an agricultural practice, thrives and dies (McNamara et al., 2004). My analysis in chapters 5 and 6, therefore, establishes how particular relationships have made and maintained this agricultural practice.

Chapter 3

Literature Review

3.1 Introduction

Actor-Network Theory (ANT) is a group of tools, sensibilities and methods of analysis which map relationships between things and concepts simultaneously and was created to understand the process of knowledge creation in the broader discipline of Science and Technology Studies (STS). ANT was developed by Michel Callon and Bruno Latour (Callon, 1986; Callon and Latour, 1981) in Paris during the 1980s. It is also often associated with John Law, who began shortly after and later collaborated with Callon (Callon & Law, 1982). Because ANT has been used to study everything from scallop fisheries (Callon, 1986) to nuclear fission (Latour, 1999a), and offers up a breadth of conceptual tools, researchers must define how they use the tools within its literature very specifically. Subsequently, I have restricted myself to a very specific and workable approach which uses a limited number of tools from the ANT literature.

Therefore, this chapter gives an overview of Latour's CSSF (1999a) and Callon's key principles (1986), the specific tools from within ANT literature used in this thesis. It also provides explanations of why these tools have been chosen and how they will be applied. Firstly, I will outline the key principles of ANT (1986) that I have used in this study. In particular, I will introduce the theoretical tools which will be utilised during Chapter 6 to explore how the relationships between human and non-human actors have made and maintained neonicotinoids in NZ's agricultural practices. More specifically this chapter introduces Bruno Latour's (1999a) circulatory system of scientific facts (CSSF), a framework which outlines five types of activities that "science studies need to describe first if it seeks to begin to understand in any sort of realistic way what a given science discipline is up to" (Latour, 1999a, p. 99). This framework guides both the data collection and analysis stages of this thesis: it will help me follow the actors and examine the differences between a network that maintains the use of neonicotinoids, and another network where neonicotinoids have been removed. I will also outline a number of criticisms and commentaries of the CSSF. This is a necessary procedure to answer my fourth research question, which aims to review the appropriateness of the CSSF for investigating hazardous chemical controversies in the future (see section 1.4). Finally, this chapter will point to some literature that makes inferences about what makes and maintains agricultural practices.

3.2 Key actor-network theory principles

ANT looks at how science and technology are made and maintained (Latour, 1987), and it directs researchers to "follow the actors" (Latour, 1993). This is a relatively simple request. However, there can be various actors involved in any network.

This, then, is the core of the actor-network approach: a concern with how actors and organisations mobilise, juxtapose, and hold together the bits and pieces out of which they are composed; how they are sometimes able to prevent those bits and pieces from following their own inclinations and making off; and how they manage, as a result, to conceal for a time the process of translation itself and so turn a network from a heterogeneous set of bits and pieces each with its own inclinations, into something that passes as a punctualized actor. (Law, 1992, p. 2)

ANT allows researchers to understand how actors come together and form networks of heterogeneous material due to some aligned interest (Law, 1992). Furthermore, ANT explores how these networks are made, maintained and broken down (Tatnall and Gilding, 1999). There are three key principles to ANT, as described by Callon (1986). These principles are "agnosticism", "generalised symmetry" and "free association" (p. 196), and their purpose is to help manage an ANT analysis while surmounting other methods of sociological analysis (González, 2013). In effect, they overcome those methods which ignore the role of nature and non-human actors in a phenomenon and instead give privilege to humans and society.

The first principle "agnosticism" suggests that non-human actors make up the world in which humans survive, and so no actor should receive a privileged status given its social standing, species or even cognition (Callon, 1999; Harding, 2017). Callon describes it as "impartiality between actors engaged in controversy" (Callon, 1986, p. 196). Researchers must be impartial in their observations by systematically avoiding censoring any interpretations provided by the actors (Callon, 1986; Law, 1986, 1987). Furthermore, researchers should abandon any preconceived ideas regarding the networks makeup, causation, creation and the reliability of an actor's account (Ritzer, 2005). "Agnosticism" ensures that there is no censorship of the actors' accounts by the researcher, even if those accounts do not align with the view of the researcher. Any form of censorship will likely hinder an accurate and developed understanding of the controversy at hand. Compliance with the principle of "agnosticism" requires consideration of "the facts of Nature and the social contexts which they elaborated and shaped" (Callon, 1986, p. 221). In addition to this, it requires a faithful report of doubts about society and potential "alliances". This principle raises several questions about how I have avoided judging the positions taken by my interview participants, reducing them to a particular sociological interpretation. Furthermore, it makes me consider how I have remained impartial in my line of questioning, my selection of interview participants and my analysis of networks.

The next principle is called "generalised symmetry" (Callon, 1986, p. 196), and it describes how conflicting opinions of actors should be explained in the same neutral terms, that work in the same way for both humans and non-humans (Callon, 1986). An actor is anything that can "exert detectable influences on others" (Law, 1987, p. 132). Fundamentally, both nonhuman and human actors can take action and can also be anything or anyone (Law, 1986). Dankert (2011) reinforces this, stating that "humans first shape the buildings and then are shaped by the same buildings." Therefore, actors should not be defined on a priori definitions but on what they do. Callon and Latour (1981) explain that actors are "any element which bends space around itself, makes other elements dependent upon itself and translates their will into a language of its own. An actor makes changes in the set of elements and concepts habitually used to describe the social and the natural worlds." A researcher who is performing an analysis using tools from within ANT should, therefore not give humans priority over non-humans to take action (Law, 1992). Because of this, status, species and even cognition should not influence the notion of what defines an actor. Humans, technology, organisms, texts, buildings and objects should all be given the status of an actor within an actor-network. Therefore, compliance with the principle of "generalised symmetry" requires sticking to the same vocabulary throughout a study. Subsequently, the terms; actors, relationships and those of Latour's CSSF (1999a), are applied without discrimination to all actors throughout this thesis (Callon, 1986, p. 221). This principle raises questions about how I have avoided using "institutional and organizational configurations to explain" the making and maintenance of agricultural practices in NZ and avoided explaining the on-going use or discontinuation of neonicotinoids as coming down to a "balance of power or to a series of conditions" (Callon, 1986, p. 222).

Lastly, there is "free association" (Callon, 1986, p. 196) which demands that potentially anticipated or pre-existing distinctions between humans and non-humans should be abandoned, like those between the social, environmental and technological (Callon, 1986). Therefore, the actors' relationships should be allowed to fluctuate and become the focal point of the analysis. Essentially, "free association" proposes that actors on all levels and of all forms can interact and influence one another (Harman, 2014) and that nature is social through and through (Castree and MacMillan, 2001). Fundamentally, the principle of "free association" demands that researchers "follow the actors in order to identify the manner in which these define and associate the different elements by which they build and explain their world, whether it be social or natural" (Callon, 1986, p. 201). Latour (2003, p. 12) explains that the directive to follow the actors is a command to search for connections and find out how and why new relationships form. Complying with the principle of "free association" requires avoiding *a priori* categories and relationships. This allows me to follow neonicotinoids through the various relationships which make up the networks central to this thesis. It also allows me to view neonicotinoids as an object which is changing rather than locking it into a

fixed role. Furthermore, the roles of each actor and their subsequent relationships can fluctuate. Therefore, unpredictable relationships which "become visible and plausible only after the event" (Callon, 1986, p. 222) can be described and understood.

These three principles are supported by Latour (1993) when he shows how other sociological theory forms distinct categories and argues that to understand controversies, we must break down those distinctions. Like Callon (1986), Latour (1993) is arguing against *a priori* assumptions. Subsequently, ANT is not a theory of the social (Latour, 1996, p. 22), but a theory of "how to study things" (Latour, 2005, p. 142). By following these three principles this research is free of the dualities between nature, society and technology that is usually assumed. This, therefore, allows me to look at the importance of non-human actors without bias and trace the making and maintenance of an actornetwork that embeds -and maintains- neonicotinoids in NZ's agricultural practices without locking actors into fixed roles or censoring their accounts.

3.3 Actor-networks and network representations

When Latour (1993) states that all classifications and categories result from rather than cause social organisation, the social organisation he is referring to are the actor-networks he attempts to describe. An actor-network is a group of actors linked together through activities, relationships and associations (Law, 1992, 2002; Latour, 1996). Callon (2012) says that everything "is an actor-network reducible neither to an actor alone nor to a network. An "actor-network is simultaneously an actor whose activity is networking heterogeneous elements and a network that is able to redefine and transform what it is made of" (p. 93).

The process called "punctualization" is the simplification of an actor-network so that it is observable as a single actor. This actor is simple, appears only for a time and masks the networks that produce it (Law, 1992, p. 385). Once an actor-network becomes stable and punctualised, it can no longer be questioned or tested, so it forms a "black-boxed network" whose behaviour is now independent of its context (Callon & Latour, 1981, p. 285). An example of a "black-boxed network" is a modern computer. When someone uses a computer, they do not consider the complexities which are involved in its making and maintenance (Pantumsinchai, 2018). Like this, the process of "black-boxing" sees sub-networks disappear and therefore, actor-networks become actors (Van House, 2004). Subsequently, a network is a collection of black boxes which are themselves networks. Callon and Latour (1981) explain this as; "a black box contains that which no longer needs to be considered, those things whose contents become a matter of indifference" (p. 285). A punctualized actor-network can change and evolve because actors outside of the punctualized network can weaken relationships in it. Therefore, a stable, punctualised or even "black-boxed" actor-network can revert to a complex network of actors (Law, 1992).

Tools from within ANT, like the CSSF, attempt to crack open these punctualized networks that appear to be singular objects. Network representations are particularly good at revealing these networks inner workings, identifying where they might face resistance and therefore allow a researcher to describe the complex network of material relations which make them up. Law (1992) maintains that only a network which is made up of and supported by various durable material relations can be stable. Therefore, only by using a network representation like Latour's CSSF (1999a), which reveals and follows material relations, can I identify whether a network is stable and therefore describe how neonicotinoids are embedded -and maintained- in NZ's agricultural practices.

In a similar train of thought, Latour (1999a) discusses how associations or relationships between actors are continually being formed, reformed and destroyed in any given setting, which sometimes creates new actors. These new actors have different attributes from those of the original actors involved in their making (Dankert, 2011). Latour (1999a, p. 180) uses the example of a gunman to explain this idea of actor creation more clearly. The gun and the man are separate entities, but when put together they form a hybrid body, the gunman, which creates a new distinct actor that is greater than the sum of its parts. Therefore, when someone is shot, it is the relationship between the gun and the man, forming the gunman that matters. Network representations, therefore, encourage their users to look at the underlying structures of fixed concepts (Waltz, 2006, p. 23). By following the actors, a comprehensive network emerges, which would otherwise appear as an intricate group of actors with misshaped connections. Thorough research will reveal the well-established associations between actors, and only empirical evidence can be described and therefore be real.

STS researchers have repeatedly critiqued the application and description of the oxymoron "actornetwork". Latour (1999b) stated that "there are four things that do not work with actor-network theory; the word actor, the word "network", the word theory and the hyphen! Four nails in the coffin" (p. 1). The first "nail" is concerned with the word "network", and Latour claims that the internet has meant that everyone now believes they know what it means. He says the typical description of a "network", "transport without deformation, an instantaneous, unmediated access to every piece of information" (p. 1), is precisely the opposite of its meaning in ANT. To Latour, the term "network" is like the term "rhizome" as used by Deleuze and Guattari (1988), it is a set of transformations. Researchers must view "networks" as a set of translations, transformations and transductions, inapprehensible to the traditional terms of sociology.

> Being connected, being interconnected, or being heterogeneous is not enough. It all depends on the sort of action that is flowing from one to the other, hence the words "net" and "work". Really, we should say "worknet" instead of "network". It's the work, and the movement, and the flow, and the changes that should be stressed. But now we are stuck with "network"

and everyone thinks we mean the World Wide Web or something like that. (Latour, 2005, p. 143)

For a researcher to effectively perform a network analysis, like the CSSF, they must be aware of the unit of study, the network. More specifically, they must understand that network representations aim to describe how links are made and maintained and what is linked together. It is for this reason that Latour claims "worknet" would be a better description for the unit of study. Furthermore, Latour no longer feels the term "network" should be used as it does not correctly describe the translations that a researcher wants to explore.

The second "nail" refers to the word "actor" and its hyphenation with the word "network". Latour states that he always opposed the use of a hyphen as he claims it reminds the researchers of "the agency/structure cliché [of social theory]" (Latour, 1999b, p. 16). Law (1999) states that:

Truth and falsehood. Large and small. Agency and structure. Human and non-human. Before and after. Knowledge and power. Context and content. Materiality and sociality. Activity and passivity...all of these divides have been rubbished in work undertaken in the name of actor-network theory (p. 3)

This relentless throwing out of the *a priori* categories is what separates network representations from other sociological approaches. Therefore, by reminding researchers of these clichés, this hyphenation of actor and network, brings them too close to those divides which they attempt to avoid.

The third "nail" considers the word theory. Latour (1999b) argues that ANT should be called "actantrhizome ontology" (p. 19). He continues by saying that ANT was never a theory of what makes up the social, as he explains:

> ANT was simply another way of being faithful to the insights of ethnomethodology: actors know what they do and we have to learn from them not only what they do, but how and why they do it. It is us, the social scientists, who lack knowledge of what they do, and not they who are missing the explanation of why they are unwittingly manipulated by forces exterior to themselves and known to the social scientist's powerful gaze and methods. (Latour, 1999b, p. 19).

This thesis takes note of this critique and instead views ANT, as a crude set of methodological tools for learning from actors without imposing *a priori* definitions on their "world-building capacities" (Latour, 1999b, p. 19).

The fourth and final "nail" revisits the hyphenation between actor and network. Latour explains that most misunderstandings about ANT stem from this linking of words (1996, p. 16). He emphasises that

the hyphenation represents the debate between agency and structure, that the fathers of ANT never wanted to enter.

Regardless of this, Latour expresses a need to "not abandon the creature to its fate but continue all the way in developing its strange potential" (1999, p. 24). Because I am attempting to find out if the CSSF (Latour, 1999a), a network representation, is useful in understanding how agricultural practices are made and maintained, I have followed Latour's direction to continue developing its strange potential. I will, therefore, continue to use the phrase "actor-network theory" and its parts throughout this thesis. More specifically, I will use Callon's key principles (1986) and Latour's CSSF (1999a) to understand how neonicotinoids have become embedded –and remain – in NZ's agricultural practices.

3.4 The circulatory system of scientific facts

Research provides a representation of reality, which enables critical reflections of that representation and specific choices about its form (Gad and Jensen, 2010). Subsequently, scholars have continuously experimented with different representations of actor-networks. Often a single image is central to a network analysis, as is the case with this thesis. Therefore, what follows is an overview of the specific network representation used in this thesis, a review of its previous case study applications and an indication of how it has been used as a research methodology here.

I have applied, explored and critiqued Latour's (1999a) CSSF. The CSSF discourages researchers from assuming a phenomenon is down to a particularly powerful actor and instead encourages researchers to focus on tracing the material-semiotic relations between actors in a network. These relations "are simultaneously semiotic (because they are relational, and/or they carry meanings) and material (because they are about the physical stuff caught up and shaped in those relations)" (Law, 2019, p. 1). So rather than assuming neonicotinoids are embedded -and maintained-because of the actions of a particularly powerful actor, adopting a CSSF approach made me re-evaluate how they became and remained embedded in agricultural practices. The CSSF approach subsequently encouraged me to explore the material relations developed between various actors who are all interested in neonicotinoids. The CSSF (Latour, 1999a) thus became a useful tool for me from the outset.

The CSSF is a visual representation of an actor-network, and its foundational purpose is to show how all scientific facts (controversial or not) can be kept alive (Tabak, 2015). Latour's CSSF shows the outcome of any scientific research as a result of five interrelated types of activity, each of which involves several actors (Latour, 1999a, p. 99). It describes how scientific facts can exist, remain stable and survive (p. 99). Latour introduces the CSSF during his book titled: *Pandora's Hope: Essays on the*

reality of science (Latour, 1999a). Not only has this visual representation of an actor-network (Fig. 1) directed me to more actors, their roles and power dynamics, but it will also highlight the kinds of actors that participate in this network and the network constructing activities which each actor undertakes (Warner, 2008).

Latour's CSSF describes

five types of activities that science studies needs to describe first if it seeks to begin to understand in any sort of realistic way what a given science discipline is up to: instruments, colleagues, allies, public and finally what I call links or knots (Latour, 1999a, p. 99).

Material removed due to copyright compliance

Figure 1 The circulatory system of scientific facts, sourced from (Latour, 1999a).

Although the CSSF diagram contains very little detail, it is far from being a simple depiction of network activities. The first loop is called "mobilization of the world", and it is about how actors convert the object of their study into discourse. It also describes how non-humans are progressively loaded into discourse and includes the expeditions, surveys, instruments and equipment that are used in scientific research. It also consists of the sites where all this equipment is assembled, such as laboratories, museums and databases (Latour, 1999a, p. 101). The second activity is "autonomization", which describes the construction of scientific professions, disciplines and institutions that support and contribute to research as colleagues (p. 102). The third activity is called "alliances", and it describes a further group of actors from outside the scientific community that must also support a programme of research, such as politicians, industrialists and the military. It

considers the enrolment of groups and individuals that would typically remain outside of the discourse and is about creating "alliances" through worked persuasion and liaison (p. 103). The fourth loop, known as "public representation", describes the creation and upkeep of relations with the outside world, that of reporters, pundits and the layperson. It considers the role of the public (p. 105). Finally, the scientific content is depicted as the last activity called "links and knots". This name was given by Latour to avoid the historical baggage that comes with the phrase "conceptual concept" (p. 106). The "links and knots" tie all the other loops together but cannot exist in isolation from them, serving as the conceptual "heart" of the network.

3.4.1 Agricultural practices and case study applications of the CSSF

In some circumstances, the CSSF has already been applied to case-studies which investigate, the making, maintenance and removal of agricultural practices (Ingram, 2007; Warner 2016). Therefore, this section analyses and synthesises a variety of works which look at how agricultural practices, as well as disciplines, are made and maintained from a CSSF perspective and the perspective of other social science methods. I have subsequently drawn a number of conclusions about what the CSSF offers studies which look at how agricultural practices are made and maintained, and therefore what the CSSF offers this research.

Overview of previous research that has studied agricultural practices

To understand what the CSSF offers research which looks at the making and maintenance of agricultural practices it is essential to highlight the conclusions that other social science methods make about agricultural practices. Agricultural practice literature often focusses on how individual motivations and benefits affect the adoption of new and innovative agricultural practices (Bangura1983; Lalani et al., 2016; McManus & Powe, 2007; Pannell, 2008;). Another common trend is to look at how the individual and situational characteristics of farms and farmers affect the maintenance of agricultural practices (Rogers & Shoemaker 1971; Waters et al., 2009).

In a study that looked at why farmers are using conservation practices, Lalani et al. (2016) found that the usefulness of conservation agriculture to the farmer and the farmers perceived importance of this practice was the most reliable indicator for a farmer's intention to use conservation agriculture on their farm. Similarly, when farmers perceive that an agricultural practice is viable on their farms (Bangura, 1983), or is better than the method or agricultural practice it supersedes (Rogers & Shoemaker, 1971; Giera et al., 2006; McManus & Powe, 2007) they are more likely to adopt alternative and novel agricultural practices. Economic reward has also been another research interest of social scientists investigating the making and maintenance of agricultural practices. Lalani et al. (2016) found that short term yield and a higher return on investment was a crucial driver in the adoption and maintenance of conservation agriculture practices.

The personal and family circumstances of a farmer are commonly considered to be a significant driver in the behavioural decisions of farmers, and therefore their intention to adopt agricultural practices. Bangura (1983) suggested that the individual goals of farmers had the most significant impact on their adoption of innovative agricultural practices. This work also suggested that those farmers with higher aspirations were more likely to adopt new and recommended agricultural practices. Those farmers who were motivated by; the recognition they would gain from adopting the agricultural practices and the perceived goals they would attain, were again more motivated to adopt the innovative agricultural practices. Similarly, some studies suggest that because farming is not just a job, but a lifestyle, farmers carefully consider the uptake of an agricultural practice because of the way it might affect their land and the environment (Vanclay, 2004; Ahnström et al., 2008). This suggests that what might be suitable for one farmer is not necessarily suitable for all farmers.

Giera et al. (2006) and Lalani et al. (2016) come to similar conclusions about what factors cause the making and maintenance of agricultural practices. They ultimately suggest that a farmers perception of the relative advantage or "the balance of benefits" (Lalani et al., 2016, p. 89) which an agricultural practice provides is what determines its adoption and therefore its maintenance.

These studies reveal that a farmer's decisions to adopt agricultural practices comes down to the selfmotivated intrinsic and extrinsic rewards which that adoption might offer. Therefore, each of these studies have a very individualistic understanding of what makes and maintains agricultural practices; they each suggest it comes down to the individual farmer. This raises questions about what previous research that has studied agricultural practices miss. Firstly, they tend to have a bias towards the individual because they suggest the adoption of an agricultural practice can be solely explained by the individuals motivations, actions and interests. In doing this, they neglect the relevance that other factors can have on the making and maintenance of agricultural practices. They also tend to rely on experimental approaches for data collection, which limits their ability to assess the messy social complexity of a phenomenon. Furthermore, these studies view adoption as a onetime event, which is not affected by previous interactions, they state that any change comes down to individual choices. So, although the findings of these studies are valuable to understanding the decision-making practices of farmers, they do not pay attention to the material-semiotic relations which contribute to an agricultural practice being made and maintained. This, in turn, undermines an understanding of the making and maintenance of agricultural practice. Subsequently, it raises several questions about what the CSSF (Latour, 1999a) can offer studies of agricultural practices and whether it is suitable for acting as a sensitising device to examine the making and maintenance of neonicotinoids in NZ's agricultural practices. Therefore, what follows is a synthesis of those case-study applications of the CSSF which have an agricultural, scientific and disciplinary focus.

Case study applications of the CSSF

Unlike the above studies of agricultural practices, the CSSF encourages researchers to follow the actors and look at how material relationships are contributing to the making and maintenance of a phenomenon. Therefore, to argue the case for the use of Latour's CSSF (1999a) in this thesis, what follows discusses five studies that use this network representation to demonstrate how agricultural practices, professions and scientific projects are made and maintained through a series of material relations.

The significant works which I have reviewed are Fitzsimmons (2004), Ingram (2007), Edwards (2014) and Warner (2008, 2016). Ingram (2007), who studies three alternative methods for improving soil fertility, and Warner (2016), who studies how the work of Rachel Carson reduced the use of agricultural chemicals, are considered for their application of the CSSF to agricultural practice case studies. Edwards (2014) who investigates the development of a genetically modified tearless onion is considered for her application of the CSSF to a NZ setting. Finally, Fitzsimmons (2004), who studies the vascularization of ecology in the USA, and Warner (2008), who studies the vascularization of agroecological principles, are considered because they step away from analysing a single project and apply the CSSF to broader disciplines.

All these studies are concerned with the activities of human and non-human actors to interest one another and translate each other's interests. Moreover, each of them uses the CSSF to emphasise the dynamic relationships which contribute to the making and maintenance of a scientific research project (Edwards, 2014), a scientific discipline (Fitzsimmons, 2004; Warner, 2008) or an agricultural practice (Ingram, 2007; Warner, 2016). Ingram builds on this by saying that the CSSF emphasises the dynamic relationships that exist with material nature, the strategies which scientists use to mould their ideas and the alliances they develop to convince others of their research (p. 300). Subsequently, the ways which actors engage material nature, create scientific facts and enrol a wide range of audiences, can become very clear (p. 309). Edwards similarly uses the CSSF to highlight the relevance of each relationship to the integrity of the circulatory system. Furthermore, Fitzsimmons (2004) uses the CSSF to examine who and what ecologists engage with, revealing that ecologists are "actively, transmitting and circulating the arguments of their community" (p. 36). Thus, allowing her to demonstrate how ecologists made a "structure of engagements for translating knowledge into practical action and political power" (p. 36). Each of these studies subsequently highlighted the versatility of the CSSF (Latour, 1999a) as a methodological tool which makes sense of the relational nature of various phenomenon. Therefore, not only does the CSSF present itself as a useful tool for making sense of how neonicotinoids are made and maintained in NZ's agricultural practices, but it offers a different way of explaining the adoption of agricultural practices which the previous literature, described above, has not considered.

Edwards (2014) demonstrates that the CSSF can be used to "convey the distribution of interests that can be coordinated to perform a fractionally coherent object" (p. 147) and to "provide a detailed account of how each actor is joined together through the circulation of various intermediaries" (p. 146). Callon describes intermediaries as "anything passing between actors that defines the relationship between them" (1991, p. 134). Although none of the other case studies highlight the presence of intermediaries, I have utilised them to try and make sense of the relationships that are maintaining neonicotinoids as an agricultural practice in NZ. It is all well and good to say that a relationship exists, but it is difficult to make inferences about that relationship unless you understand how it functions, or understand what resources each actor is trading in order to interest one another in the relationship and their individual goals. Therefore, I have referred to this transfer of intermediaries as "material relations" throughout this thesis. Furthermore, it is these intermediaries that I identified while using the CSSF in the data collection of my thesis (e.g. crop protection programmes and critical use cases), and which have then directed me to the actors taking part in their trade.

Warner (2016) utilises the CSSF to analyse scientist's engagement with the world outside of the "cloistered laboratory" (Warner, 2016, p. 30). Edwards (2014) similarly details the involvement of various actors who "extend beyond the walls of the laboratory to include a heterogeneous mixture that breaks down the divides between science, society, and nature" (p. 120). While Ingram argues that actor-networks concerned with soil fertility management rely on the development of scientific knowledge to gain legitimacy with farmers, consumers, policymakers and others (p. 299). Furthermore, Warner (2008) describes how agroecology cannot be made or maintained in California without the association between various different actors who exist both inside and outside, the laboratory and the field. All of these studies suggest that looking beyond the boundaries of the laboratory or a particular community can provide new light to understand the making and maintenance of scientific facts or agricultural practices. Once again, this is something which other methodologies for studying the adoption, making and maintenance of agricultural practices do not consider but which the CSSF offers. Thus, by following in the footsteps of these CSSF researchers and looking beyond the individual decisions of farmers, this thesis will potentially offer a new direction and considerations for researchers who want to investigate hazardous substance or agricultural practice controversies in the future.

These previous case studies have each used Latour's loops to help them make sense of a phenomenon and identify those actors who are involved. "Mobilization of the world" (Latour, 1999a, p. 99) is the tools and sites which scientists use (Fitzsimmons, 2004), or in other words the process by which humans investigate natural phenomena and subsequently forge social behaviour (Warner, 2016). Warner (2016) uses this loop to demonstrate how Rachel Carson joined farmers in the field

armed with scientific methods, devices and statistics to illustrate the ecological degradation caused by agrichemicals (Warner, 2016, p. 31). "Autonomization" (Latour, 1999a, p. 102) describes the development of institutional structures and their activities (Fitzsimmons, 2004, p. 37), the relationship between extensionists and agricultural scientists (Warner, 2008, p. 772), or the importance of peer validation (Warner, 2016). Warner (2016) emphasises that without the approval of these peers "resources dry up, disciplines wither, and lines of scientific investigation fail" (Warner, 2016, p. 32). "Alliances" (Latour, 1999a, p. 103) describes governmental as well as non-governmental organisations (Fitzsimmons, 2004, p. 40), grower organisations (Warner, 2008) and those patrons who put scientific knowledge to practical use. Warner 2016 states that "knowledge contained by the academy is feeble compared to that put into economic action" (Warner, 2016, p. 32) and therefore, he emphasises that this loop moves in more than one direction because industry asks scientists for help and scientists need to convince industry of the commercial potential of their research(p. 32). "Public representation" (Latour, 1999a, p. 105) is the representation of ecology as that discipline who speaks for the laypeople and the millions of non-humans which humans share the environment with (Fitzsimmons, 2004, p. 40) or how environmental agencies represent agricultural pollution to the public, while also representing the public in the agricultural industry (Warner, 2008). These case studies which have previously used the CSSF as a way of representing, analysing and finding a network have not given away many of the practicalities involved in using this model. However, their descriptions of each loop gave me some indication of what to look for when applying this model to NZ's neonicotinoid use (see sections 3.2.4 and 4.2.2).

Warner (2016) brings forth considerations about how the circulation of scientific knowledge demonstrating the detrimental effects of agricultural chemicals can catalyse change in on the ground practices. This raises questions about the third research question of this thesis: How are neonicotinoids as an agricultural practice embedded and maintained in NZ, and why has opposing knowledge been unsuccessful in removing it? Warner's (2016) application of the CSSF revealed how Rachel Carson was able to vascularize the knowledge from her scientific discovery to the public, which meant that the public began working on behalf of nature. Subsequently, Carson also circulated knowledge from the public, back to the field, which reduced the use of agricultural chemicals (p. 32) and strengthened the network. Warner (2016) continued by insisting that even when the public's view of agricultural practices is distorted from the truth, they have a say in how agricultural practices are done. Warner (2016) further shows that Rachel Carson's critics could not disrupt the network destabilizing her research because she had vascularized her scientific discoveries to the public. Therefore, in terms of this thesis, the CSSF encourages me to take a closer look at how neonicotinoid proponents have vascularized the benefits of neonicotinoids to the public and powerful allies,

thereby strengthening the network and preventing the scientific research opposing the use of neonicotinoids from disrupting the network.

Warner (2016) also gives insight into how a slight change in one of Latour's loops (Fig. 1) can circulate through all the other loops, moulding the network into something different. Similarly, Edwards (2014) found that the tearless onion project failed because an attempt to move the onions out of the laboratory brought anti-GMO spokespeople into the network, subsequently destabilizing the network stopping the life-support of the project. Furthermore, Warner (2008) shed light on how the CSSF (Latour, 1999a) is a useful tool for understanding how knowledge interventions can stimulate activity in multiple loops of a network. He states that "well-designed interventions ripple through multiple loops, and successful partnerships have devised strategies that echo through all of them" (Warner, 2008, p. 772). Warner explains that any new knowledge which stimulates economic activity will strengthen the network (2008, p. 772). These works highlight the CSSF's ability to pinpoint where a change in the network occurred and therefore describe how that affected all the other loops. This subsequently reinforces Latour's statement that "each of these five activities is as important as the others, and each feeds back into itself and into the other four" (1999a, p. 99). Moreover, these works have encouraged me to pay particular attention to those relationships or partnerships which design strategies to influence other parts of the network, to gain their support for neonicotinoids and therefore strengthen the network which maintains neonicotinoids.

Edwards (2014) demonstrates how the CSSF can be applied to present-day environmental issues. She uses the CSSF to represent and study an environmental controversy while it was ongoing, which enabled Edwards (2014) to interpret the outcomes of the GMO onion project. Similarly, the CSSF allowed Warner (2008) to interpret the trajectory of change within the Californian public extension programme during a period of flux, created by pressure to reduce agricultures environmental impacts, which subsequently forced additional actors into the network (p. 771). Both Edwards (2014) and Warner (2008) were able to follow the actors as they built relationships with one another, negotiated and moved "interessement devices". Other studies portray the adoption of agricultural practices as a one-time event. Studies which use the CSSF, on the other hand, can investigate a phenomenon as it changes and transforms. By facilitating Latour's directive to follow the actors, the CSSF allows actors to fluctuate and enables researchers to consider how the historical setting and present-day changes can impact a network. This comes back to the CSSF's ability to show how a change in one part of the network can flow through all the other loops. If something changes in the present because the researcher has highlighted the material relations and the vascularization of the object, it becomes easy to make inferences about how that change will affect the other loops and therefore the outcome of the network. The ability to trace actors and the object, as they fluctuate, is vital to this thesis because neonicotinoids are still used in NZ.

Ingram (2007) explains how the three methods for improving soil fertility and therefore the three networks she studied each involved various actors. She then highlights that in every case these actors were brought together due to their shared belief that agricultural science needed to do a better job at working with nature. The CSSF similarly allowed Edwards (2014) to show how a variety of actors not typically connected have come together because of their shared interest in the GMO onions project. While for Warner (2008) it is the principles from the science of agroecology and for Fitzsimmons (2004) it is the scientific centrality of ecology, great social significance and urgency which bring together various actors into a stable network. These studies have shown that a variety of things make up the pumping heart, or the "links and knots", of a network. Subsequently, they shed light on how the CSSF can be used to examine how a variety of actors, with different political and ethical attitudes, can be joined together in a coherent network of material relationships around an object which they are either directly or indirectly interested in. This has significant implications on this thesis because unlike the other sociological methods usually associated with studies of agricultural practices, the CSSF can facilitate a study of an object by following that object through a network of relationships. This, in turn, uncovers actors and research participants which other sampling methods would not have identified and allows researchers to observe how an object is being framed and translated in a broader variety of material relationships.

Both Ingram (2007 and Warner (2016) have strictly adhered to Latour's original description of the CSSF (1999a). Both place the researcher or research team in a central role who mobilise the world, form a scientific profession, build alliances and convince the public of their research. In the case of Ingram (2007), this resulted in the success of an agricultural practice, while Warner (2016) described how Rachel Carson was able to change an agricultural practice. In both these applications, the construction or reduction of an agricultural practice is described as a scientific endeavour, which highlights the potential for an agricultural practice to be made and maintained because the scientific research concerned with its performance has gained legitimacy with colleagues, influential groups and the public. They both highlight that the participation of scientists and the pursuit of scientific research has led to a range of agricultural technologies and changing agricultural practices. This raises questions for this research about whether the making and maintenance of neonicotinoids is solely a scientific enterprise. Therefore by using the CSSF, this research looks at whether things like political, environmental and scientific settings inform the network and motivate the relationships of each actor.

Warner (2008) and Fitzsimmons (2004) step away from studying a specific instance of scientific research and show that the CSSF is a useful tool for studying broader case studies. Therefore, rather than representing how a researcher builds relationships to make and maintain their project as scientific fact, here the CSSF is used to describe how various actors have become interested in the

disciplines of ecology (Fitzsimmons, 2004) and agroecology (Warner, 2008). They also show how the material-semiotic relations of actors contribute to a stable network. This bodes well for this thesis because both these works have shed light on how the CSSF can be used to investigate more than just small one-off projects as described by Edwards (2014).

All these works (Fitzsimmons, 2004; Ingram, 2007; Edwards, 2014: Warner, 2008, 2016) highlight the benefits of the CSSF and are useful for identifying that the CSSF (Latour, 1999a) can enhance an understanding of how agricultural practices are made and maintained, by bringing forward its relational nature. They highlight that the making and maintenance of agricultural practice cannot be fully understood by solely looking at individual farmer decision-making practices. Furthermore, these studies which use the CSSF also provide insight into the role of various actors in the making and maintenance of agricultural practices and use the CSSF to understand controversies by describing the network as it appears. However, these works raise serious questions about how the CSSF can be implemented as a practical methodology for identifying actors, following these actors and analysing interview as well as documentary data. Therefore Chapter 4 highlights the practicalities that I employed while using the CSSF (Latour, 1999a) as a research methodology.

3.5 Limitations of the key principles of ANT and the CSSF

So far this chapter has explained why Callon's key principles (1986) and Latour's CSSF (1999a) might be useful for understanding how neonicotinoids are embedded -and maintained- in NZ's agricultural practices. However, it is essential to highlight the main criticisms of these tools to understand what they fail to achieve. Therefore, this section introduces three criticisms of Callon's key principles and Latour's CSSF.

The first criticism raised by Harding (1992, 1998, 2008), Bloor (1999) and Restivo (2010), argues that Callon's principle of "agnosticism" ignores social demographics and power structures like racism and patriarchy. However, "agnosticism" does not encourage researchers to ignore these structures outright; it merely removes any *a priori* relevance they have (Dankert, 2011). Instead, it encourages researchers to find out about power structures through the course of the research. Ingram (2007) also critiques Latour's CSSF (1999a) for its failure to deal with these relations of power and social hierarchies. She goes on to state that even though Latour claims to put the political back into the laboratory, he does not relate his ideas to the influential fields of academic study that focus on patterns of power and dominance (Ingram, 2007, p. 301). Ingram also disagrees with Latour's notion that the success of a network can be caused equally by a well-designed strategy of the actors or by being in the right place at the right time (p. 302).

The next criticism is of Callon's "generalised symmetry" principle (1986). Whittle and Spicer (2008) have heavily critiqued this assertion that a non-human can be an actor. They argue that this removes the pivotal role of humans and also state that this idea runs the "risk of displacing the defining human characteristics of the polis as a space of meaningful, purposeful, self-aware and non-repetitive action" (p. 620). Whittle and Spicer (2008) argue that the political action of humans is non-routine, which stands in contrast to those routine actions of non-humans. They continue by stating that "generalised symmetry" degrades an understanding of action "by obscuring the fact that it is only through the intervention of humans that agency—and thus political transformation of social arrangements—can occur" (pp. 620-621). Callon and Latour (1981) response to this is that whatever unit of discourse is used "is no way limited to 'human'" (pp. 301-302). Therefore, this symmetrical point of view is supposed to overcome that exact over-emphasis on humans agency which other sociological approaches have.

In response to these critiques, the research discussed in this thesis adopts a stance which acknowledges these social hierarchies exist but allows them to reveal themselves through the analysis of empirical evidence. By using the CSSF (Latour, 1999a) to find interview participants, follow the actors and analyse the material relations which make up the network which embeds -and maintains- neonicotinoids in NZ's agricultural practices, I have ensured that the *a priori* relevance of social hierarchies and priority to humans are removed. However, by presenting the conflicting viewpoints of all the actors and quoting the exact words of the interview respondents, limiting my biased commentary, I have still identified the power relations and social hierarchies as they have emerged from the data. Furthermore, by adopting the principle of "generalized symmetry" as an analytical stance and not an ethical position (Law, 1992) and therefore explaining the interview participants views in the abstract and neutral terms of Callon's key principles (1986) and Latour's CSSF (1999a) I have been able to describe the key role which non-humans play in the embedding - and maintenance- of neonicotinoids in NZ's agricultural practices.

A final criticism of Callon's key principles (1986), and network representations like the CSSF (Latour, 1999a) is regarding the reflexive approach. Murdoch (2001) argues that the "theoretical lexicon" (Whittle and Spicer, 2008, p. 620), which these tools impose does not match the accounts or language that the interview participants and therefore the actors would use themselves. Whittle and Spicer (2008) similarly argue that researchers, who use these tools, believe that they offer "a 'superior' or 'expert' view implying that the explanations of members are either 'naïve' or 'wrong'" (p. 618). They suggest that users of network representations need to hold "a more rigorous commitment to reflexivity" and "treat all accounts as in principle equal, including the one produced by the analyst" (p. 618). Whittle and Spicer (2008) also suggest paying close attention when using Callon's four-stage model (1986) to study anything other than Scallop fishing in St Brieuc Bay. They

argue that the "un-reflexive application of the four-stage model to other settings belies a positivistic attempt to verify the universality of Callon's original account" (Whittle & Spicer, 2008, p. 618). They are arguing that when applying this model unreflexively to new case studies, the researcher is merely reducing their research to a "series of deductible tests that confirm or refute the four-stage model of translation" (p. 618). They go on to suggest that the model should be used by the researcher as a concept to understand complex situations, without losing track of the empirical complexity of each case. Although I am not using Callon's model (1986) in this thesis, this criticism can likely translate to any network representation. Therefore, it raises questions about how I might have changed the data when making it fit the CSSF (Latour, 1999a).

I have continued to acknowledge these criticisms throughout this thesis attempting to be selfreflexive along the way, in the hope that this might alleviate some of the issues which arise from using Latour's CSSF (1999a) and Callon's key principles (1986). These criticisms will be revisited in Chapter 6 to discuss how they shaped this research and how I was able to overcome issues which arose from the methods chosen in this thesis.

3.6 Summary

ANT offers a wide range of theoretical and methodological tools that can be used across a variety of different social science disciplines to study various controversial phenomena. However, to understand how neonicotinoids have become embedded -and maintained- in NZ's agricultural practices, I have limited the tools which I will apply in this thesis. Therefore, this literature review has introduced the tools from within ANT which are used in this thesis.

Network analysis and more specifically, Bruno Latour's CSSF (1999a) are central to the study of neonicotinoids in Chapter 6. Latour intended the CSSF for the analysis of those mechanisms which make and maintain scientific facts within and outside conventional academic disciplines (Ingram, 2007). Moreover, it is a methodological tool for analysing how scientists engage with the social world outside of the cloistered laboratory (Warner, 2016). However, this thesis somewhat steps away from the study of how scientific research is made and maintained. Instead, it investigates how an agricultural practice can become embedded, even when the scientific consensus surrounding neonicotinoids is still up for debate (McGrath, 2014). I will use the CSSF to identify key documents and actors interested in neonicotinoids, follow the present-day material-semiotic relations of those actors and therefore represent and describe how neonicotinoids have been made and maintained in NZ's agricultural practices.

A series of case studies have informed this introduction to Latour's CSSF (1999a). Each of them adopts this network representation to make sense of the material relations that make up a network.

Furthermore, this thesis draws on those case-studies to make inferences about how neonicotinoids have become embedded -and maintained- in NZ's agricultural practices.

Alongside the CSSF (Latour, 1999a), I will attempt to adhere to the key principles of ANT, described by Callon (1986). "Generalised symmetry", "agnosticism", and "free association" not only ensure that the network representation is a more honest reflection of reality, but they help me minimise my biases throughout all stages of this thesis.

This chapter also highlights criticisms of these tools which I have considered while performing this research. Primarily these criticisms have encouraged me to be self-reflexive at every stage of this research, acknowledging where my methods, my chosen theoretical tools and my affiliations might introduce bias, giving an unfair reflection of reality. Furthermore, these criticisms have encouraged me to take care when adhering to Callon's key principles (1986).

Finally, this literature review considers nine papers which explain the making and maintenance of agricultural practices as coming down to the motivations, principles and actions of individuals. These will be considered alongside my discussion to highlight what the CSSF (1999a) and the Callon's key principles (1986) taught me about agricultural practices.

The following chapter describes the methodological aspects of using this approach. It introduces the case-study methodology commonly associated with the CSSF and outlines the specific applications of the CSSF applied during the data collection and analysis stages of this thesis. It also describes how the CSSF is used to represent the neonicotinoid network.

Chapter 4

Methodology

4.1 Introduction

The previous chapter introduced Callon's key principles (1986) and Latour's CSSF (1999a) as the tools which are used to make sense of how neonicotinoids have become embedded -and maintained- in NZ's agricultural practices. These tools, particularly Latour's CSSF (1999a), have helped inform and perform the process of data collection in this research.

Therefore this chapter will continue to describe the role of the CSSF in following the actors during this thesis. It will begin by discussing the case study methodology commonly associated with ANT. Then It will define and describe the exact methods of data collection used in this thesis. At the same time, I will discuss how I went about finding potential interview participants, how I accounted for ethical considerations, how I convinced participants to partake in the research and how I attempted to ensure that the interview data reflected reality. Lastly, this chapter introduced how I made sense of, organised and presented the data.

Researchers need to acknowledge how they shape the research process and its outputs (Palagnas et al., 2017). Furthermore, researchers need to recognise their "personal and methodological concerns" (p. 426) and their own "contributions to the construction of meanings and of lived experiences throughout the research process" (p. 426). By doing this and realising how one influences, acts upon and informs the very studies they engage in, a researcher not only honours those involved in the research but enhances the theory building and the research quality (Palagnas et al., 2017). Therefore throughout this chapter, I have attempted to be self-reflexive, by discussing where my interests, academics and career have introduced biases into this research. Predominantly this comes down to my work as a beekeeper, my pre-existing relationships with some interview participants, my time spent in Europe and my motivations for doing this research in the first place.

4.2 Case study

A common approach to social science research is the use of case studies. The meaning of the word "case" varies drastically, from an individual to a complex series of relationships (Marby, 2008). Bromley (1986) described case study research as beginning with the desire to derive an up-close or in-depth understanding of a single or a small number of cases set in real-world contexts. This comprehensive approach aims to produce an invaluable understanding of those cases, hopefully substantiating a better, or new understanding of real-world behaviour and its meaning. Similar to how the nature of the "case" can vary, the methodology which informs case study research is also not clear. Hammersley and Gomm (2008) suggest that the term "case study" overlaps with other research methods, such as ethnography, participant observation, fieldwork, qualitative research and life history. Case study research typically wants to discuss a broad range of contextual and other conditions related to the case. Subsequently, to fulfil case study research, the relevant data needs to come from multiple sources of evidence. By doing this, "case study research goes beyond the study of isolated variables" (Yin, 2014, p. 4) and describes how the context and other complex conditions relating to a central phenomenon are necessary for understanding it.

Yin (2014) describes three situations where case study methods can be applied. Firstly, to answer the projects research question (Shavelson and Towne, 2002). Case study methods are relevant to descriptive research questions, they ask what is happening or what has happened, and explanatory ones, they ask how or why something has happened (Yin, 2014, p. 5). This thesis asks "how" neonicotinoids have become embedded in NZ's agricultural practices. Secondly, case study methods favour the collection of data in natural settings, which is stressed by their emphasis on studying phenomena in their real-world context (p. 4). By performing some original fieldwork and interviewing participants concerned with the use of neonicotinoids in NZ, I gained a better understanding of the actors, their relationships and the phenomenon at hand, compared to if I had relied on a questionnaire or another instrument. The last situation is that case study methods can be used in evaluations (Yin, 1992, 1994, 1997; U.S. Government accountability office, 1990). This research project fits this criterion because it could be labelled as an evaluation of NZ's neonicotinoid use.

Yin (2014) emphasises that case study research methods "can cover the entire range of situations, from initial exploration to the completion of full and final authoritative studies, without calling on any other methods" (p. 6). A common concern with case study methods is a lack of trust in the procedures of case study researchers. This lack of confidence often stems from exposure to case studies where the material was altered to demonstrate a particular point more effectively (Gavin, 2003; Ellet, 2007). However, this alteration also happens in experiments (Rosenthal, 1966), questionnaires (Sudman and Burns, 1982) and historical research (Gottschalk, 1968). However, case study research is repeatedly criticised because these biases occur frequently and the method is youthful. Subsequently, researchers using case study methods must pay even more attention to the potential for bias to enter their research. Furthermore, case studies, like experiments, are "generalisable to theoretical propositions and not to populations or universes." (Yin 2014, p. 15). Fortunately, this inability to generalise is not an issue when using the CSSF because it does not produce grand narratives; instead it describes how a scientific fact or agricultural practice is made and maintained (Latour, 1999a, p. 99).

Case studies are "important elements in the epistemic landscape of STS" (Beaulieu, Scharrnhorst, & Wouters, 2007, p. 673) because "the discipline of STS works through case studies" (Law, 2008b, p.2). Moreover, case study methodology is the standard approach for research utilising network representations like the CSSF, and has been applied to everything from a bicycle (Bijker, 1995), nuclear fission (Latour, 1991) accounting management (Elsayad, 2016) and agricultural practices (Ingram, 2007; Warner, 2016). Case studies are a suitable methodology for CSSF research because it allows researchers to describe exemplary, historical and contemporary, moments and episodes. Therefore the reader can learn about the theory, the essential features of an argument and the historical cases or the contemporary controversies, through instances rather than in abstract (Law, 2008c, p. 6). Furthermore, case studies allow researchers to find out what "theories mean in practice, and about where they apply" (Law, 2008c, p. 7). Subsequently, researchers add to their ideas and build their arguments through case studies (Law, 2008c, p.6). For example, Latour (1988) uses the pasteurisation of French agriculture to argue how scientific facts circulate through laboratories into other sites. Case study methodologies are also relevant to research with an agricultural focus. Ingram (2007) used three cases of alternative soil management, to argue that a variety of different languages, frameworks and technologies are required to convince farmers to go back to nature. Thus case study methodologies appear to be a good fit for this thesis which focusses on how neonicotinoids are embedded -and maintained- in NZ's agricultural practices.

4.2.1 Case study methods

Yin (2014) discusses the six most common sources of evidence used in case study research; document analysis, archival records, interviews, direct observations, participant observation and physical artefacts. Yin emphasises that no case study method has a complete advantage over all the others. Instead, each has its comparative strengths and weaknesses. Yin (2014) also highlights that using multiple sources of evidence is vital for dealing with issues of data validity and reliability. Yin et al. (1985) found that case studies using many sources of evidence were rated higher in terms of quality and reliability than those relying on isolated evidence. Moreover, multiple sources of evidence allow the researcher to address a much broader context, which an isolated source might not identify. Additionally, this enables a researcher's lines of inquiry to converge, whereby the intersection can calculate the precise location of an object (Yin, 2014, p. 120). Subsequently, multiple sources of information make case study findings more accurate and convincing.

I chose to use interviews and document analysis over the other four sources mentioned by Yin (2014). This is because they are the sources, I am most familiar with, and they offer many strengths for this thesis. Firstly, the document analysis is predominantly used alongside the CSSF (Latour, 1999a) to identify and follow potential interview participants. I chose document analysis because

publicly available documents are easily accessed, they can be reviewed repeatedly, they are not created as a result of this case study, they provide broad coverage of the controversy, and they offer physical linkages which can be followed to identify actors interested in NZ's neonicotinoid use.

Interviews were chosen because they focus on the controversy, they provide novel conclusions about the research, and they allow me to engage with the actors I am trying to describe. I did not utilise physical artefacts or archival records because they were not readily available and I was limited in the time I had with the participants. Lastly and although participant observations are considered a cornerstone method of ANT, I did not utilise them here. Participant observation in ANT studies involves extended periods of fieldwork, the learning of a new language and ultimately becoming integrated into the community, like how Latour (1999a) spent an extended period in the Amazon observing an expedition with the aim of collecting and reporting its soils. Eventually this should allow the researcher to become an insider and be treated as such allowing their research to be based off pure observation. The controversial nature of this research topic, the time limitations of a one year master's thesis and the requirement of my chosen method which meant I had to interview many individuals from the entire network being observed meant that participant observation was not only impractical but it was simply not feasible. In simple terms I did not have the time to perform participant observation up to a standard which would have provided me with sufficient data to draw a good picture of those relationships that make up the neonicotinoid network. Therefore, what follows is a brief description of my chosen methods, and section 4.2.2 will detail how I applied these methods to this thesis.

Interviews

Interviews are the primary method for data collection in this thesis, as they are consistent with its interpretive nature (Creswell, 2007). Interviews allow researchers to understand actors' viewpoints (Clarke and Dawson, 1999) and give a "voice to common people, allowing them to freely present their life situations in their own words" (Kvale, 2009, p. 481). Although interviews take many forms, they all involve a consistent line of questioning. Because qualitative interview techniques view participants as "meaning makers" (Warren, 2002), in-depth interviews aim to provide participants with an opportunity to speak. The researcher's job is to allow "the interview subject to talk a lot–openly, trustfully, honestly, clearly, and freely – about what the researcher is interested in" (Alvesson, 2003, p. 17). In-depth interviews allowed me to capture the respondents' points of view without pre-determining their options through *a priori* questionnaires (Patton, 2001).

Effective in-depth interviews ask "how" questions and not "why" questions because they are nonthreatening and friendly (Becker, 1998; Yin, 2014). "Why" questions can create defensiveness in the participants and "how" questions are the preferred method of addressing "why" questions in an actual conversation (Yin, 2014). In-depth interviews are relatively easy to perform and do not require much of the participants time (Yin, 2014; Cloke, 2004). On the other hand, they demand a large investment of time on the part of the researcher. Because they encourage participants to speak freely, in-depth interviews often reveal novel information. Therefore, researchers must prepare for this with follow up questions to discover even more new data (Mabry, 2008).

There are two problems with in-depth interviews that researchers using this methodology should take into consideration. Firstly, they should maintain a level of reflexivity. Researchers must be aware of their involvement in the interview process (Bryman and Cassell, 2006; Pessoa et al., 2019) as they can easily impose their view on the respondent. This can occur through body language, terminology or by asking a question that offers insight into the rest of the interview schedule (Alvesson, 2003). Secondly, interview responses should not be taken as the truth (Randall & Pheonix, 2009, p. 125). Researchers must understand that "stories are told in particular situations to particular listeners for particular reasons" (p. 126) and "stories – not 'facts' and not mere 'information' – are ultimately what we are hearing in a qualitative interview" (p. 126). Essentially interview participants may want to portray themselves as something they are not and in a more favourable light (Alvesson, 2003).

To prevent these problems from occurring in my research, I exercised more caution when analysing my interview data, I built a rapport with the participants to gain data that more accurately reflected reality (Alvesson, 2003), I showed each respondent a level of respect which demonstrated that their responses were essential to my case study (Patton, 2001) and finally I never took for granted an interview participants answer in the first instance. Further detail of the practicalities involved in performing these interviews are presented in section 4.2.2.

Document analysis

The primary use of a document analysis is to "corroborate and augment evidence from other sources" (Yin, 2014, p. 107). They also can be gathered from a wide range of sources which have become easily accessible since the development of the internet although the sheer amount of data available can itself be problematic. Nonetheless, the document analysis does not play a central role in the data collection of this thesis. Instead, it has been used to make inferences which act as clues for further investigation (Yin, 2014, p. 107). More specifically, I have performed a document analysis in conjunction with Latour's CSSF (1999a) at the outset of this thesis, to identify potential interview participants (see section 4.2.2). Documents have also helped me verify spelling, titles and names.

By utilising a document analysis in this way, I have also been able to overcome some problems with case studies' reliance on documents. The first problem is the idea that researchers mistakenly assume all documents tell the truth (Yin, 2014, p. 108) One must also remain aware that documents can not only store information but build the heterogenous networks being studied (Prior, 2008).

They are active parts of the subject, event or case being studied and can eventually remove the need of someone to convey instructions (Cambell, 2004; Prior, 2008). Therefore, a researcher utilising documents must be attentive to the context in which they were written. Subsequently, by using the CSSF (1999a) to identify and follow the actors involved in the making and maintenance of agricultural practices in NZ, I am actively seeking to identify the intended purpose and audience of each document identifying not only how the document is used but also how it was made. Therefore, I can act as a vicarious observer, analysing documents as communications between actors, where one is attempting to achieve some objective. Yin (2014) states that "by constantly trying to identify these objectives, you are less likely to be misled by documentary evidence and more likely to be correctly critical in interpreting the contents of such evidence" (p. 108). Another problem with documents is that researchers can quickly become lost in reviewing the available material. Therefore, they must design strategies to deal with all the information they inevitably come across. Once again by using the CSSF, I have not only created a strategy to deal with all the information, but I have also inadvertently gone along with Hinchcliffe's solution (2001, p. 6) to follow the "controversies, arguments and debates through the empirical materials".

4.2.2 Methods of data collection used in this study

This section describes the research process and provides an overview of this thesis. It focuses on how I selected participants, how I performed the interviews, and how I accounted for ethical issues. This section predominantly discusses the interviews as they are the primary source of data collection, while a document analysis was used in conjunction with Latour's CSSF (1999a) to select research participants and to provide some background information at the outset.

Selecting research participants

This section describes how I chose research participants at the outset of this project. Firstly, I have attempted to adhere to Palaganas's (2017, p. 426) comment that researchers "need to acknowledge that indeed it is impossible to remain "outside of" one's study topic while conducting research." Therefore, I hope that by being self-reflexive and acknowledging where I have biases, I am making the findings of this thesis more credible. It is essential to recognise that I did not attempt to select participants randomly. At every point of the research process, I likely had some effect on the data, but it was at the beginning where I had the most impact. Since 2016, I have supported myself by working as a beekeeper, and it was because of this that I became interested in neonicotinoids. Because I experienced beehive deaths caused by the unlawful use of pesticides, because I own my own beehives and because I spent a significant amount of time in Europe while three types of neonicotinoids were being banned (see section 2.6), I have become an advocate for changing agricultural practices in NZ. Therefore, it is hard to deny that I am biased.

To identify potential interview participants and therefore follow neonicotinoids, I utilised Latour's CSSF (1999a) in conjunction with a document analysis. Publicly available documents regarding the sale of neonicotinoids, agri-seeds and pest monitoring services were vital in finding research participants. However, other materials including press releases, newspaper articles, company websites and public submissions on planned regulatory changes were also useful. The CSSF helped me follow the movements of human and non-human actors through their translation operations and material relations. While performing a document analysis, I utilised Latour's CSSF to direct my attention to those actors who are interested in NZ's neonicotinoid use. It also expanded my horizons to include actors whom I would have missed if I had solely relied on other sampling methods, such as snowball sampling. Latour's CSSF not only enabled me to expand my horizons but filter through a large number of actors indicated in the documents, focussing my attention on the most influential ones. The CSSF (Latour, 1999a) identifies five loops to pay attention to and urges for a close analysis of the networks that develop when experts, things, colleagues and laypeople interact. Thus, I attempted to apply the CSSF to the documents and find those actors and relationships Latour describes. One thing I noticed was that Latour's loops often directed me to the same actors, which was beneficial to my project because it again reduced the number of potential interview participants and suggested that some actors and some relationships were more significant than others.

Firstly, I tried to identify how neonicotinoid proponents have used instruments and loaded their arguments with non-humans. I also considered the chain of circulating documents, equipment, plants and places that uphold the vitality of neonicotinoids. Or, what Latour describes as "mobilization of the world" (Latour, 1999a). In doing this, I looked at actors like international chemical companies, NZ seed and chemical companies and products groups who were circulating these documents. The documents included press releases, media interviews, public submissions on planned regulatory changes, crop protection programmes, maximum residue levels and much more. And they were subsequently mobilised to the media, landowners, the NZEPA as well as one another.

Similarly, Latour's "autonomization" loop helped direct me to those actors considered agricultural practice experts. I identified fact sheets and pest control programmes written for growers and found the actors producing these documents and using them as intermediaries in their relationships. Those actors included organisations such as AGCARM, FAR, apples and pears NZ, Horticulture NZ (HortNZ), international chemical companies and NZ seed and chemical retailers. Latour's suggestion (1999a, p. 102) that experts are made through the convergence of numerous individuals who are all convinced of scientific research encouraged me to delve deeper into the relationships which have created these experts. By utilising snowball sampling, I was able to identify the other actors in the "autonomization" loop of each network (Chapter 6).

Next, I used Latour's "alliances" loop (1999a, p. 103) to identify the diverse actors involved in the making and maintenance of neonicotinoids. This directed me to look at their regulatory management, and I became aware of the neonicotinoid call for information made by the NZEPA (Environmental Protection Authority, 2018b). I also became aware of a list of priority chemicals whose regulation is out of date. The NZEPA constructed this list, yet it did not contain many of NZ's most controversial chemicals, including neonicotinoids (Hutching, 2018). By continuing to follow neonicotinoids, I was able to identify actors making submissions to preserve this chemical. I continued to follow the directive of the "alliances" loop and discovered a NZ house of representatives document (Ardern, 2014), which discussed neonicotinoids and the relationship between beekeepers and landowners.

I also utilised Latour's "public representation" (1999a, p. 105) during the document analysis, which describes how the benefits of neonicotinoids are popularised through the media and other publications. It's also about how people's everyday practice, system of beliefs and opinions are affecting agricultural practices. Subsequently, I looked at how some members of the public are becoming interested in neonicotinoids. I was also able to identify the global trend away from neonicotinoid use, and I began to consider the impact this would have on the production of NZ's exports. Sure enough, by following neonicotinoids and their bans in Europe, I highlighted the preferences of international markets for and against neonicotinoids. Subsequently, this guided me to look for those actors holding direct relationships with these markets.

As well as using Latour's CSSF (1999a) and a document analysis, I also drew on my pre-existing relationships to identify potential interview participants. Before embarking on this research project, I had pre-existing relationships with beekeepers and growers who deal with neonicotinoids in NZ. I chose to perform interviews with them and engaged a level of snowball sampling during those interviews. Snowball sampling is "a technique for gathering research subjects through the identification of an initial subject who is used to provide the names of other actors" (Lewis-Beck, 2004, p. 2). I utilised snowball sampling to continue following the actors. Essentially, during each interview, I paid particular attention to the people, organisations, regulating bodies and non-humans that the participants mentioned. I also invited each interview participant the opportunity to identify individuals who they believed would offer new data and be interested in this research. Often the individuals identified through snowball sampling, were colleagues, friends and relatives of those individuals I had already interviewed.

Interview schedules

Before conducting the face to face interviews, interview schedules were written up (appendix A.1) and approved by the Lincoln University Human Ethics Committee (see appendix A.3). I used five

different interview schedules. One which contained eight questions that were presented to all the participants and then a separate one for growers, regulators/product group representatives, seed and chemical company representatives and beekeepers. All the questions were open-ended to encourage the participants to tell their own stories. The interview schedules held the interviews focus on critical areas of interest. But some flexibility was allowed to let the participants speak about any new information which I had not yet come across, or which they found particularly important. Furthermore, this allowed me to use follow up questions where needed, mainly to understand those interesting and relevant points that an interview participant may have otherwise quickly passed over.

Conducting interviews

I conducted sixteen interviews with participants from NZ's arable sector as well as NZ's apples and pears sector between June and September of 2019. To improve anonymity and protect their identity, I collated participants data depending on their similarities. I interviewed three Canterbury-based arable farmers, here referred to as "Arable Grower" and they had a range of different farm sizes and crops. Each had a different approach to arable production, but they all supplied the same NZ based seed and chemical companies. I also interviewed two horticultural growers, here referred to here as "Fresh Product Grower", who grew various crops and had a variety of different business sizes. They came from across NZ and produced for both the domestic and export markets. The growers of different products in NZ often pay levies to corresponding product groups. These groups often lobby the NZEPA to keep chemistries within NZ, due to their perceived critical use. They also offer market access and crop protection services for growers. Subsequently, I interviewed three representatives from a number of these product groups, referred to here as "Product Group Representative". I also interviewed one "Researcher" with a background in wild pollinators, honeybees, seed production and chemical efficacy and represented a public research institute. Two representatives of NZ's regulatory groups charged with the assessment, reassessment and regulation of hazardous substances were also interviewed, and they are referred to here as "Regulator". Lastly, three representatives from NZ based seed growing organisations as well as seed and chemical retailers were interviewed, and I refer to them as "Seed and Chemical Company Representative". These individuals ranged from the technical field reps, company directors and crop monitoring coordinators.

The Participants' details came from publicly available records, like MPI's "listed beekeepers" (Ministry for Primary Industries, 2020), and the interviews were done face to face when possible, with three conducted over the phone. Before doing an interview, I contacted the participant via the phone introducing the study and inviting them to participate in an interview. If they verbally accept this invitation, I then forwarded an email that reintroduced the research and again asked them to join. This email (appendix A.5) contained a research information sheet (appendix A.2),

which explained the project in further detail, outlined the participants' involvement and included the measures taken to secure their anonymity and reduce personal risk. This email also invited participants to determine a suitable date and time which their interview could take place and assured them of their right to withdraw from the research. It also emphasised that participants' data would be treated in confidence and kept private. A set of open-ended questions guided the interviews which followed a semi-structured procedure to allow the interviewees to tell their own story and answer in their own words. All the interviews took place at the participants' workplaces accept for those conducted via telephone. After a personal introduction, I briefly described the purpose of the research and how their contribution would add to the final results. I then gave each of them another resource information sheet to look over and a confidentiality agreement to sign (see appendix A.5). To build some rapport before formally beginning the interview, I spoke to each participant about my interests in their work. Usually, this involved discussing my experience working and studying in NZ's agricultural industry. As agreed in the Lincoln University Human Ethics Committee approval (see appendix A.3), I spoke to each participant about the voluntary nature of the interviews and reinforced that they could withdraw at any stage.

Furthermore, issues about confidentiality and anonymity were addressed in the research information sheet and again on the consent form. In all cases, the interviews were less than one hour, they were recorded with each participant's consent, and each was later transcribed verbatim. As I gained a better understanding of how to draw out interesting information from the participants, I began to be more relaxed and followed the interview schedules more loosely, relying heavily on my ability to ask follow-up questions to gather information. This procedure was effective in getting participants to talk extensively about their activities and relationships, which was consistent with the concerns of this thesis. The interviews were particularly good at capturing the flavour of the participants material relationships, how they use intermediaries, and how they translate meaning.

Loose conversational style interviews can often stray away from their core focus (Guest, 1995, p. 137). Fortunately, I was able to refer back to the interview schedule, which served as a constant guide. However, I did find that when an interview was straying away, the most novel and interesting information emerged. Although each interview focussed on the same or similar schedules, they were all different. Some individuals had a heightened understanding of neonicotinoids technical use while others seemed very knowledgeable about the material relationships I am attempting to describe. In some cases, participants avoided discussing particular relationships presumably because of how it might reflect on them. While others, to my surprise, were very open about relationships that I thought would be highly sensitive. I sought to continue discussions about the relationships I had not heard about previously and decided to stop interviewing participants when I was able to answer my

research questions and was not hearing anything new. On completion of each interview, I emailed the participant thanking them for their time.

4.3 Organising the data

This section describes how I organised and interpreted the wealth of empirical data collected from the 16 in-depth interviews mentioned above. There is no clear direction or outline of how to arrange interview data in a case study, nor is there any direction given in the previous iterations of the CSSF (see section 3.4.1). Therefore a thematic analysis was applied to the interview data so that I could make sense of the wealth of information, present it in such a way that readers would get the most out of it and organise it so that I could then discuss it in terms of Latour's CSSF (Chapter 6). Furthermore, Braun and Clarke (2007) state that thematic analysis is the "foundational method for qualitative analysis" (p. 78). If anything, I chose this approach because I began to notice that several specific issues, themes, relationships and individuals were dominating the responses of interview participants from different actor groups.

Thematic analysis is among a cluster of qualitative methods which focus on identifying patterned meaning across a dataset. There are many different types of thematic analysis, but they all involve the process of carefully reading and re-reading the data to identify, analyse and interpret patterns (Braun and Clarke, 2006; Rice and Ezzy, 1999). However, different versions of thematic analysis vary considerably as their underlying philosophy and procedures for producing those themes vary. Braun and Clarke (2006) discuss there being two significant types of thematic analysis present in the literature. The more inductive approach, which is driven by the current data and the more deductive approach which is driven by assumptions of previous theories. Here, the inductive approach was adopted. I paid little or no attention to those issues or themes highlighted in previous research and have instead focussed on the themes dominating the raw data.

Because there are no qualitative studies which look at neonicotinoids as an agricultural practice in NZ, this approach was the obvious choice. The inductive approach is "a process of coding the data without trying to fit it into a pre-existing coding frame" (Braun and Clarke, 2006, p. 83). Therefore, the research findings will emerge from the themes, which are the most dominant in the raw data (Thomas, 2006, p. 238). This approach has allowed for themes to emerge from the data and allowed me to present the results in such a way that the CSSF (Latour, 1999a) can be used to make sense of it. It is important to note that Braun and Clarke (2006) described a six-step process to use when performing a thematic analysis. I have not adhered to this process, but I did utilise some procedures to identify key themes. The initial step of any qualitative analysis is the process of reading and rereading the transcripts, thus becoming familiar with the information. During this step, I noted down my early impressions. This step began before any interviews took place, during the initial

investigations of documentary evidence and continued as the interviews were undertaken and transcribed. After completing the interview transcriptions, some reading and re-reading allowed me to gain an overarching sense of the data. While doing this, I continued to note points relevant to understanding the embedding of neonicotinoids in NZ and more specifically, those material relationships frequently mentioned by interview participants.

Because I was concerned with discrete connections, material relations and the activities of those actors involved in the embedding of neonicotinoids in NZ, I did not use traditional coding techniques. Instead, you could say I performed open-coding because I did not have any pre-set codes, and I ended up with loose themes. A theme is "a pattern that captures something significant or interesting about the data and/or research question" (Maguire and Delahunt, 2017, p. 3356) and there are no rules about what makes a theme (Braun and Clarke, 2006). By looking back and forth between the data and the loops of Latour's CSSF (1999a), I sought out and developed potential themes based on the participants material relationships. These are: NZ is a small market for agricultural chemicals, Not enough tools in the toolbox, Market pressure, A disconnect between landowners and farming, Grass Grub and the Relationship between beekeepers and farmers. To illustrate these themes, quotations were extracted from the interviews and are presented in the following chapter.

To further address the research questions of this thesis and organise the data I utilised Latour's CSSF (1999a) as a methodology for identifying key documents and actors, representing the network, analysing the relationships which have made and maintained neonicotinoids as an agricultural practice in NZ and organising the empirical data. In other words, I used Latour's framework to follow the actors.

Hacking (1991) highlights that when looking at diagrams, we should ask if it aims to convey information or convince us that it is not to be challenged. Therefore, the CSSF (1999) should not be applied uncritically. Subsequently, I have tested this network representation in Chapter 6, adapting it for this thesis, while highlighting its benefits and shortcomings for understanding hazardous substance controversies.

As a thematic analysis is used to organise the data into themes based on interview participants material relationships, Latour's CSSF (1999a) is used to further refine the data. Where the thematic analysis has highlighted important relationships the CSSF is used to make sense of those relationships. Latour's five loops (1999a) directed my attention to a broader array of actors interested in neonicotinoids in NZ than other qualitative methods in the social sciences (such as snowball sampling) would have done. It has also allowed me to filter through the large number of actors present in the documents, focussing my attention on those that fit Latour's framework (1999a) and are involved in the vascularization of neonicotinoids in NZ. The CSSF has made me view

neonicotinoid use as an outcome of the interactions between the other loops (p. 108). This meant that once I identified initial actors I was directed to other actors relevant to this case-study by tracing their movements through material relations and translation operations. I did this rather than assuming the embedding and maintenance of neonicotinoids was down to a particularly powerful actor or the science alone.

These loops shape each other and are all equally important to the network (Latour, 1999a, p. 99) and, therefore, the making and maintenance of neonicotinoids as an agricultural practice in NZ. They also feedback onto themselves as well as all the others, forming interactive loops of material-semiotic relations (Latour, 1999a, p. 99). While the conceptual heart of any circulatory system is visible to the outside world, the rest of it remains hidden from view. Therefore, the CSSF helps to break open the neonicotinoid network, trace the relationships which make it up and therefore describe how neonicotinoids are embedded -and maintained- in NZ's agricultural practices.

4.4 Summary

This chapter has introduced the research design, data collection and analysis methods used to answer the research questions of this thesis. This thesis being a case study of neonicotinoids as an agricultural practice in NZ.

This research mainly uses data from 16 in-depth interviews performed with beekeepers, growers, regulators, seed and chemical company representatives, grower group representatives and researchers who are involved in NZ's arable sector as well as NZ's apples and pears sector. A variety of documents were utilised in conjunction with Latour's CSSF (1999a) to follow the actors, identify potential interview participants, decide whom to talk to, represent the network and analyse the relationships which have made and maintained neonicotinoids as an agricultural practice in NZ. Or in other words this methodology chapter has outlined how I followed the actors. These documents also played a small role in verifying some of that data collected through the in-depth interviews. I also drew on my personal relationships and implemented some snowball sampling techniques to continue to follow the actors and identify any further interview participants.

This chapter has also outlined how I utilised a thematic analysis to organise, make sense of and present my interview data. By merely reading, re-reading and referring to the material-relations, or loops which Latour points to in his CSSF (1999a), six themes emerged from the wealth of interview data I collected. Therefore the following chapter presents this thematic analysis of the interview data and draws on direct quotations to make inferences which act as clues for further investigation in Chapter 6.

This page intentionally left blank

Chapter 5 Results

5.1 Introduction

This chapter provides an overview of the findings of this research, which are derived from the 16 indepth interviews introduced in Chapter 4. The interview participants are all involved in NZ's agricultural industry and are similarly all interested in neonicotinoids. The interviewees include beekeepers, arable growers, fresh product growers, seed and chemical company representatives, researchers, regulators and product group representatives. The ultimate goal of the fieldwork was to examine the use of neonicotinoids in NZ's agricultural industry and uncover those relationships which contribute to its continued use. As explained in Chapter 4, I used the loops from within Latour's CSSF (1999a) to help me find interview participants and then identify the actors and relationships most commonly referred to during the interviews. This chapter will illustrate how it is these same actors and relationships which are subsequently embedding -and maintainingneonicotinoids in NZ's agricultural practices.

This chapter addresses the first question: Who and what are the major actors that influence the scale and frequency of neonicotinoid use in NZ, and what have been their roles? This is achieved by introducing those actors concerned with neonicotinoids and describing their relationships with the object of this study, and with one another. Subsequently, this chapter addresses the second question: How have certain relationships been made and maintained between actors? It is divided into six themes which have helped to organise my data. Each theme is based on interesting and common conversation topics that emerged during the interviews. This approach to organising my data also has the added benefit of allowing me to hone in on the interesting and more prominent actors and relationships contributing to the making and maintenance of neonicotinoids in NZ.

5.2 Emergent themes

The following themes have emerged from the interviews performed. Although each theme is generally associated with a question asked through the interviews, this is not strictly the case. Furthermore, these themes are organised according to their connection with each other. They are:

- 1. New Zealand is a small market for agricultural chemicals
- 2. Not enough tools in the toolbox
- 3. Market pressure
- 4. A disconnect between landowners and farming
- 5. Grass Grub
- 6. Relationship between beekeepers and farmers

5.2.1 Theme 1: New Zealand is a small market for agricultural chemicals

A persistent pattern throughout the interview process was the mention of NZs size as a market on a global scale. This pattern was frequently mentioned in regard to the availability of agricultural chemicals for NZ's primary producing sectors.

But we're a small country, we've got four and a half million people. We're trying to play on a global Marketplace. You know we're never going to be able to get everything. Right? (Product Group Representative 1)

For new chemistries or new methods of application to be imported, sold and used in NZ, it must first be approved by the NZEPA under the HSNO act. It is also the responsibility of the applicant to provide all the relevant information, e.g. environmental toxicity studies, to the NZEPA and cover the costs of the application process. Several participants reported that it is multinational crop protection companies, like Bayer or Syngenta, who usually apply for hazardous substance approval in NZ.

> In terms of things like new actives [chemistries]. I think recently I guess the big players are people like Bayer, Dow [chemical company], BASF, Syngenta those are kind of the big chemical companies. They tend to have someone who will manage the kind of regulatory activities in New Zealand. (Regulator 1)

Participants relayed that there are high costs associated with applying for new chemistry approval in NZ and that it is a prolonged process.

The highest kind of fee for the most in-depth assessment that we do is something like \$25,000. (Regulator 1)

I think we aim to recover something like about 10 to 12 per cent. Of costs through application fees something like that. (Regulator 1)

The pipeline for you know registration can be three to five years in terms of Trials and registration data packages getting together and then being assessed. (Regulator 2)

Horticulture in NZ is based on crops considered minor internationally, and our agricultural commodities similarly represent a fraction of total global production. When international crop protection organisations are making decisions about where to and what to invest in they take into account volumes of sales, transport and regulatory barriers. Subsequently, this results in very few situations where they are applying for new chemical registrations in NZ.

So because it's a high cost but small market, the economies of scale just don't work and bear in mind that a lot of products are quite crop-specific. So if you had something that was particularly good for onions or carrots, but only one of those crops and that market isn't sufficiently big enough to justify the importation of that product from a supplier's point of view. Yeah, so there's a lot of commercial restrictions anyway. (Product Group Representative 2)

The cost associated with the NZEPA applications process, together with the NZ market representing a fraction of those organisations' global sales means that they are not interested in paying for the new chemistries to be tested and then introduced to NZ.

The players are only getting bigger, and they are not interested in spending \$150,000 on a product before they can even sell a little bit unless they know that the market is there for it. (Arable Grower 1)

The economic incentive for those companies to introduce new products to NZ is not there and therefore the availability of crop-specific, pest-specific and softer chemistries is much less than elsewhere in the world. The following comments provide some examples:

> So there's a registration process which is quite tedious, and it's quite expensive as well. So a lot of the chemical companies see New Zealand as a small market. So to register that particular chemical in New Zealand actually costs quite a bit of money and quite a bit of effort and the return on investment as far as they're concerned is probably not that big. So, we miss out on a lot of chemicals that are actually available overseas but not available here. (Fresh Product Grower 2)

Several underlying ideas have come through from such comments. Firstly that multinational crop protection companies are uninterested in replacing a product if the return on investment is not there. As explained above, this is generally due to the small size of NZ's market and the high cost of the application process. However, they are also uninterested in introducing new chemistries to markets where alternative chemicals that have the same or similar efficacy already exist.

If they sell product A and it sells but product B is better but gives them a similar margin, why would we replace it? What's the interest? (Arable Grower 1)

Furthermore, when registering an agricultural chemical, the methods of application and the crops to which it will be applied must be outlined in detail. Subsequently, when a crop is lacking in hectarage, it is unlikely that a multinational crop protection company will register a chemical for it because the application costs likely outweigh the return they gain from chemical sales.

Crop protection companies tend to register on, you know, larger crops with a higher return that will use more product. So, you know, they'll always register on wheat, barley and pip fruit and kiwifruit and potatoes and onions, but they sort of don't because the return on investment is not there, you know, register a claim for leeks or for leafy vegetables or for asparagus or for, you know, clean greenhouse tomatoes because simply the number hectares aren't there and there's not that drive, so it's very difficult for those growers to access new chemistry. (Regulator 2) Lastly, those organisations who bear the cost of registering a new product or method in NZ, have not been guaranteed a competitive advantage over other agricultural chemical suppliers. Essentially, the HSNO and ACVM Acts (see section 2.3), protect confidential information given in support of applications for five years. HortNZ outlined that this period of data protection is not long enough to provide the registering organisation with a sufficient economic incentive to register new products across a wide range of crops (Ministry for Primary Industries, 2012; Chapman, 2016). Currently, generic companies can copy the product label and undercut the registrant once the data protection period ends. Subsequently, the registrants often do not have enough time to ensure a return on investment before this happens.

In some cases, the registrant is unable to recoup the costs associated with the NZEPA registration process (Chapman, 2016, p. 4). Furthermore, NZ's data protection regime is not currently in line with our major trading partners (Ministry for Primary Industries, 2012, p. 8). Subsequently, multinational crop protection companies are more likely to invest in countries where the volume of sales is higher, the logistical costs are lower, and the regulatory barriers are either not there or not significant. HortNZ says that there is a need to amend laws to increase data protection. They claim that this will mean that "companies will be more willing to register compounds for smaller crops which currently do not provide a return on investment for the registration process" (Chapman, 2016, p. 4).

So what does this mean for neonicotinoids? Well, because of the registration process and the lack of incentives for multinational crop protection organisations to register new chemistries in NZ, there remains a lack of more modern, softer alternatives across the industry. Growers, product group representatives and regulators all refer to this as there not being enough tools in the toolbox.

5.2.2 Theme 2: Not enough tools in the toolbox

One of the most dominant patterns throughout this research was that the participants all believed that there was a lack of agricultural chemicals available in NZ or, not enough tools in the toolbox.

I think our belief is that we need all the tools in the toolbox and the ones we've got we need to use responsibly. (Arable Grower 2)

It was also clear that growers blamed the NZEPA, the regulatory body charged with assessing and reviewing hazardous substances, for the lack of tools available.

Chemicals are constantly being dropped from the list but not many, there's not many new chemistries, nothing much being added on. (Fresh Product Grower 2)

You need more tools you need more chemicals because at the moment, you know like there are quite a few chemicals that we can only use it once or twice right, you know, you know that you can't keep using it because otherwise, it builds up resistance, so you need different chemistry throughout the year. So you need to, so you need more chemicals. (Fresh Product Grower 2)

I mean ultimately they are always reviewing chemicals and taking them away from us. (Arable Grower 3)

The majority of individuals understood that there is a need for chemicals like neonicotinoids in NZ, mainly when there are no alternatives readily available, or when the available options are much worse for the environment and human health. Many also expressed how paramount a situation is where particular chemicals are taken away, particularly for smaller product groups who do not have the means to find alternatives.

You know horticulture is a shitty thing and if you've got one particular beast that you can only control with a particular chemical, and you're told that you can't use it. It can be a stressful situation, so especially for smaller growers that don't have the, you know the resources to be able to try and find alternative products so it can be a really big thing. (Product Group Representative 1)

Throughout the interviews, participants consistently mentioned that there is room to get rid of some chemicals and, particularly from the growers that they would like to use softer options and non-chemical alternatives. However, it is made very clear from the following comment that this is only true when they do not face any personal risk.

Would like to use more environmentally friendly/biological/whatever products. The problem is it's really hard to make ends meet on a farm. That you can't have too many cock-ups, nature will provide cock-ups for you. So you got to get everything right. Yeah. And so the last thing you want to be doing is creating a cock-up yourself that wasn't needed to by not using a product that can give you some reliability. (Arable Grower 3)

In some cases where there are little or no alternatives, the growers and their representatives will put forward a case to the NZEPA that claims a particular chemical is of critical use. As explained below, a critical use chemical is necessary for the viability of a specific crop, and this can come down to a lack of alternatives.

> Yeah, that's an unfortunate situation in New Zealand that we're trying to. I say we, the product groups of New Zealand are trying to work with the EPA and help them to understand that, if they wanted to do a reassessment program if there was plenty of other alternatives and options in the toolbox then that would great, we wouldn't need to defend. We could say actually we have ten other alternatives and actually that products not being used and in some cases, that's what we did. That's what growers did say but for a number of chemistries a number of actives you know they were critical use because there was just no alternative. (Product Group Representative 3)

The NZEPA was consistently blamed by participants, for the lack of alternative pest controls available in NZ. It was expressed that the government did not care about, or consider the implications that removing and not replacing chemistries would have on NZs primary industry.

Governments will just say you can't use it. Not really understanding the absolute consequence. It might seem good on paper, but the reality will not be good because you'll have such devastation in terms of crops. (Product Group Representative 2)

A consequence of this lack of tools in the toolbox, resulting from NZ's small agricultural chemical market and its slow regulatory system (see section 5.2.1), is that growers are calling for improved research into alternatives.

Yeah, and like I say research should be going in to let's give us some other alternatives that will be reliable and actually work because I think you'll probably find that a lot of farmers would like that and I'd fall in this category. (Arable Grower 3)

Growers have explained that the lack of alternatives is an industry issue and that the introduction of

new chemistries should be demand-driven and not left up to chemical companies and the NZEPA.

So in many ways in that sense. Instead of chemical companies driving new products into the market introduction could be should be demand-driven. And it's actually the users. But were not organised. Could have a body that requests introduction for the bringing in of products. (Arable Grower 1)

They appear to be tired of waiting for international chemical companies to introduce alternative crop protection methods, which may or may not be useful for NZ's specific purposes. Ironically, those product groups, to which growers pay a levy in exchange for research and marketing, admit that they are not very good at finding new chemicals or biocontrol agents.

I think we're probably not that good or EPA is not that good, and the MPI aren't that good. At promoting or investigating and researching alternative products. (Product Group Representative 1)

Many interviewees agreed that NZ's agricultural communities are "putting the blinkers on" (Arable Grower 1), outwardly ignoring or lacking the foresight to see how global trends and market pressure might affect the future use of those chemicals available today. Furthermore, many agreed that "necessity is the mother of invention" (Product Group Representative 1) when it comes to changing agricultural practices and particularly crop protection methods.

Until there's somebody who tells you, you just absolutely can't do something. Then you don't look for another alternative. (Product Group Representative 1) However, this was not unanimous. Some interviewees insisted that it is NZ's agricultural industry leading the charge, in terms of seeking new and reliable alternatives.

They have been successful at getting a PGP [Primary Growth Partnership] project, which is 26 million over seven years and the whole basis of that is to be proactive about the long-term use of agri-compounds and how we can diminish the use. (Product Group Representative 2)

One interviewee had an interesting take on why chemistries are not being replaced at the same rate as they are being removed. They explained that modern agricultural chemicals are formulated and trialled in such a way that there are significant costs and long periods involved. This means that to find a softer compound with equal or better efficacy on a target species is very difficult, let alone replacing a broad-spectrum chemical which is efficacious on many target species.

> Yeah, so I mean, of course, you know there's always research just saying you know, what's bad so, over time, you know more and more those harsher ones, you know like DDT and so on are gone and but they're not be replaced at the same rate. A lot of new chemistry nowadays are quite targeted. So it's quite hard to find something specific that actually targets a particular pest. (Fresh Product Grower 2)

Another consequence of this theme is the proliferation of the off-label use of chemicals in smaller crops. As shown in the following quotations, due to the cost of registration, many products in New Zealand do not carry label claims for all the crops they are used on, therefore for smaller crops (by value and area), chemicals must be used off-label. Essentially smaller crops in NZ do not have many if any, on-label tools in their toolbox. Subsequently, they rely on off-label use, which is where a registered chemical is used in a manner that is not specified on the product label.

Well, there's less available here. What's helping the industry here is this. It's not a loophole it's a thing in legislation that you can actually use them on minor crops because we know that if it's not usable on cereals the chances of getting in here is absolutely zilch. So you've got too actually, you're using chemicals for the seed production of vegetables but off-label because there's not a big enough market to actually justify the companies registering it. Long may it last because otherwise, we won't have any chemicals. (Arable Grower 2)

Those growers using chemicals off-label do not have access to label directions and NZ good agricultural practice (NZ GAP) advice, and therefore, the off-label use of agricultural chemicals ultimately leads to misuse (Chapman, 2016, p. 4).

Let's say a mass of bees are killed somewhere, and it's proven that it's because of neonicotinoids it's more likely if not a hundred per cent because of a misuse of products in an off-label situation than because of an on-label use. (Arable Grower 1) One reason why chemicals are misused in New Zealand is that the portfolio available is relatively narrow; it's very expensive to register a chemical [for smaller crops]. (Arable Grower 1)

Another consequence of the lack of tools in the toolbox is that growers of more diminutive crops and their subsequent product groups, fight to keep those broad-spectrum chemistries, like neonicotinoids, which they already have access to. This fight is likely exacerbated by the fact that several NZ and offshore customers do not accept products if off-label compounds have been used in its production (Chapman, 2016, p. 4).

Yea so I help them [growers] with that [protecting available chemistries] as well and put the benefit case forward to EPA about critical uses. Yes, of course, there's a need because otherwise, they have nothing else. (Product Group representative 3)

These benefit cases usually present information regarding the crop protection services which a chemical provides growers and the current lack of alternatives. When performing a reassessment of a hazardous substance, the NZEPA will compare these benefit cases with any Ecotoxicity data and environmental risk information they have access to. (Regulator 1).

Interestingly the beekeepers interviewed believed that it did not matter what crop protection method was the usual agricultural practice as they all serve the same purpose, to kill insects.

If it's an organic that kills insects, it's still the same end product for us. You're still killing insects. (Beekeeper 1)

Because the problem is, if you get rid of neonicotinoids, what are you going to replace them with? I'm more scared of what you're going to replace them with. (Beekeeper 2)

Because of NZs small size as an agricultural chemical market, the small size of its agricultural production and its slow hazardous substance assessment process (theme 1) there are not many tools in the growers' toolboxes of crop protection methods (theme 2). This lack of tools is exacerbated by the governments and some product groups inadequate research of alternatives and inability to foresee market trends. Growers and their representatives are therefore made to fight to keep what broad-spectrum chemicals, like neonicotinoids, they already have available to them. Interestingly in a situation where new chemicals were regularly introduced to NZ, it is unlikely that agricultural practices would change.

We're unique in that we export for those crops that do export into the different markets and often, you'll be aware, often we have to meet some niche requirements in terms of MRLs [Maximum Residue Levels] to access those markets and so as well as that, you know if there's not a, if there's a new product, a new chemistry and there's not a global MRL set. You know, you just can't use it you are reliant on that older chemistry if there's an MRL and it controls the pest and disease so you can continue to produce. (Regulator 2)

Therefore, as theme 1, New Zealand is a small market for agricultural chemicals, related to this theme, not enough tools in the toolbox, this theme relates to the next, pressure from domestic and international markets.

5.2.3 Theme 3: Market pressure

Ultimately, NZ's primary sectors are answerable to the consumer. Consumer movements focussed on environmental and human health issues are leading to more stringent environmental standards and an industry-wide shift towards strategies that link environmental performance with consumer expectations (Dalziel et al., 2018).

> Consumers globally are demanding higher standards for building trust that food is safe, healthy and good to eat, including requirements for stronger systems of food fraud detection and food traceability, and that production systems meet ethical expectations for responsible innovation, including high animal welfare (Dalziel et al., 2018).

There was an acknowledgement amongst the interviewees of their role in the agricultural supply chain. Each of them recognized that they were responding to the action of an international market.

So we'll look at the chemical attributes of each different chemical and then we'll test that against regulatory requirements or regulatory concerns in terms of Market access getting into countries primarily around MRLs, or maximum residue levels, so we as growers we need to achieve MRL levels under a regulatory approach but our growers who've actually got relationships through marketing and sales with customers quite often. A lot of our customers have a lot lower MRL levels than the regulatory requirements. (Product Group Representative 1)

But then ultimately at the end of the day even though we are right at the bottom of the chain and we're just producing, you know wheat and barley that's made into beer or whatever. We still are answerable to the consumers to a certain extent. (Arable Grower 3)

Interestingly, the things which each international market acts through appears to change depending on the agricultural commodity which they are demanding and on the NZ based agricultural sector they are demanding it from. To this end, two different stories have emerged from the interview data, which demonstrate how the presence and absence of maximum residue levels (MRLs) are resulting in very different outcomes in terms of agricultural practices. The first story features those participants involved in the production, marketing, sale and regulation of apples and pears. Here, consumers do not want toxic chemicals, like neonicotinoids, used in the production of their fruit.

A lot of our markets particularly, UK and Europe. Have been going through that transition process away from certain types of chemistry. And in Europe,

in particular, the neonicotinoids are not favoured by many of our customers in those regions. So growers [in NZ] have tended to move away from them and look for alternative options. (Product Group Representative 1)

I guess the other one would be a lot of our exporters who are looking to move product into market to customers where there's not a huge fondness for neonicotinoids. So that's probably the big ones as well. (Product Group Representative 1)

This consumer demand is turned into action through MRLs of chemicals allowed on fresh products. Essentially, domestic and international markets are demanding that regulatory and self-imposed MRL standards are met. MRLs are "the highest acceptable level of a specific agricultural chemical residue (including veterinary medicines) allowed in food" (Ministry for Primary Industries, 2019) and are set by international and domestic regulators as well as supermarkets and individual customers. The following quotations, therefore, demonstrate that MRLs are often determining NZ's agricultural practices.

> For our guys, our volume we're about 85 per cent export and our main markets at the moment are the States, Asia, UK and Europe. I find UK and Europe is by far the most stringent in terms of customer requirements driven by the likes of the larger Supermarket chains, like Aldi and that sort of stuff and the UK Marks and Spencer and stuff like that have had high requirements for a long time. We're finding now that other markets, particularly in Asia, are starting to pick up those MRLs as well. So that's becoming our main driver really, so we find that we have different market requirements for different companies and different countries, but also different companies within countries. We have growers that are growing for particular markets, and that might be based around varieties or sizes, and then they'll have Agri-chemical MRLs by country by grower, by customer, so it's all driven around that. (Product Group Representative 1)

So, for instance, things like there are certain fungicides and insecticides the regulatory levels might be at such a level but our customers will actually they'll actually require 0 MRL levels altogether. (Product Group Representative 1)

Yeah, then obviously the retailer part is undertaken by the retailers because as far as they're concerned, they want to manage and control their own process. Yeah and interestingly enough when you take the hundreds of thousands of tons of product and I'm talking fresh every year we would get probably I'm just saying probably because it's roundabout half a dozen and no more than 10 situations where it would be over the MRL level but bear in mind the MRL level is so much lower than the global level. (Product Group Representative 2)

This results in an interesting dilemma for the apples and pears sector where:

Consumers are saying we want fresh fruit and vegetables every day on our shelves, but we don't want you to use chemicals. We're restricting the use of water and it's becoming strangled from a regulatory point of view. But the same consumer's saying oh but I want fresh fruit and vegetables every day so. That's the dilemma we have as an industry. (Product Group Representative 2)

When you're relying on 100% of your consumables based on the vagaries of weather and disease and pest then that becomes a lot more complex. So that's why the market has become what it has from an Agri compound point of view for suppliers and growers. But the industry has taken a very proactive view in saying 'well despite that. Long term we have to find how we can use softer options or BCAs [bio control agents] or other options' and that's been driven by industry, not the government. (Product Group Representative 2)

Thus, the apples and pears sector no longer applies neonicotinoids to their orchards. Furthermore, Apples and Pears NZ no longer put neonicotinoids in their crop protection programmes because consumers do not want neonicotinoids or other harmful compounds on their fresh products. Subsequently, the high-value markets which this sector exports to, turns this consumer demand into action through stricter MRLs and in some cases, zero-tolerance standards. Essentially, NZ fresh product growers must comply with MRL regulations to gain market access, which often ends up meaning they cannot use chemicals like neonicotinoids.

The second story revolves around the arable sector, and like the apples and pears sector, arable growers, product groups, regulators and exporters are all responding to consumer expectations and the subsequent markets they enact.

So our customers drive a lot of what we do, were very, especially in particularly one company, so it's very. Europe drives, what's happening in government, regulatory based in Europe drives a lot of what we do. (Seed and Chemical Company Representative 3)

It's not just our government that regulates. We're mainly an export company so, what's happening with the European Union has a huge effect to play with also what happens here. (Seed and Chemical Company Representative 3)

However, this story is vastly different when compared to the first one because MRLs are absent from seed production in NZ.

As I say, a lot of our stuff's not food grade, so we don't operate within an, within generally the label grade rules if that makes sense. (Seed and Chemical Company Representative 3)

MRLs do not exist in the arable sector because the majority of the product is not food grade. In the apples and pears sector, the product being scrutinised with MRL regulation is from the generation which the final consumer eats. Whereas the arable sector is "actually one generation of seed away from the consumer because we produce seed" (Arable Grower 2).

So, rather than the consumers demanding less or no chemicals be used, which is then turned into action by the international markets via MRLs. In the arable sector, consumers expect high yielding and high-quality crops, which is then turned into action by offering more money to those who meet these expectations.

They'll breed a new stock a new line, and they'll send it to New Zealand to be grown out. They might only get 50 kilos out of a glass house that they've grown over there then they send that here, and we can multiply that 50 kilos into you know five-ton for them and then keep doing that. So that's because we can grow stuff so clean because we're free of um weeds and other types of grasses and that side of things so that's why it appeals for companies to pay a bit extra because we're probably the dearest country to grow stuff in compared to other places around the world, but it's because we can produce, purity levels that are a hundred per cent. (Seed and Chemical Company Representative 1)

It is the NZ based seed companies who hold relationships with both the international market and the NZ growers. It is also these seed companies who subsequently translate the markets action to the NZ growers via irrigation, fertiliser and crop protection programmes. As explained below, NZ based seed companies gain the contracts of the international seed market not only because NZ has access to land and water, is free from pests and weeds and has a longitude perfect for northern hemisphere offseason seed production, but because NZ has access to agricultural chemicals banned elsewhere.

A lot of why we get a lot off those European seed production is because we have, dare I say it, more relaxed rules [hazardous substance regulations]. For instance, keeping our carrot seed crop alive for 15 months can become an issue, Europe with more regulations and limited chemistries available, things become harder. (Seed and Chemical Company Representative 3)

The above quotations describe the relationship which exists between international seed companies, NZ seed and chemical companies and then NZ arable growers. Essentially an international seed company will send a NZ based seed company a small amount of new hybrid seed to be multiplied and grown out in NZ, e.g. turn 50 kilograms into 5 tonnes. NZ is chosen for several reasons including access to water, high yields, freedom from pests and weeds, a longitude perfect for northern hemisphere offseason seed production and availability of specific agricultural chemicals, like neonicotinoids. The international seed company expects that the resulting crop will be of very high quality, free of impurities and have a high yield. To ensure this occurs the NZ seed company, who is charged with finding the farmers and land to multiply the seed, design irrigation, fertiliser and crop protection programmes. These programmes are written and performed in such a way that they minimise the vagaries of growing outdoors. The following quotation explains the importance of agricultural chemicals in guaranteeing market access and maintaining the relationship with international seed companies.

Ah if they banned all chemical, we'll be devastated. You wouldn't be able to produce what like we're producing. Ah most of our stuffs going to export so you just wouldn't be able to produce it to send it anywhere because it would be full of weeds to start with so that's the first thing will be thrown out, and you wouldn't be able to get, grow it with you know the fungicide side of things, the insecticides it just wouldn't work. (Seed and Chemical Company Representative 1)

This relationship between international seed companies and NZ seed companies, as well as the expectations of purity and yield, will likely mean that neonicotinoids remain a central part of NZs arable crop protection programmes. Furthermore, it has already resulted in some interviewees struggling to purchase seed which has not been treated with neonicotinoids.

Often I ring up asking for just the seed without the insecticide, and I can't have it. It's like you must have that, and then it's another six or seven hundred dollars a ton. So it's another major cost so instead of being a thousand dollars a tonne it's like 15, 1600 dollars a tonne. So it's like half as much again for the insecticide. You don't really want. Particularly in springtime on crops like so that's a bugbear I find I don't like being. The end of the day. We're a consumer and where else does the customer get told no, this is what you'll have. (Arable Grower 3)

It is fascinating that NZ's arable sector is not subject to the same amount of scrutiny as the apples and pears sector. Where in the apples and pears sector, the presence of MRLs has meant neonicotinoids are no longer used, in the arable sector the absence of MRLs is helping to embed neonicotinoids as an agricultural practice.

The arable sector has an additional generation between the use of neonicotinoids and the end consumer, so the demands of those consumers and the global trends around neonicotinoids are not being turned into action through MRLs. Instead, those consumers purchasing the generation of seed produced in NZ are having their demands of high purity and yield turned into action. This is then being translated by NZ seed and chemical companies as irrigation, fertiliser and crop protection programmes, which aim to minimise the risk of any crop failure.

These NZ seed and chemical companies manage the seed multiplication process from start to finish, allowing for minimal input from the growers whose land is being used. Subsequently, the mounting pressure from international and domestic markets (Theme 3) is driving the disconnect between landowners and agricultural practices in the arable sector (Theme 4).

5.2.4 Theme 4: Disconnect between landowners and agricultural practices

Let's be honest like farming now. We are accountable to the consumer way more than what we used to be and people now have the ability to see things right or wrong very easily. And so yeah, I think somehow we've got to, and you know the same old Story the gap between the urban and the rural is getting wider. (Arable Grower 3)

As above, many interviewees highlighted both the growing accountability of farmers to the end consumer and the ever-increasing urban-rural divide. However, the interviews often focussed on a lesser-known and much more exciting theme. This theme is the idea that there is a growing disconnect between NZ landowners and the agricultural practices occurring on their property. As seen in the following quotation, this disconnect is created in situations where landowners are not involved in the day to day decision making and farming duties.

I mean, we've got a neighbour we do harvest for, and he does absolutely nothing. He owns no machinery, relies entirely on reps and does what he's told to a certain extent. (Arable Grower 3)

This disconnect was highlighted in the interviews by the relationships between landowners, seed company reps and contractors.

Well some seed reps basically write up the recommendations to apply the chemical and then give that to a contractor, the contractor then comes and does it, and the guy who owns a place hardly has any input at all. (Arable Grower 3)

It was particularly apparent in the arable sector, explicitly surrounding the multiplication of hybrid seeds. Some interviewees even went as far as to suggest that "as far as the country goes there will be more relying on reps, than probably doing it themselves" (Arable Grower 3). This reliance on reps suggests that increasingly growers are relying on the representatives of various organisations to decide, direct and perform the daily farming practices on their property.

So we go out find the ground um, get the growers to sign contracts to grow the crop and we manage it from start to finish. (Seed and Chemical Company Representative 1)

As shown in the following quotations the practices which growers are increasingly reliant on reps for include the application of agricultural compounds, irrigation, planting and harvesting.

So we're mainly on spray and fertiliser and also the likes of cutting times, harvest times and all that sort of thing. (Seed and Chemical Company Representative 1)

Several ideas stemming from this theme reinforce why particular agrichemicals, including seed treatment neonicotinoids, are being used in NZ. It was suggested that at times the reps are there to "pedal their chemicals" (Arable Grower 3) and they aren't giving good advice to the farmers.

They pedal already treated seed like that and I know from my time in the seed company that they make a good margin on it, but it's almost like they do that to say, oh then you don't have to worry about grub or aphids well

that's not the case like you've still got to monitor it anyway. (Arable Grower 3)

The problem is too, and I've been in this game, and the seed companies make a truckload out of selling it, so they put it on because they want to sell it. (Arable Grower 3)

Many participants have also suggested that reps recommendations are not always well informed. This can be due to many reasons, but the most apparent one was that the reps are not always experienced enough to be making informed crop protection decisions. As explained below, this comes down to a high staff turnover rate, a lack of technical knowledge and experience.

> Often reps for the seed companies come from a family farm and end up going back to the family farm. So the reality is, is that they do turn over. That's just the nature of the beast. I mean, look at me. I was one, I was an agronomist, and then I came and went in a couple of years. (Arable Grower 3)

I think many of them are just nut cases they don't, they are not sufficiently trained. There are very good men that know their products, but people are more hired because of their skill to get along with a wide variety of people instead of because of their knowledge. (Arable Grower 1)

This begs the questions why are there even reps at all and if they are not making informed decisions based on experience and technical knowledge, are they merely salesmen? Interestingly one participant explained that one of NZ's leading agricultural service companies no longer has field reps.

They've just come through a big thing where they don't have field reps anymore. The woman decided that they're a waste of time. They're just out drinking coffee all day, you know paying reps x amount of money to go and deliver products or chemicals wasn't cost-effective. So she got rid of them all. (Fresh Product Grower 1)

This, supported by the following quotations, also suggests that reps are not making crop protection decisions based on field observations. Essentially, they are not designing crop protection programmes and making decisions about when to spray based on active and in-depth crop monitoring.

So it's not until they actually, something happens to them when they think oh crop monitoring, crop monitoring you know. But until something destroys one of their paddocks, that's when they think right we've got to get crop monitoring. Yes do something about this, and it's also good for the rep to see that too because yea they think wow crop monitoring, because they were just looking at the paddock, but they wouldn't have been doing the job that we do, yea and now they see that within our business what good it does. But it just takes something like that [crop failure] to wake them up and realise that yea, the value of our service. (Seed and Chemical Company Representative 2) It was suggested that in cases when crop monitoring did occur, farmers were finding that they were frequently applying agricultural chemicals to crops that did not require it.

I have a particular grower that when we started [crop monitoring] he actually saved money on sprays because the TPP [tomato potato psyllid] wasn't there, he was just spraying every 10 days regardless. That's what the rep said, spray every 10 days. And then he saw the value with what we were doing that there was no TPP out there or no aphids out there at that time, why am I wasting this money on spraying. (Seed and Chemical Company Representative 2)

The above quotations discuss a crop monitoring service offered by another one of NZ's largest seed, chemical and agricultural service companies. It was implied that even though this service is available, it is hard to get farmers on board. The lack of farmer engagement with this program is not due to the value which it provides farmers, but the relationships that exist within this agricultural service company, the relationship that farmers have with the rep and the disconnect which landowners have with farming practices. Subsequently, despite the existence of a good crop monitoring programme that reduces chemical usage and saves land-owners money, agricultural chemicals like neonicotinoids are still used.

It is easy to refer back to theme 3 (pressure from international and domestic markets) in a situation where a landowner is disconnected from the daily agricultural practices occurring on their property. Because, as shown in the following quotation, the NZ seed company reps are basing their decisions on the expectations of their international customers.

It's been relayed to me that the overseas companies said they do not want the excuse of the weather or anything as to why there's not a crop. (Arable Grower 2)

This results in the NZ seed company rep doing everything in their power to control the vagaries of growing outdoors, which might impact the yield and purity of a crop.

There's a big disconnect of that actually coming in hitting it with. If there was an aphid there. They'll kill it. Hmm. Okay. Yeah, that's the whole attitude if there's a bit of fungus. They'll kill it. (Arable Grower 2)

Normal practice is the company rep comes in and says you've got to spray this tomorrow and they just blindly do it. (Arable Grower 2)

What results from this relationship between the NZ seed rep and the international customer is that the NZ seed rep becomes unwilling to give up decision-making responsibilities to the landowner. During this research, a seed company rep was asked about their relationships with farmers. They subsequently described an example where a farmer was involved in the decision-making process. For instance, we're running some trial at the moment where the farmer is heavily involved in suggesting treatments so I would say our relationship is very good with landowners. (Seed and Chemical Company Representative 3)

The rep then went onto explain that grower and landowner involvement in the decision-making process "doesn't happen very often" (Seed and Chemical Company Representative 3). Not only are the seed and chemical company reps unwilling to give up control to the landowner, but often the landowner is not knowledgeable enough or too busy to be making the crop protection decisions.

I think a lot of other farmers will just. The company rep will just say this and do that, and they have always sprayed an aphicide on the wheat in September or August or something. I'll just do it. It's just part of their spraying of weeds spray I'll put it on. Yeah, there's no, no thought about anything else. There might before, but often farmers might not have the knowledge or the do. You know, they focus on other things. (Arable Grower 2)

Well, most of the information comes from the rep. Farmers are often following what they're told. They don't really know anything they go well the label says this, and they don't actually know. Their depth of knowledge is really low. Not all farmers. I've got some farmers who are amazing, but let's be honest they're probably the 1-2% the rest of them know basically nothing. They rely heavily on the rep whatever the rep says they do, or the firm they're growing for. Who normally says get the chemical rep in. (Beekeeper 2)

Another idea that stems from this theme and likely contributes to the embedding of neonicotinoids also relates to theme 3. Due to the expectations of their customers to receive a pure and high yielding crop, the NZ seed company reps are hesitant to try new or alternative crop protection methods. The following quotation explains that seed company reps are primarily focussed on the success of a crop, which ultimately means that they rely on the older chemistries, like neonicotinoids, that they know will work.

> Their [seed reps] emphasis is on focused on the one crop they're dealing with, and they've got to actually get a crop out of it at the other end. They're not. They are more cautious about trying new stuff most of my discussion about using chemicals is with reps saying 'just use it, you just put an OP [organophosphate] or something like that on' and it's because they know it works. But they don't actually understand the implications, and they generally have got a very closed mind. (Arable Farmer 2)

So NZ seed and chemical company reps are translating international markets to growers in such a way that neonicotinoids are made out to be necessary to both ensure crop yield and market access. This has meant that neonicotinoids are heavily embedded in NZ's arable sector mainly due to it being those seed and chemical companies who both supply the seed to the grower for multiplication and hold the relationship with the international customer. This allows those reps to control the process

of multiplying a seed crop from start to finish and therefore, the growers or the landowners themselves are disconnected from the daily farming practices occurring on their farm.

5.2.5 Theme 5: Grass Grub

Particular chemicals are necessary for the types of agricultural systems being implemented in NZ, and this can vary depending on the industry. Interestingly, one of the regulators interviewed stated that the difference between NZ, Europe and North America in terms of neonicotinoid regulation comes down to their different agricultural landscapes.

It's difficult to end, also we've got quite a different agricultural landscape. The majority of like, you know, we've got small amounts of crops, cereals and vegetable crops and then we've got a huge area of pasture and we're using neonicotinoids on pasture, but obviously not the whole, we don't grow pasture for the whole area every year. It's like just small patches where there's pasture renewal or changing use to pasture so and that compared with say the US or Europe where they're growing lots of cereal crops lots of maize and corn and using maize as an animal feed. And they plant those every year, the seed treatment. Well, they have been until recently. So yeah, it's quite a different kind of yep quite different quantities being used in different ways. (Regulator 1)

As well as the pastoral nature of NZ's agricultural landscape, some agricultural pests which make up this landscape are also acting to make NZ different in terms of neonicotinoid use. Throughout this research, there was a constant mention of an endemic beetle species or pasture pest, known as grass grub (Costelytra zealandica). This inconspicuous insect was often at the centre of conversations when discussing the use of neonicotinoids in the arable sector. Grass grub appears to be the driving force behind the use of neonicotinoids in the arable and pastoral sectors. Arable growers plant neonicotinoid-treated seeds "as a tool for grass grub control in establishing grass and cereals" (Arable Grower 2) and "also as an aphid control in early establishment" (Arable Grower 3).

There is one alternative chemical grass grub control currently available in NZ. It is a highly potent organophosphate (OP) called diazinon. Interviewees described it as the only way to protect their crops from grass grub during early establishment besides neonicotinoids.

So 90 per cent of that stuff [seeds] will be treated [neonicotinoids], yea and if not they'll be drilling diazinon or something like that. (Seed and Chemical Company Representative 1)

All though neonicotinoids are the more widely used method for controlling grass grub, diazinon offers a safety net for growers. However, in 2012 the NZEPA reassessed diazinon determining that it would be banned. After considering several benefit cases like those described in theme 2, the NZEPA implemented a 15-year extension on diazinon use to help landowners control grass grub. This extension was to enable industry to find another crop protection method besides neonicotinoids and

diazinon. However, there has not been a push for people to look for an alternative (Product Group Representative 3). Many participants expressed that neonicotinoids, as a seed treatment, is the most environmentally friendly way to combat grass grub. Furthermore, they explained that the compounding effects of pest pressure from grass grub, the toxicity of diazinon, and it coming to an end, means that they have no option but to rely on neonicotinoids.

> We have very lacking in company apart from liquid diazinon sprayed on top, which is horrible for the environment kills all the worms. And that one's just about come to an end anyway. There's no get out of jails. And so if you don't get it right up front, you're cooked, and then there's sweet all you can do about it. Hence the reliance on the seed treatments. (Arable Grower 3)

> Thanks to the use of very low rates of neonicotinoids we've reduced volumes of organophosphates massively and the neonicotinoids have only had an impact on sucking insects, where the organophosphates they kill everything. They kill the worms, the slugs. They would give birds a headache if I sprayed over that crop. (Arable Grower 1)

Well, when they came on it was a game-changer compare it to before that, you had to use an organophosphate or some granule or something like that to mix with the product which was a lot worse for your health and the environment than a little bit on the seed. (Arable Farmer 2)

Furthermore, neonicotinoids have been described as more cost-effective, less labour intensive and practical than the alternatives.

And the other benefit for us is. I know today that if we apply this product, I don't have to look at the crop every two days to see if there is an insect Invasion coming. (Arable Grower 1)

If you want to produce it economically, without it, the cost would be really, really high. So like nowadays we just can't pay, can't afford to pay people to go in there and weed things by hand. Yeah and with pest and disease. Even if you were to send people in there you couldn't possibly kind of like, you know, pick billions of or millions of aphids off the veggies. (Fresh Product Grower 2)

Even those interviewees who were considered the most prominent opponents to neonicotinoids by the others agreed with its use.

Neonicotinoids are an awesome chemical for a farmer. They are just great, you can apply it to your seed and you can have a product that keeps your plants safe for very low costs, the farmers would say that and the chemical companies would say that they're great for the environment. Because the amount of spraying is less, which in truth I would be inclined to agree with them. (Beekeeper 2)

There are non-chemical alternatives available to treat grass grub. However, like diazinon, they are incredibly destructive and much worse for the environment than a seed treatment neonicotinoid. As

explained below, farmers can perform intensive cultivation to kill grass grub. Not only is this option not a guaranteed fix, but it destroys the soil structure, resulting in several flow-on effects.

Actually, it might be worse for the environment because if you, for argument's sake, now you're doing minimum cultivation or direct drilling of autumn cereals. You need something down there to control the grass grub and the slugs. The only other alternative is intensive cultivation to actually destroy the grass grub. Yeah, so that's not going to be as good for the environment but even that is no guarantee. (Arable Farmer 3)

Although interviewees continually expressed their desire to move away from these destructive agricultural practices, particularly those which harm beneficial insects, they do not have any other option. Ultimately this comes down to the fear of catastrophic crop failure.

It's not cheap putting all this seed down and to have it all blown out by grub. And then it's just like you've got to re-drill it and then you've lost it, you know the yield on a crop and then like feed and a pasture situation like it's so. For years we've tipped phorate, you know horrible stuff organophosphates down the spout around here well dad has and then to a lesser extent I have because of the fear of Grub. Because you get out there you dig with your spade and I've done it. Dads done it. You dig all your spades across the paddocks and you find none. So sweet I won't put any stink down and then you get hammered and you do it the other way, where you find heaps and so you put it down but then who knows they might have come and gone and you would have been okay, it's it can be hit and miss with just finding the little suckers. And so the problem is once they're in and your crops in there's not a lot you can do about it so it's the fear of well, it's like what do I do? (Arable Grower 3)

Furthermore, several participants expressed that even if they were to stop using neonicotinoids, they would end up treating their crop with some other insecticide.

So, you know, it's like everything in agriculture, sort of. Often what the pure science says is not really what happens in reality, and there's always give and take. So if you did drop that [neonicotinoids] out and you look after your soil by not putting that on well you're throwing on an extra insecticide on top anyway, so, you know. Yes, you've got to weigh it up in terms of what you want to, what you're trying to achieve, I suppose. (Arable Grower 3)

There are many reasons why neonicotinoids are embedded in NZ agricultural practices. One of the most interesting was described by Product Group Representative 3. They explained that not only do grass grub help embed neonicotinoids because they are a significant pest, but also because they are an endemic species. What this means is that places, where neonicotinoids have been banned and restricted, are not exposed to the same pest pressure and limited crop protection methods that NZ agriculture endures. Furthermore, those international crop protection companies charged with the

task of introducing new chemistries are unlikely to formulate a chemical that is specially efficacious for grass grub.

Grass grub is endemic to New Zealand and not really a problem overseas so there's not a lot of global companies working on controlling grass grub. (Product Group Representative 3)

This is because NZ only represents a fraction of their global sales (theme 1), grass grub only exists in NZ, there is already a working alternative (neonicotinoids), and chemical discovery is very complicated.

The Bayer's and BASF's of the world they won't make a chemistry for grass grub they just discover molecules and then see what they might be efficacious on. So it's a little bit difficult. Yeah. (Product Group Representative 3)

Grass grub control brings to light the idea that necessity is the mother of invention, as presented in theme 2. The quotation below describes how there is not a drive from within the industry to seek out other options that deal with grass grub, despite one of the two available options being phased out.

> Growers are saying to me know that actually, you know it's not too many seasons away. Before diazinon's gone what have we got coming up and as you'll be aware the pipeline for you know registration can be three to five years in terms of trials and registration data packages getting together and then being assessed so there's not a lot coming through and if there was they would be saying you know that's fine we've been given a long phaseout but ever since 2013 there hasn't been a push to. Ask people to look for chemistry but I mean, I don't know if you talk to anyone in Discovery, but it's quite a complicated process and it's also very expensive. (Product Group Representative 3)

Essentially neonicotinoids are used in NZ for the early growth stage control of grass grub and aphids. The necessity of this early stage protection, the dominant pastoral landscape and the small amounts of pasture renewal and therefore treated seeds planted each year, means that neonicotinoids, as a seed treatment, will likely remain a part of NZ's agricultural practices. Furthermore, as explained in theme two, this section again demonstrates that there are not enough tools in the toolbox. There are only two easy to use and readily available options for dealing with grass grub in NZ, and one of them will be banned as of 2028. Rather than searching for a replacement, those relationships serving to cement neonicotinoids in NZ are further strengthening around their shared goal to hold on to this agricultural practice.

5.2.6 Theme 6: Relationship between beekeepers and landowners

A NZ House of Representatives document, discussing the health of bees, described how the relationship between beekeepers and landowners is very sensitive. It went on to explain that the

relationship is so sensitive that beekeepers are unwilling to complain if and when they suffer from the use of insecticides (Ardern et al., 2014). It suggested that beekeepers rely on the goodwill of landowners and are unlikely to put that under threat. The document also indicated that the NZEPA and the government cannot move on anything legislatively and cannot monitor the use of pesticides without complaints. This means that any potential reassessment of neonicotinoids will only occur if beekeepers record and complain about any beehive or pollinator losses caused by neonicotinoids in NZ.

My research has revealed that beekeepers are not merely not complaining, it also shows that it is nearly impossible for them to prove who and with what has affected their hives and that their complaints are often being ignored. The first of many reasons as to why beekeepers aren't complaining or their complaints are being ignored is that there is usually only "verbal contracts" (Arable Grower 1) held between landowners and beekeepers. However, this was not always the case; often, there is excellent relationships and formal contracts between beekeepers and landowner.

> Um, what we have in our contract we talk about not, it's a signed agreement don't use spray without talking to us. Generally, if we're going out into north Canterbury here that's a pretty intensively sprayed area, to be honest. So we always go, I always talk through what you use generally, what are your top 8 products, generally, 8 will cover it. And then we talk about what we need to know when we, you have like we basically have a rule that says tell us whenever you're gonna spray and I'll say yea sweet or I'll say give me a day I need to get our gear out of there. (Beekeeper 1)

The ones that have to pay compensation, make sure their spray contractor calls us beforehand to give us notice to move. They're not very good at it, they normally do it as an afterthought. So one time there we had just moved the hives in and the contractor called us saying that he was going to spray tomorrow. And were going you, you can't spray tomorrow we've just moved in 200 beehives, how the, we can't move them now, why didn't you ring us last week. Oh, we forgot to get around to talking to you, were required to talk to you before we spray, and I went well you can't spray until all the hives have moved. (Beekeeper 2)

In one situation a beekeeper described the formal contracts as only being in place to protect the landowner from being sued by the beekeeper if they were to kill or harm the beehives.

On another couple of the farms, the farmer has it the other way around. So the contact is done very much to protect him from us suing him for chemical poisoning. (Beekeeper 2)

Beekeepers are willing to sign such contracts due to the value of the Manuka honey which that bee site might yield (see Chapter 2 for more detail) and because they make a lot of money selling pollinators to crop farmers (Arable Grower 3). Following on from this beekeepers are unlikely to complain at the risk of losing those valuable sites. As one beekeeper explained It's against the law for them to poison the bees, they're not legally allowed to, but I have lost bee sites for complaining that they've poisoned the bees. (Beekeeper 2)

In contrast to the NZ House of Representatives document, this thesis research revealed multiple situations where beekeepers have complained about their beehives being harmed or killed. However, complaining does not always result in compensations for the beekeeper. Another reason why the NZ House of Representatives has not come across any complaints about beehive poisonings is that, as revealed by one beekeeper, no one is listening to those complaints.

> The problem is that federated farmers, they can't do anything. There's no one out there. Unless you, you can ring Ecan and try and get Environment Canterbury or one of the local regional councils in. It's quite difficult, they as a rule. A lot of the people there are pro-farmer, 'they need to do this for their business.' (Beekeeper 2)

Another problem proposed by the beekeepers interviewed was that even if they did complain, it is challenging to prove who killed or poisoned their hives. Which, not only comes down to the ability of bees to travel long distances to forage crops but also agricultural chemicals ability to drift.

One down the road from us where three different farmers' paddocks were sprayed. There were 70 hives on the block, the hives were either dead or nearly dead. But we didn't know which of the farmers and their contactor had sprayed the chemical that killed the hives. We're pretty sure it was the same contractor that sprayed all the paddocks, we assume it was the paddock that the hives were in got sprayed and we reckon the drift killed them. But you could easily argue that it was the neighbour's paddock that was the killer. So, therefore, who pays compensation? And we just bore the loss of the 70 beehives. (Beekeeper 2)

As suggested in the following quotation, it is incredibly difficult for a beekeeper to prove, or even understand what is killing or affecting their hives. Furthermore, even if a beekeeper were to undergo the rigorous testing necessary to determine what agricultural chemicals were affecting their bees, there would still be a "problem with cost" (Beekeeper 2).

> The problem with agricultural chemicals is that you don't know unless you're doing testing, you often don't know if you've got a problem. So you will see colonies with issues and you don't actually know why they've got those issues. And you can say, oh they've got a failing queen or they've just got, you don't know whether the queens got a pesticide issue. Unless you're testing the wax and testing the honey and testing the, doing the involved testing regularly, you won't actually know what's causing it. (Beekeeper 2)

The above comments not only outline issues surrounding the protection of beekeepers from chemical poisoning, but they demonstrate a number of problems which arise when attempting to restrict the use of chemicals like neonicotinoids. Essentially beekeepers are more likely to bear the hive losses than go through the complicated and expensive process of proving particular chemical and applicator killed their hives. Subsequently, there is a lack of complaints, and those complaints that do occur are ignored or disputed. A lack of complaints, regulators ignoring complaints and beekeepers inability to prove who and what poisoned their hives is inhibiting the reassessment of neonicotinoids. Which is subsequently preventing the NZEPA from justifying any restrictions on their use.

5.3 Summary

This chapter described the accounts of arable growers, fresh product growers, beekeepers, regulators, seed and chemical company representatives, product group representatives and researchers. Six themes were highlighted including NZ's small market for agricultural chemicals, not enough tools in the toolbox, market pressure, the disconnect between landowners and farming, grass grub and the relationship between beekeepers and landowners.

More specifically, theme 1 discussed the common notion that in NZ, the availability of agricultural chemicals is controlled by the NZEPA. It went onto demonstrate that a slow registration process and a lack of incentives for multinational crop protection companies to register chemistries in NZ is causing the lack of alternatives.

Theme 2 also starts by demonstrating the blame put on the NZEPA, but it also suggests that a lack of industry foresight is also responsible for the lack of neonicotinoid alternatives. It goes on to discuss how this encourages growers and their representatives to fight to keep those crop protection methods they have available.

Theme 3 discussed how consumer pressure is determining agricultural practices in NZ. More specifically, it demonstrated how consumers expectations and the way those expectations are being turned into action can vary depending on the agricultural sector. Furthermore, it showed how the presence and absence of MRLs have resulted in the removal of neonicotinoids in one and the maintenance of neonicotinoids in another of NZ's agricultural sectors.

Theme 4 illustrated the growing disconnect between arable growers and the daily farming practices occurring on their land. More specifically, it suggested that those reps taking over the crop management decisions are not well enough informed or sufficiently trained to be making good crop protection decisions. Subsequently, they rely on what they know works and therefore help to embed neonicotinoids in NZ agricultural practices.

Theme 5 introduced the concept that grass grub is the main reason why neonicotinoids are necessary to NZ. It went on to explain that the presence of this endemic species is a major difference between NZ's use of neonicotinoids and the rest of the worlds. It also suggested that NZs pastoral landscape

and low levels of pasture renewal means that it cannot be compared to Europe and the USA in terms of neonicotinoid use.

Finally, theme 6 explained why beekeepers do not complain when their hives are poisoned and why if they do complain, they are not being heard. It also suggested that a lack of complaints will mean that the NZEPA cannot justify restrictions on neonicotinoid use.

The next chapter provides a discussion around how particular relationships present in these themes have led to neonicotinoids becoming embedded in NZ's arable sector, while simultaneously being removed from the apples and pears sector. It will also show how a strict application of Bruno Latour's Circulatory System of Scientific Facts (1999a) only revealed half the story. Subsequently, it will go on to demonstrate why deviations from this framework were necessary to reflect the data better. By pairing this framework with the interview data, the next chapter will show how neonicotinoids as an agricultural practice is the sum of those relationships described above.

Chapter 6

Discussion

6.1 Introduction

The main aim of this research is to understand how neonicotinoids are embedded -and remain- in NZ's agricultural practices. By pairing the interview data presented in Chapter 5 with Latour's CSSF (1999a) and Callon's key principles (1986) of "generalised symmetry", "agnosticism" and "free association", this chapter will show how the use of neonicotinoids, as an agricultural practice, is the result of various material relationships. This chapter also uses these tools to describe the inner workings of NZs agrichemical market and agricultural community.

As explained in section 4.3 I did not initially assess my results using the CSSF because the thematic analysis was a good way of making sense of the data for myself. Also by presenting the conflicting viewpoints of all the actors and quoting the exact words of the interview respondents, I have limited my biased commentary, identified the power relations and social hierarchies as they have emerged from the data and therefore combatted some of the limitations of the CSSF (Latour, 1999a) and Callon's key principles (1986) described in section 3.5. I have not strictly abandoned the thematic framework presented in Chapter 5. Instead, I have followed Warner (2008, 2016), Edwards (2014), Ingram (2007) and Fitzsimmons (2004) and have framed those themes, or the key relationships they revealed, through the CSSF lens. In doing this, I have also contributed to answering the fourth research question of this thesis: How can Latour's CSSF be used to investigate similar controversies in the future, in particular, those which involve hazardous chemicals used in NZ agriculture?

Practically, this chapter utilises Latour's CSSF (1999a) to represent the data outlined in Chapter 5, catalysing an understanding of those actors influencing the scale and frequency of neonicotinoid use in NZ and the relationships existing between them. This framework has encouraged me to follow neonicotinoids as they are created by the interdependent relationships existing between those actors described in the previous chapter. Subsequently, this chapter will address the first two research questions presented in Chapter 1:

- 1. Who and what are the major actors that influence the scale and frequency of neonicotinoid use in NZ, and what have been their roles?
- 2. How have certain relationships formed between actors?

Not only will this chapter bring together the different accounts presented in Chapter 5, but it will ask if the CSSF (1999) has offered a useful way of extracting critical narratives about neonicotinoids from the data. Subsequently, it will address the third research question presented in Chapter 1:

3. How are neonicotinoids as an agricultural practice embedded -and maintained- in NZ, and why has opposing knowledge been unsuccessful in removing it?

This chapter has revealed how knowledge opposing the use of neonicotinoids has contributed to it being removed from NZ's apples and pears sector and it has also tried to make sense of why it has not done the same in the arable sector. Finally, this chapter will bring together the key content of previous chapters, discussing the strengths and weaknesses of the CSSF (1999a) and considering how it has allowed for a useful understanding of neonicotinoids. This will subsequently help to answer the final question:

4. How can Latour's CSSF be used to investigate similar controversies in the future, in particular, those which involve hazardous chemicals used in NZ's agriculture?

6.2 The arable sector, the stable network

The interviews outlined in Chapter 5 revealed that neonicotinoids are used as an agricultural practice across NZ's agricultural industry. However, it also revealed that how they are used depends on the commodity being supplied and the agricultural sector itself. What follows is an analysis of that data using Latour's CSSF (1999a) to demonstrate how neonicotinoids are embedded in the agricultural or farming practices of NZ's arable sector.

Autonomization

Latour's loop called "autonomization" refers to how a "discipline, a profession, a clique or an invisible college becomes independent and forms its own criteria of evaluation and relevance" (Latour, 1999a, p. 102). Essentially this loop illustrates how an area of expertise has been created around neonicotinoids. In terms of NZ's arable sector, this clique, college or area of expertise is made up of those actors who are convinced of neonicotinoids importance for growing high-value export quality seeds, it therefore closely resembles the market pressure theme but it also draws on data from theme 1, 2 and 4 of the results chapter. My data is useful for understanding how neonicotinoids have been made and maintained in NZ's arable sector, and therefore this loop includes the relationships between arable growers, NZ seed and chemical companies and international seed markets. It is important to note that this area of expertise has not been built through these actors coming together and learning about neonicotinoids. Rather, their relationships and this area of expertise rely on the trade of material things and the translation of interests. I will refer to this area of expertise as the

arable profession because it is made up of intercommunicating groups and individuals who are working to produce high yielding and pure seed crops.

Many other relationships likely exist that are not discussed here. However, for the purpose of this research, I have focussed my analysis on the three-way relationship between arable growers, NZ based seed and chemical companies and international seed companies. This will also enable me to demonstrate how this profession is contributing to neonicotinoid use in NZ's arable sector and is simultaneously a consequence of this use.

As outlined in section 5.2.3 NZ's arable sector exports the majority of its products overseas, particularly to Europe. More specifically international seed companies are interested in multiplying hybrid seeds and want to ensure the subsequent crops are high yielding and 100% pure. Therefore, they deploy the multiplication of these hybrid seed crops to NZ. To gain access to land, resources and growers, these international seed companies must form relationships with NZ based seed and chemical companies who almost act as a middleman between the international company and the arable grower. To this end, the international seed companies mobilise various intermediaries by which it can act upon NZ based seed and chemical companies. These include a higher payment for the NZ based seed and chemical company compared to if they had the crop multiplied elsewhere in the world and a set of standards about quality and yield (see section 5.2.3 and 5.2.4). Essentially these international companies have outlined expectations of yield and purity. They do not want the seed cross-pollinated, biosecurity dictates it cannot be contaminated with weeds and "they do not want the excuse of the weather or anything as to why there's not a crop" (Arable Grower 2). The NZ based seed and chemical companies are not passive in this relationship. They interest the international companies with NZ's longitude which is perfect for seed production during the European off-season (Scottville Farm, 2017), access to water, good farming practices and NZ's more relaxed rules when it comes to some substances deemed hazardous elsewhere (section 5.2.3 and 5.2.4). Presumably, both actors have various representatives who engage in face to face meetings with one another to align their interests and strengthen this relationship.

The negotiations between International seed companies and NZ based seed and chemical companies are translated into the farming practices of NZ's arable growers. The NZ based seed and chemical companies mobilise intermediaries which serve to influence the farming practices of NZ's arable growers and subsequently align the interest of all those actors in this "autonomization" loop. These intermediaries include fertiliser, irrigation and crop protection programmes which serve to ensure that those yield and purity expectations set by the international companies are met (see section 5.2.3). They further interest the arable growers with payment for multiplying the seed and market access. To secure their position and gain the contract to multiply the seed NZ arable growers have to

mobilise intermediaries of their own to interest the NZ based seed and chemical companies. These are often simply a commitment to follow those crop management plans set out by the NZ based seed and chemical company. Furthermore, they use programs such as Agworld to confirm they have performed crop treatments, irrigation and fertiliser applications as directed.

Getting back to the core tenant of this thesis, the crop protection plans mobilised to and followed by arable growers always contain the treatment of seeds with neonicotinoids. Due to the lack of alternatives for neonicotinoids (see section 5.2.2), the prevalence of grass grub (section 5.2.5) and NZ seed and chemical company reps reliance on older chemistries (see section 5.2.4), NZ based seed and chemical companies translate the expectations of international seed companies as; 'to minimise the vagaries of growing outdoors you must use neonicotinoids'. Subsequently, the security of arable growers' contracts with a NZ seed and chemical company and even their ability to buy seed is dependent on the use of neonicotinoids as a seed treatment (see section 5.2.4).

Alliances

Latour's loop called "alliances" describes how various actors who are generally not interested in the object of the network are captivated by it (see section 3.6). Or as Latour illustrates "the military must be made interested in physics, industrialists in chemistry, kings in cartography, teachers in educational theory, congressmen in political science" (Latour, 1999a, p. 103).

In the case of NZ's arable sector, this loop represents how a diverse array of powerful actors are involved in the successful production of high yielding and 100% pure hybrid seed crops. More specifically, it explains how the NZEPA and beekeepers are involved in embedding neonicotinoids in the farming practices of the arable sector, therefore it draws on data from theme 1 and theme 4 of the results chapter. Essentially high yielding and pure hybrid seed crops cannot be grown without the use of neonicotinoids, but the arable profession cannot use neonicotinoids unless these diverse actors become interested in them. This loop, therefore, demonstrates how those actors (the NZEPA and beekeepers) become interested in neonicotinoids. Unlike the above loop which discussed a three-way relationship and the formation of an invisible college, the "alliances" loop discusses two separate relationships which have formed between actors within the arable profession and actors that are "rich and well-endowed" (Latour, 1999a, p. 104).

The first relationship is between the NZEPA, arable product groups and arable growers. Essentially this relationship describes the activities that result from there not being enough tools in the toolbox (see section 5.2.2) for NZ's arable growers to protect their crops from pest species such as grass grub (see section 5.2.5). Once again, this relationship relies on the trade of resources; product groups primarily function as a representative body of NZ's various agricultural sectors. They collect levies from growers based on yield, land area or at a fixed rate and in return, growers receive marketing,

research and development as well as some practical support. In terms of this loop, arable product groups also provide lobbying of the government and its agencies.

The NZEPA made a call for information regarding the use of neonicotinoids in NZ during August 2018. This was the first step in assessing whether this group of agricultural chemicals needed to undergo a reassessment. NZ grower groups were able to convince the NZEPA that neonicotinoids are essential for the success of key export crops, particularly those smaller crops which do not have access to a wide range of alternative crop protection methods (section 5.2.2). Therefore, by mobilising intermediaries in the form of critical use cases to the NZEPA, product groups interested this powerful actor in neonicotinoids. NZ grower groups have been able to create a strong "alliance" with the NZEPA, linking their interests (keeping tools in the toolbox) to the interests of the public and the NZEPA. Not only is the NZEPA convinced of neonicotinoids importance to NZ's agricultural production, but they are also convinced that biodiversity declines are not caused by neonicotinoids. This is because, as suggested in section 5.2.5, the pastorally dominant landscape of NZ is different from those in Europe and North America. The NZEPA is not passive in this alliance. But rather than it being about what resources they trade, here it is about what they do not trade. They have not changed neonicotinoid regulations, and therefore, the actions of grower groups and growers have not changed. This alliance, built by product groups, has resulted in the authoritative acceptance of neonicotinoids as a safe, highly targeted and essential agricultural chemical by the NZEPA.

Another "alliance" which is informal, but still significant to this case study is that which exists between arable growers (landowners) and beekeepers (see section 5.2.6). Essentially landowners need to convince beekeepers of their agricultural practices. This is because the NZ House of Representatives can only move on neonicotinoid regulations and keep accurate hazardous substance records with complaints (see section 5.2.6). The relationship between landowners and beekeepers is very sensitive and highly valuable to both parties. Landowners mobilise intermediaries including formal contracts which offer beekeepers compensation if their hives are poisoned, access to stances of highly valued Manuka and a fee for pollination services (see section 5.2.6). These intermediaries serve to convince beekeepers of their agricultural practices. Beekeepers are not passive in this relationship. Rather they mobilise intermediaries of their own to influence the activities of the landowners/arable growers. They similarly mobilise contracts which protect their hives from being poisoned and pay landowners for access to honey-producing flora. This subsequently strengthens their "alliance" and when contractually obligated, landowners inform beekeepers of when they are using hazardous substances (see section 5.2.6).

Due to the growth of NZ's honeybee population and the Manuka honey market (see section 2.4), those beekeepers with access to Manuka are unlikely to complain even in cases where their hives

have been poisoned, this is due to the risk it poses their income. Returning to the core tenant of this thesis, landowners and beekeepers will sometimes negotiate about the agricultural practices being used. Understanding that neonicotinoids are necessary for crop protection and seeing as there is no NZ based evidence suggesting they harm honeybees, beekeepers are often relieved if neonicotinoids are used, particularly when they are being used in place of another chemical (see section 5.2.2).

Public representation

The next loop of Latour's framework, "public representation", considers how people's everyday practice, system of beliefs and opinions are affecting how science is done, or in the case of this research how an agricultural practice is done (Latour, 1999a, p. 105). For example, Latour illustrates that "a major part of advanced research in molecular biology in France ... depends on a private charity's annual telethon to combat muscular dystrophy. Every argument for or against genetic determinism will feedback into this funding" (Latour, 1999a, p. 106). Because NZ's arable growers are "one generation of seed away from the consumer" (Arable Grower 2), or the public which this loop describes, there is much less accountability than Latour may have anticipated when he initially described this loop. Nevertheless, this loop represents how consumer demands are shaping the agricultural practices occurring on NZ arable farms and it therefore draws on the data of Theme 3.

Consumers are demanding fresh produce on their shelves every day, even if it is out of season and even in places where that product might not grow (see section 5.2.3). One could argue that this helps drive the demand for NZ grown hybrid seeds and the resulting use of neonicotinoids. As demonstrated in section 5.2.3, this demand for fresh produce has shaped the market into "what it has from an Agri compound point of view for suppliers and growers" (Product Group Representative 2). So, as the public is becoming more and more interested in gaining access to fresh food every day, they are subsequently and inadvertently embedding the use of hazardous substances like neonicotinoids in NZ's arable sector. It is the relationship which international seed companies have with this public, the consumer, and the subsequent generational gap which means that there is no drive to get rid of agricultural chemicals as there is in the apples and pears sector (see section 6.3). One arable grower (Arable Grower 2) explained that it is the next generation who are growing vegetables from NZ grown seeds who have to respond to market demands around hazardous substance use.

Consumer demand for clean and high yielding seeds out of season is also a part of this loop. Here a relationship exists between the users of NZ grown hybrid seeds and international seed companies. This once again relies on the trade of resources. The international consumer or those vegetable growers mentioned in the paragraph above are demanding high-quality vegetable seeds which express certain traits and are free of weeds, pests and diseases. Simply enough the consumers

interest the international seed companies with payment and those companies provide the consumers with the seed, which is of high-quality, expresses the desired traits and is free of weeds, pests and diseases. This allows both actors to fulfil their respective goals that neither can achieve alone. Profit for the international seed company and vegetable seeds for the consumers.

Mobilization of the world

The next loop of Latour's CSSF is "mobilization of the world". As explained in Chapter 3, it describes all the means by which non-humans are progressively loaded into discourse (Latour, 1999a, p. 99). When used to study how science is done, this loop deals with the instruments, equipment, expeditions, surveys and sites involved in scientific research (Latour, 1999a, pp. 100-101).

Given this thesis is not concerned with how science is done but with how an agricultural practice is done, "mobilization of the world" is a matter of how things and non-humans are of moving towards the world, making it mobile, bringing it to the site of controversy, keeping it engaged and making it available for arguments. Essentially it is about how non-humans are connecting the sites of this controversy, for example, the hybrid seeds have become a commodity that connects something as big as the international commodity market to something as small as a farmer's paddock. It is also about how these things are being mobilised and made available for arguments. For example, "An ecologist whom nobody used to take seriously can now intervene in a debate with beautiful satellite photographs that allow her, without budging from a Paris laboratory, to observe the advance of the forest in Boa Vista" (Latour, 1999a, p. 101). Although this section draws on data from all the themes in Chapter 5, it is particularly reliant on the Grass Grub theme.

As the ecologist loaded satellite photographs into her discourse about the Boa Vista, arable product groups have loaded grass grub into their defence of neonicotinoids. If grass grubs were not present in NZ's agricultural landscape, there would be no basis for the neonicotinoid critical use cases put forward to the NZEPA and these actors would, therefore, be unable to fight for what they have got (see section 5.2.2). This pasture pest is made to move in the critical use cases and the supporting documentation supplied by product groups to the NZEPA opposing neonicotinoid reassessment and supporting its initial registration (see section 5.2.2). So through the mobilization of pasture pests like grass grub, we can see how the offices of the NZEPA are linked to the day-to-day practices occurring in a South Canterbury field.

As grass grub was mobilised to connect sites and load reality to claims about neonicotinoids, the 100% pure seed crops demanded by international seed companies have become a commodity that serves to connect arable fields with the international seed market. Essentially international seed companies will not deploy seed production to NZ without a guarantee of purity and yield. Subsequently, NZ based seed and chemical companies translate this demand and load their

arguments with seed treatment neonicotinoids. They both mobilise these treated seeds in the form of crop protection plans to landowners, arguing it is the only way they can gain market access (see section 5.3.3). Simultaneously NZ seed and chemical companies are mobilizing the treated seeds to the international seed companies arguing that they are utilizing these crop protection methods as a way of minimising the risk of losing a crop to the vagaries of growing outdoors, therefore guaranteeing a pure high yielding seed crop. The mobilization of treated seeds has been so successful that arable growers struggle to buy any seed which is untreated (Arable Grower 3). This mobilization of neonicotinoid-treated seeds subsequently strengthens the relationships in the arable profession and further embeds neonicotinoids in the farming practices of NZ's arable sector.

Honeybees are progressively loaded into the discourse of many actors throughout this network. More specifically product groups mobilise the lack of complaints regarding honeybee deaths associated with neonicotinoids in NZ and similarly the lack of cases of CCD (see section 2.7), in those critical use cases put forward to the NZEPA. When the call for information on neonicotinoids went out in 2018, these product groups, mobilised this non-human along with grass grub to defend neonicotinoids (see section 5.2.2) and subsequently help embed -and maintain- them in the farming practices of NZ's arable sector. In a similar way, landowners have mobilised this lack of complaints, cases of bee deaths and cases of CCD to strengthen their "alliance" with beekeepers.

Links and knots

Latour (1999a) calls the fifth and final loop of the CSSF "links and knots". This is done to avoid the historical baggage associated with the term conceptual content. This central loop is not "a pit inside the soft flesh of a peach" (p. 106), but it is "a very tight knot at the centre of a net" (p.106). In Latour's world of studying scientific practices, this loop is the scientific content that ties all the loops together, but which cannot exist in isolation from them, serving as the conceptual heart of the network. Without this loop, the others would die, but this would happen just as quickly if any of the other loops were amputated (p. 107).

In terms of this case study, the NZ arable sector and an investigation into how neonicotinoids have become embedded in NZ's agricultural practices, the "links and knots" are neonicotinoids. Neonicotinoids are everywhere in the network but at the same time nowhere. No single part of this network can be used to explain the use of neonicotinoids in NZ's arable sector because it is distributed throughout the entire network. In terms of NZ's arable sector, the network itself and therefore, the "links and knots" are stabilised because of the action and interaction present in the four other loops. This will be explained in more detail below.

Similar to how a "concept does not become scientific because it is farther removed from the rest of what it holds, but because it is more intensely connected to a much larger repertoire of resources"

(Latour, 1999a, p. 108), agricultural practices become embedded because they are more intensely connected to more things, sites and actors. The "links and knots" described in this case study is exactly that agricultural practice which churns, steers, moves and connects the rest of the world of data, colleagues, allies and spectators.

The arable sectors blood flow

The CSSF (Latour, 1999a) suggested that five activities, which have simultaneously been performed and worked, have embedded neonicotinoids in the agricultural practices of NZ's arable sector. These five activities or loops have described the key relationships, connections, things and sites that have been made and maintained in the on-going use of neonicotinoids. Due to the interconnection of actors, things and sites described above, they are all necessary for the overall stability of the arable sectors network. If any of the loops were broken, the blood flow of the entire network would stop. Therefore, it is no individual actor or relationship but all of those described above which have made and maintained and therefore embedded neonicotinoids in NZs arable sector. These loops or activities are the creation of neonicotinoids as an agricultural practice (links and knots), the making of things, documents, sites and other non-humans available for arguments about neonicotinoids (mobilization of the world), the forming of an arable profession who share the same goal of producing high yielding pure seed crops and see neonicotinoids as one of the methods for achieving this (autonomization), making "alliances" with important and powerful actors outside of the arable profession to support neonicotinoids or at the very least not oppose their use (alliances) and aligning the arable sectors agricultural practices with the demands of consumers (public representation).

I have used Latour's model to interpret the use of neonicotinoids in NZ's arable sector. Seeds and other arable crops have been grown in NZ since the early days of European settlement (see Chapter 2), but more recently arable growers have collaborated with product groups as well as NZ seed and chemical companies to capture European off-season production. Furthermore, since the introduction of the HSNO ACT, the subsequent establishment of the NZEPA and the systematic reassessment of older hazardous substances, these actors have had to engage with additional actors to develop new and maintain existing farming practices. Furthermore, the growing social expectation of reduced hazardous substance use in food production has forced additional actors (consumers and the public) to participate in the network. Lastly, the death of all unmanaged honeybee hives in NZ and the subsequent re-birth of the apiculture industry (see section 2.4) has forced beekeepers to participate in the making and maintenance of agricultural practices.

What I have described above are the key components of the arable sectors network representation which illustrate the activities of human and non-human actors as a kind of blood-flow which sustains the life of neonicotinoids as an agricultural practice. This approach is a communicative understanding

of the maintenance of neonicotinoids as an agricultural practice in NZ's arable sector, which emphasizes the presence of dynamic relationships with material nature. The circulation of each loop is dependent on that of all the other loops. Therefore, the blood flow of this network is what maintains neonicotinoids as an agricultural practice. How much neonicotinoids are used, therefore, how embedded they are is proportional to the speed of this blood flow.

To study "mobilization of the world" is to study the logistics that are so indispensable to the logics of agricultural practices. Subsequently, to study the logistics of neonicotinoids in NZ's arable sector required not only the study of the things which neonicotinoids combat, support and potentially harm (grass grub, clean high yielding crops, honeybees), but the places which these things are found, translated and made available for arguments (international markets, farmers' fields, NZEPA offices). This research revealed that the high yielding and 100% pure seed crops are loaded into discourse to strengthen the three-way relationship between arable growers, NZ seed and chemical companies and international seed companies. Furthermore, the seed coating neonicotinoids, as a method of meeting international seed company expectations, are also loaded into the discourse between these three actors. Subsequently strengthening the arable profession and speeding up the circulation of the "autonomization" loop.

NZ seed and chemical companies have mobilised neonicotinoid seed coatings in the form of crop protection programmes convincing arable growers that neonicotinoids are the only method of ensuring market access as well as protecting their crops from pest pressure during early establishment. This also strengthens the NZ seed and chemical companies argument that international seed companies do not want any excuses as to why there is no crop. Similarly, the NZ seed and chemical companies mobilise the treated seeds to the international seed companies to demonstrate that everything is being done to minimise the potential of crop failure and loss, therefore convincing the international seed companies to both pay a higher rate than if the seed was grown elsewhere in the world and to continue having their seed multiplied in NZ. What this demonstrates is that the circulation within the "mobilization of the world" loop is dependent on the circulation of the "autonomization" loop. Therefore, if actors within the arable profession fail to translate their interests and the interests of others in any of the other loops, the whole network would collapse. For example, if the NZ based seed and chemical companies failed to translate the interests of international seed companies, high yielding and pure seed crops in the European offseason, to NZ arable farmers' interests, follow these crop management programmes and you will gain market access, the whole network would fail. The goal to produce high yielding and 100% pure arable crops is one of the most powerful accelerators of this circulatory system. It was frequently the primary trigger for neonicotinoid use in this sector, and it forms "the seeds of all relationships amongst" the arable profession (Latour, 1999a, p. 103).

By aligning their interests with those of beekeepers, the arable profession can form "alliances" with them. Landowners (arable growers) mobilise non-humans such as access to manuka and contracts which demonstrate that they will inform the beekeeper of when they are applying chemicals. Furthermore, in direct relation to neonicotinoid use, the arable profession mobilises government and historical data which suggest that no honeybees have died in NZ as a result of neonicotinoid use and the commonly associated bee disorder (CCD) has also never occurred in NZ. By mobilising these non-humans the landowners form an "alliance" with beekeepers who subsequently support, or at the very least do not directly oppose neonicotinoid use in the arable sector. This informal "alliance" can become so strong that in many cases even if beehives are poisoned the beekeeper will simply bear the hive losses. Essentially the value of the non-humans being mobilised, access to Manuka and pollination contracts, is so great that beekeepers are unwilling to threaten their circulation by complaining.

If the interests of the landowners were not translated to those of beekeepers and subsequently these non-humans were not mobilised, socialised and loaded into the negotiations between the landowners and beekeepers, it is likely that at times where beehives are poisoned the beekeeper would be more inclined to complain. As explained in section 5.2.6, if there were more complaints, the NZ government would be able to accurately measure hazardous substance use and would be able to justify further regulation or even bans on neonicotinoids. Furthermore, product groups would be unable to mobilise the lack of bee deaths associated with neonicotinoids in their critical use cases sent to the NZEPA. This is just one example of how a change in one relationship or one loop being cut off would result in the breakdown of all the other loops and therefore, the entire network. Subsequently, the circulation of access to Manuka, contracts and the aforementioned records concerned with bee deaths, are speeding up the circulation of the "alliances" loop. This, in turn, accelerates the circulation of all the other loops and further embeds neonicotinoids in the agricultural practices of NZ's arable sector.

By aligning their interests with the NZEPA's the arable professions can form a strong "alliance" with the regulatory body charged with registering and reassessing hazardous substances in NZ. They have achieved this and strengthened the "alliance" by mobilising non-humans like grass grub, honeybees and the seed coating neonicotinoids in the form of critical use cases for neonicotinoids. The arable profession has been able to form an "alliance" with the NZEPA who are now convinced that seed coating neonicotinoids are highly targeted, environmentally friendly, are the best way to combat the endemic pest grass grub in terms of efficiency and environmental impact and are not linked to any cases of bee death in NZ. This strong "alliance" demonstrates the blood flow between the "mobilization of the world", "autonomization" and "alliances" loops, as well as the "links and knots" or the neonicotinoids that tie them all together into a coherent network. If one of those non-humans

was not circulated to the NZEPA or was cut out of the network, the entire system which embeds neonicotinoids in NZ's arable farming practices would likely collapse. For example, if those levy based product groups which represent arable growers at the regulatory level, did not mobilise seed coating neonicotinoids in their critical use cases to the NZEPA demonstrating that they are the only safe option for combatting grass grub, the NZEPA would likely be more inclined to reassess and further regulate neonicotinoids. As Latour expresses "each of the five activities is as important as the others, and each feeds back into itself and into the four others" (Latour 1999a, p. 99). Therefore, without allies no neonicotinoids and no agricultural practice for combatting grass grub; therefore, no proof to build an arable profession, and no hybrid seed production in NZ. Or in simpler terms, without the "alliance" of the NZEPA neonicotinoids would be more heavily regulated or even banned in NZ. Subsequently, the arable profession would not be able to grow 100% pure high yielding seed crops, and arable farmers would have to revert to using other hazardous substances to deal with pest pressure. This would simultaneously contribute to increased cases of beehive deaths, consumer demands would no longer be met, and neonicotinoids would ultimately die in this agricultural sector.

Public concern over the quality, accessibility and traceability of their food likely impacts regulatory decision made about hazardous substance applications and regulatory controls around food safety. However, as explained throughout this thesis, there is a generational gap between arable production occurring in NZ and the end consumers, or the public interested in those agricultural practices contributing to the production of their food. Nevertheless, public demands internationally and particularly those in Europe are helping to define the object of this network. It is defining the agricultural practices occurring in NZ's arable sector. Similar to how finding solutions for public concern is a major motive for the activities of academic communities, meeting consumer demands is a major motive of agricultural ones. For example, there is a growing demand, from consumers, to have fresh food available all year round, which has led to the globalization of food production. As explained in Chapter 2, this export of agricultural commodities has long been a part of NZ's supply chain. Thus, one could argue that NZ's arable production is answerable to the concerns of international publics. It is international seed companies who translate the interests of their publics, fresh produce every day even out of season, into their own interests, gaining a 100% pure and high yielding seed crop out of season. It is here that the arable profession becomes concerned with public demands, and therefore the "public representation" loop is connected to the "autonomization" loop. Subsequently, this public demand is strengthening neonicotinoid use in NZ's arable sector. As with the previous blood flow between loops, if the international seed companies did not translate the interests of their consumers, the entire network would fail.

Another public, whose impact on NZ's arable sector's neonicotinoid use is likely more difficult to disprove, are the users of that seed grown by NZ arable growers. These are those fresh product

growers situated internationally who purchase the NZ grown hybrid seeds from international seed companies. As discussed in the "public representation" section, they demand high yielding highquality seeds which express certain traits and are free from pests, weeds and diseases. Here the arable profession becomes concerned with public demands, the international seed companies translate these demands to the NZ based seed companies and subsequently the arable growers. Like above, if this translation did not occur, or there was not the same scrutiny around the quality of seeds, neonicotinoids may not be embedded in NZ's agricultural practices because the international seed companies might simply contract their seed production to a different country where it is a lot cheaper. Or if seed production remained in NZ, but international seed companies' expectations were different, NZ seed and chemical companies would similarly have different translations and may no longer mobilise seed treatment neonicotinoids to arable growers, and arable growers might be able to gain market access without the assurance that they have used neonicotinoids.

The "links and knots" of the model described above are directly moved by the four other loops, but at the same time, the "links and knots" are what "keeps them [the other loops] altogether, what strengthens their cohesion, what accelerates their circulation" (Latour, 1999a, p. 108). So essentially as Latour puts it "we have followed the veins and arteries and arrive now inevitably at the pumping heart" (Latour, 1999a, p. 106). This heart is both hard to study and hard to define as it is a very tight knot at the centre of a net which must hold many diverse resources together. If circulation stops in only one loop, the circulation of the whole model will stop because each loop is connected to each of the other four loops. Therefore, each of the activities, things, places and relationships described above are directly connected to the "links and knots" of this model. In the case of NZ's arable sectors and its use of neonicotinoids, the "links and knots" is the strong and heavily embedded agricultural practice of using neonicotinoids. This is linked to the "autonomization" loop as it is the arable profession or actors within it who are directly applying the neonicotinoid to seeds and then planting those treated seeds. This link between these two loops is so strong the "links and knots" might appear independent from the other loops. However, any change to the other loops can change the circulation between the "links and knots" and the "autonomization" loop. For example, an increase in consumer demand for transparency through more generations of food production, or an NZEPA reassessment of neonicotinoids would directly stop this circulation, NZ seed and chemical companies would no longer include neonicotinoids in their crop management plans, and therefore arable farmers would not use it anymore. However, as it goes, neonicotinoids are heavily embedded -and maintained- in NZ's arable sector.

6.3 The apples and pears sector, the destabilized network

This research revealed a second network that was not anticipated at the outset. While following the actors using Latour's CSSF (1999a) I was pointed towards the relationship between growers and grower groups and it was through interviews with these actors, particularly the grower groups that this loop emerged. Because my research began as an investigation of NZ's agricultural industry as a whole, I thought it was a good idea to interview actors who were involved in the production of various agricultural commodities. Therefore, I reached out to grower group representatives from not only the arable sector but also NZ's apples and pears sector. It was during interviews with these individuals that it emerged that while in the arable industry neonicotinoids are heavily embedded in crop management practices, that is not the case for the apples and pears sector. What follows is an analysis of the apples and pears sector using Latour's CSSF (1999a) that subsequently reveals the discontinuation of neonicotinoids rather than its on-going use observed in the arable sector.

Autonomization

Once again, this loop refers to how a "discipline, a profession, a clique or an invisible college becomes independent and forms its own criteria of evaluation and relevance" (Latour, 1999a, p. 102). But unlike the previous network here this invisible college is made up of fresh product growers, apples and pears NZ and domestic as well as international supermarkets. This invisible college is made up of those actors interested in the production of high yielding and high-quality fruit crops.

A fruit crop is subject to a different level of scrutiny and undergoes different tests depending on which supermarket is buying the product. This scrutiny normally includes various quality assurance tests, including checking the size, quality and ripeness of the fruit as well as testing the levels of chemical residues that are on it. The relationships in this loop are built on the exchange of resources. The supermarkets circulate quality standards to apples and pears NZ, who in return, connect the supermarkets to NZ growers. In exchange for a levy paid by growers, apples and pears NZ exchange crop protection plans which ensure residue levels are met, marketing services, industry representation and market access. Having been put in contact with the domestic and or international supermarkets growers send produce to the supermarkets in exchange for payment, under the condition that those quality assurance standards mobilised by the supermarkets are met. All going well resources and negotiations circulate through this loop and subsequently strong relationships and the apples and pears profession is made and maintained.

Alliances

This loop bears the most similarity to that described in the arable sector. It is again made up of two separate "alliances" in which an exchange of resources effectively interests powerful and wellendowed actors. Here only one actor has changed, and that is simply due to the agricultural sector which this network represents. Instead of an arable product group, this loop now describes apples and pears NZ. Functionally this loop has not changed, the action or the circulation of this loop is virtually identical to that described in the arable sector. As above it considers the relationships between product groups and the NZEPA as well as landowners and beekeepers. Once again, it represents the action of the invisible college to interest powerful actors in their farming practices.

The first "alliance" exists between the NZEPA and apples and pears NZ. Once again, this relationship relies on the trade of resources, apples and pears NZ mobilise critical use cases for particular agricultural chemicals or crop protection methods to the NZEPA, and in exchange, the NZEPA may tighten regulation or allow there continued use under current regulation. As described in the arable sector, in terms of neonicotinoids, the NZEPA is convinced of its necessity to produce smaller crops, and because of the money, it generates for NZ.

The second "alliance" is that between landowners and beekeepers, once again this "alliance" relies on the transfer of pollination services from the beekeeper in exchange for finances and sometimes access to honey-producing stances of native bush. It is still an informal "alliance" where landowners must convince beekeepers of their farming practices, particularly those with the potential to harm beehives (see section 5.2.6). Landowners (fresh product growers) in the apples and pears sector, like those in the arable sector, generally mobilise contracts which offer beekeepers compensation if their beehives are poisoned and a reasonable warning time before they spray. Furthermore, landowners might mobilise access to stances of highly valued Manuka (see section 5.2.6). These intermediaries serve to convince beekeepers of the agricultural practices occurring in the orchards. Beekeepers are not passive in this relationship they also circulate resources to influence those agricultural practices, including contracts which protect their hives from being poisoned, pollination services and in some cases where bee sites yield Manuka honey, a fee for leasing that site. Like in the arable sector, in the unlikely case where a beekeeper's hives are poisoned, they are unlikely to complain due to the risk which this may pose their income. That income being those finances they earn from either the pollination services they provide or the Manuka honey they produce. Essentially this "alliance" is strengthened, and beekeepers are convinced of the farming practices because of the finances they earn, and the risk which complaining about the practices might cause their business.

Public representation

In this "public representation" loop, we can see some of the most significant differences between the arable sector and the apples and pears sector. Where consumer demand in the arable sector helped drive the use of hazardous substances like neonicotinoids in farming practices, here it is doing the opposite. There is no generational gap between the fresh product growers and the consumers like there was in the arable sector. The apples and pears sector is producing fresh products going directly

to supermarkets around the world for immediate consumption. As before consumers are demanding fresh produce on their shelves every day. However, due to global trends (see section 2.6), these same consumers are becoming more and more conscious of what is going into the production of their food. Subsequently, consumer demand is now having a large effect on those agricultural practices occurring in a farmer's field or a grower's orchard. As explained in section 5.2.3, many of NZ's high-value apple and pear export markets have been going through a transition process away from certain chemicals, neonicotinoids being one of the most controversial of these. So a growing consumer awareness of the practices contributing to the production of their food, a demand for a more transparent supply chain and stigmas around particular chemicals, particularly in regions of the world which have previously banned their use, are influencing the agricultural practices occurring in NZ's apples and pears sector.

There is a relationship here which again relies on the transfer of resources. International consumers of NZ's apples and pears are exchanging payment to supermarkets, and in return, they are receiving and expecting fresh fruit, which is free from agricultural chemicals all year round. Essentially this consumer demand for fewer chemicals is being circulated through all the loops of the network. This circulation and the effect which this is having on the ultimate use of neonicotinoids in NZ's apples and pears sector will be explained below.

Mobilization of the world

There are a number of differences between this loop and that one described in the arable sector. Grass grub is not present in this network; however, other insect pests, like aphids, are and are likely the reason why neonicotinoids have been used in apple and pear orchards in the past. A significant difference is that these insect pests do not come with the same inability to treat and lack of alternatives that have been a problem for grass grub control. Another significant difference is the presence of seed treatments, due to the nature of the orchard industry neonicotinoids or similar insecticides are applied as a spray or soil drench. This means that the apples and pears sector cannot benefit from the low eco-toxicity levels that seed treatment neonicotinoids exhibit. Although I could likely go on and on about the different things and sites being mobilised in the apples and pears sector compared to the arable sector, there are just two interesting things which significantly impact the circulation of this network and the discontinuation of neonicotinoids in the apples and pears sector.

These things are honeybees and maximum residue levels (MRLs). Although honeybees are mobilised in the arable sector, they are mobilised here in a different loop of the network representation and to a much different effect. More specifically, the mobilization of honeybees by consumers are leading to the mobilization of MRLs by supermarkets which is subsequently impacting the way in which neonicotinoids are being used in the apples and pears sector. So as economists spit out reliable data thanks to a new survey (Latour, 1999a, p. 101), consumers of NZs apples and pears, particularly those in Europe can now demand lower use of agricultural compounds in the production of their food, thanks to regulatory changes, transparency of supply chains and consumer trends. More specifically and getting back to the core tenant of this thesis, European consumers are able to mobilise the regulatory changes around neonicotinoid use in Europe and the effects which neonicotinoids have been shown to have on honeybees (see section 2.6) to demand that neonicotinoids be removed from the production of their food. Immediately this brings together multiple sites of this controversy, the consumer's home and the supermarket. Immediately this demand, these regulatory changes, the effects on honeybees and global trends are translated by international supermarkets into a form which renders itself immediately useful in arguments with the apples and pears profession. This form is MRLs or the maximum levels of chemicals allowed on fresh products by supermarkets. Consumers of NZ's apples and pears and therefore international supermarkets do not favour neonicotinoids. Therefore, they implement a zero-neonicotinoid residue level to influence the agricultural practices occurring in NZ's apple and pear orchards (see section 5.2). Essentially if NZ's apples and pears do not meet the quality assurance standards implemented by the international and domestic supermarkets, they will not be accepted. How this mobilization of MRLs circulates through the "autonomization" loop and then the rest of the network will be discussed below.

Links and knots

Once again the "links and knots" draw our attention to what is ordering, managing and collecting together the world described in the other four loops of the network, that world composed of "data, colleagues, allies and spectators" (Latour, 1999a, p. 108). Like in the arable sector, the "links and knots" of the apples and pears sector are neonicotinoids as an agricultural practice. What has been demonstrated is that the area which manages, propagates, determines and implements neonicotinoids as an agricultural practice is much larger and more heterogeneous than one would initially recognize. Farming practices are not simply a determinant of farmer preferences, market pressure, or legislation. This Latourian representation of the machinations which deliver and determine neonicotinoid use in NZ's apples and pears sector suggests that it depends on the continued formation of a collective. This collective which links together various sites, from farmers' fields to a European household or a French supermarket to the offices of apples and pears NZ etc. Furthermore, this collective is dependent on the trade of resources, the translation of interests and the strengthening of relationships.

The apple and pear sector's blood flow

As explained in the blood flow of the arable sector, a slight change in any of the loops would stop the blood flow of the network and virtually kill neonicotinoids as an agricultural practice. So, while

neonicotinoids as an agricultural practice are very much embedded -and maintained- in the arable sector, it has been removed from NZ's apples and pears sector. This is due to differences in key relationships, actors, things and sites. This section describes those differences which have stopped the blood flow of the apples and pears sector, which subsequently lead to the removal of neonicotinoids as an agricultural practice.

The circulatory system of scientific facts suggested that five activities, which have simultaneously been performed and worked, have embedded and maintained neonicotinoids in NZ's arable sector, while slight differences to them have led to neonicotinoids being removed from NZ's apples and pears sector. In both networks these activities are; the creation of neonicotinoids as an agricultural practice (links and knots); the making of things, documents, sites and other non-humans available for arguments about neonicotinoids (mobilization of the world); the forming of an invisible college who all share the same goal (autonomization); making "alliances" with important and powerful actors (alliances); and aligning the agricultural practices with the demands of consumers (public representation). Unlike the arable sector, neonicotinoids have not become embedded here, suggesting that some part of the network has not worked in harmony with the rest of it. Instead, it has disrupted or changed the blood flow, which was maintaining neonicotinoids.

Although it is not normal to follow the blood flow of a Latourian network by beginning at the "public representation", for the purpose of demonstrating how neonicotinoids have been removed from the apples and pears sector this is a suitable place to begin. In interviews with members from both sectors (Chapter 5), it was very apparent that both arable and fresh product growers are answerable to the demands of the consumer. As explained in the blood flow of the arable sector, a generational separation and a translation of consumer demands by international seed companies contributed to the circulation of specific interests which led to the embedding of neonicotinoids. On the other hand, the apples and pears sector has no generational gap, and the demands of the consumer are slightly different. Rather than just demanding fresh food on their shelves every day, consumers of NZ apples and pears want to know what practices are contributing to the production of their food. Moreover, they want to make sure that chemicals which they are concerned about and are banned where they live are not being used in the production of the food which they eat. Just like how members of the arable profession translated the demands of their consumer and aligned their interests to meet those demands, the apples and pears profession has done the same. This is because meeting consumer demands is a major concern of agricultural communities. What results from this is a connection between two sites of this network. Supermarkets are directly connected to the consumer, through the transfer of resources, consumers give payment in the form of money to supermarkets and receive fresh products which are free from unwanted agricultural chemicals, like neonicotinoids, in return. As stated earlier, it is the supermarkets which translate this demand, fresh

products free of neonicotinoids, into something which is understandable and aligns with the interests of the apples and pears profession, that interest being market access and sale of product. What this demonstrates is that changes in the interests of the public have changed the way the "public representation" loop is acted out: a lack of generational gap and the interest of the public in consuming food free of agricultural chemicals, particularly neonicotinoids, alters the role of the market (here the supermarket, in the arable sector the international seed market/companies).

If we continue to follow how this different consumer demand moves through the network, altering it and stopping the neonicotinoid blood flow as it goes, we next come across the international and domestic supermarkets and how they translate this demand. In order to meet those demands of their consumers and maintain the supply chain of NZ apples and pears, those supermarkets which are a part of the apples and pears profession make these demands mobile by converting them into MRLs. These MRLs are immediately useful in arguments with their colleagues or those other actors in the "autonomization" loop (Latour, 1999a, p. 101). As stated, earlier supermarkets like to control their own regulation and quality assurance programs by implementing strict standards which products must meet. These standards act as an industry-standard which individual producers must meet if they wish to gain market access and sell their product. So, the question of what agricultural practices should be used in the production of apples and pears is not just a focus of fresh product growers and apples and pears NZ but a focus of supermarket quality assurance standards that pay close attention to consumer demands. The mobilization of MRLs are likely the most significant difference between the two networks, and if I was to pinpoint the cause of neonicotinoid removal in the apples and pears sector, I would say it was here.

The mobilisation of MRLs by supermarkets are serving to directly destabilise neonicotinoids. From this point, the entire network begins to break down and the circulation of relationships, things and sites that embed neonicotinoids halts. Instead, what circulates through the "autonomization" loop is quite the opposite. If for no reason other than to maintain market access, apples and pears NZ has translated these MRLs into the crop protection plans and advice which they issue growers. Simply they remove neonicotinoids from the crop protection plans, introducing an alternative and advising growers about changes in supermarket regulation and consumer demand. Apple and pear growers have responded to this by doing exactly as advised, removing neonicotinoids from their crop protection programmes and implementing the recommended alternative. Essentially fresh product growers have "tended to move away from them [neonicotinoids] and look for alternatives" (Product Group Representative 1). Therefore, the differences in consumer demand and the subsequent mobilization of MRLs has led to the removal of neonicotinoids in NZ's apples and pears industry. Even though this was a minor change in the "public representation" loop of the network, it rippled through the "autonomization" and "mobilization of the world" loops.

There is not much more to say about the removal of neonicotinoids in NZs apples and pears sector. The pumping heart at the centre of this network, or the "links and knots" tying it all together, has changed because the blood-flow no longer circulates, neonicotinoids no longer circulate. I will explain how these changes circulate through the remainder of the network all though it seems impractical and irrelevant to do so. It is unlikely that this would change the "alliance" between landowners and beekeepers at all because as explained by one beekeeper it does not matter what insecticide a landowner is using "it's still the same end-product for us. You're still killing insects" (Beekeeper 1). Furthermore, this relationship was never dependant on the specific use of an agricultural chemical. Instead, all that mattered, for both parties, was an exchange of resources and hopefully some level of protection for the bees. But the removal of neonicotinoids will not affect the price of Manuka honey, the cost of pollination or the sensitivity of this relationship, so beekeepers still will not complain because of any potential risk of losing their income. There is potential for the removal of neonicotinoids to trigger some complaints, but complaints about another chemical. Because beekeepers or at least those interviewed as a part of this thesis are more scared about what you would replace neonicotinoids with than the neonicotinoids themselves (see section 5.3.6).

In terms of the "alliance" which apples and pears NZ hold with the NZEPA, one could speculate that if there is any future call for information about neonicotinoids or a future reassessment of this class of insecticides, the lack of benefit and critical use cases from this highly valuable sector might contribute to their eventual banning. What I am suggesting here is that in the past this sector would have put forward benefit cases to defend neonicotinoids, but since they no longer use them and it has effectively been removed from agricultural practices, it is unlikely that apples and pears NZ would invest resources into helping defend a compound which they do not use, and which their consumers do not want used. It is more likely that their defence and commitment might relocate to another chemical. It is likely that the relationship between these two actors remains, but no longer to embed neonicotinoids. This is because there remains the issue of there not being enough tools in the toolbox (see section 5.2.2).

So what does this all mean? Well, at the outset of this thesis, I thought neonicotinoids were heavily embedded in all NZ's agricultural practices. What was being presented in the media at the time was rhetoric suggesting NZ's entire agricultural production and therefore, the economy, relied on this insecticide. What I have demonstrated here is that this is not actually the case. I have shown that NZ's agricultural industry cannot be considered as a single case study in terms of agricultural practice or hazardous substance research and I have also demonstrated that neonicotinoids are at various stages of making and maintenance depending on which agricultural sector you look at. More specifically neonicotinoids as an agricultural practice are very much embedded -and maintained- in the seed-producing and arable sectors of this country while at the same time it has been removed

from those fresh product sectors selling directly to international supermarkets, like the apples and pears sector.

6.4 What has the CSSF taught us about agricultural practices

During this research, I used Callon's key principles (1986) and Latour's CSSF (1999a) to answer the primary objective, how have neonicotinoids remained embedded in NZ's agricultural practices. To answer the fourth and final research question of this thesis: How can Latour's CSSF be used to investigate similar controversies in the future, in particular, those which involve hazardous chemicals used in NZ agriculture?; this section discusses how this tool helped deepen an understanding of hazardous chemical and agricultural practice controversies by bringing forward some aspects obscured in previous studies. During the presentation of the interview data (Chapter 5) and its subsequent analysis (Chapter 6), it has emerged that the making and maintenance of neonicotinoids, cannot be solely described as being a scientific enterprise (Ingram, 2007; Warner, 2016) or coming down to individuals' decisions, motivations and interests (Bangura, 1983; Lalani et al., 2016). Instead, it should be seen as having a collective relational nature, whereby each relationship is defined by the trade of material intermediaries (Callon, 1991, p. 134). Therefore, this section describes the main avenues in which Latour's CSSF (1999a), helped in understanding neonicotinoid use.

Previous studies have explained agricultural practices as being driven through economic, physical and environmental factors as well as by the values of individual farmers (Lalani et al., 2016). This study holds some underlying idea that the embedding, change or uptake of agricultural practices can be explained solely by the motivations, principles and actions of individuals. By doing this and adopting an individualistic point of view, it has overlooked the collective and relational nature of agricultural practices, which is obvious in this thesis. This collective and relational nature, is exactly what the CSSF has demonstrated in this thesis, neonicotinoids are made and maintained not just by the individual decisions of farmers, but it is dependent on the product being produced, the development of material relationships (Latour's 5 loops), the presence or absence of non-humans and the trade of intermediaries to translate meaning and interest powerful actors. By attempting to explain agricultural practices as being caused by isolated motivations, Lalani et al. (2016) has oversimplified the complexities behind agricultural practice decision making. By using the CSSF, this case study has highlighted the complex relational and collective nature of agricultural practices and moved away from the traditional explanations of action. Agricultural practice decision making must now be treated as a relational phenomenon which contradicts the view that farmer motivations and characteristics determine agricultural practices (Bangura, 1983). The CSSF has helped bring forward the relational nature of agricultural practices, doing away with the individualistic approach normally associated with studies of agricultural practices (see section 3.4.1). The relational focus of this

research is not restricted to humans but includes non-humans like grass grub and MRLs. Callon's key principle of "generalised symmetry" insist that non-humans are brought to the fore and are acknowledged as being active in relationships. By utilising these tools, I have observed how nonhumans are active partners in some relationships. Subsequently, I have been able to describe how grass grub is contributing to the embedding of neonicotinoids in NZ's arable sector, and MRLs are contributing to the removal of neonicotinoids in the apples and pears sector.

The CSSF interpretation of neonicotinoid use (section 6.2 and 6.3) shows how the trade of resources (intermediaries) has influenced this agricultural practice in many different ways. For example, it shows how neonicotinoid use can vary between sectors, which ultimately depends on the material relations that connect growers to their markets and the presence or absence of particular non-humans. Rather than describing the embedding of neonicotinoids as being determined by the actions of particular individuals, I have followed Callon's key principles and used the CSSF to give greater importance to the relations between actors. Subsequently, I found that neonicotinoids are only used following successful negotiations between several actors. By adopting ANT, I was encouraged to view agricultural practices as being formed through the development, maintenance and break down of relations between actors. Subsequently, it is not accurate to describe agricultural practices as being isolated in action. Instead, they are in a continuous flux, shaped by the relational processes occurring in and out of NZ's agricultural communities. Agricultural practices, from a CSSF perspective, are shaped by the continuous negotiations, trade of "interessement devices" and resources that happen between actors over time. These processes subsequently strengthen and weaken the relationships between actors that embed neonicotinoids.

There is growing pressure for agricultural practices to change (Ingram, 2007), yet the factors driving agricultural practices remain poorly understood. Because Latour's CSSF (1999a) brings what is commonly called the "context" into the material relations that must be traced (pp., 98-99). The case study, or should I say case-studies, central to this thesis suggests that the context surrounding agricultural sectors play an influential role in shaping the on-going use -or discontinuation- of agricultural practices in each sector. Aspects such as political, ethical, environmental and scientific contexts help shape agricultural practices (Ingram, 2007; Warner, 2016). In line with these studies, the CSSF has improved an understanding of how the context surrounding an agricultural sector influences the making and maintenance of neonicotinoids in NZ. My research suggests that the relationship between the agricultural profession, their markets and the public is particularly important when identifying what shapes agricultural practices. For example, in the apples and pears case study it is a change in the public representation loop which has led to neonicotinoids being removed from agricultural practices. Like studies which suggest that looking beyond the boundaries of the laboratory can provide new light to understand the making and maintenance of scientific facts

(Edwards, 2014; Warner, 2016), this research suggests that the adoption of agricultural practices is better understood by looking beyond the boundaries of a particular community.

The CSSF did not explain the embedding of neonicotinoids as solely coming down to the need to deal with pasture pests across NZ, nor was it explained solely by the need to meet high yield requirements. Instead, the CSSF revealed that neonicotinoids use is being shaped by the relations developed between actors within and across networks, and the context surrounding NZ's agricultural sectors. For example, NZ's arable sector is one generation away from a final consumer, and its public can be considered those fresh product growers in Europe who buy their seed. That public is demanding the production of high-quality seed that is free of pests, weeds and diseases. A series of translations of this demand circulating through various relationships results in NZ's arable growers using neonicotinoids to minimise the vagaries of growing outdoors and secure market access.

On the other hand, there is no generational gap between the apples and pears sector and the final consumer, who are consumers of NZ's apples and pears in Europe. This public is demanding minimal use of agricultural chemicals in the production of their fruit and have a stigma against neonicotinoids, because of the bans in Europe. Again, through a series of relationships, translations and mobilizations of non-humans, apple and pear growers have been connected to their market and are shaping their agricultural practices to meet those demands of the public, they have removed neonicotinoids from their crop protection programmes.

6.5 Overcoming the limitations of the key principles of ANT and the CSSF

The previous section showed how this study has provided an alternative interpretation of hazardous chemical use in NZ's agricultural practices, particularly surrounding those which are very controversial. I argued that the CSSF (1999a) is a powerful theoretical tool for demonstrating and bringing forward the material, relational, historically shaped and evolving nature of agricultural practices, like the use of neonicotinoids. This section follows on by providing insight into how this research was shaped by the CSSF and Callon's key principles (1986). It also discusses how I was able to overcome some of these tools' limitations.

6.5.1 Responding to general critiques of ANT

Callon's key principle known as "general symmetry" demands that agency be ascribed to human and non-human actors (Callon 1986). It is argued that humans and non-humans must all be seen as active entities with an ability to "exert detectable influences on others" (Law, 1987, p. 132). However, this principle has been criticised by many who believe that the adoption of a symmetrical view "degrades our understanding of action by obscuring the fact that it is only through the intervention of humans that agency - and thus political transformation of social arrangements - can occur" (Whittle and

Spicer, 2008, p. 620). As Latour (1983) demonstrated how the development of an anthrax vaccine in 19th century France is made possible by Pasteur's ability to bring together diverse human and nonhumans elements, I have demonstrated how neonicotinoids are embedded in NZ's arable sector because of a similar coming together of various humans and non-humans. Thus, by following the principle of "generalised symmetry", I have demonstrated how neonicotinoids are embedded -and remain- in NZ's arable sector. Essentially this principle enabled me to demonstrate how non-humans like grass grub, MRLs, seed treatment neonicotinoids and honeybees had agency in both the arable sector and the apples and pears sector networks. Like Dankert's (2011) assertion that "humans first shape the buildings and then are shaped by the same buildings", MRLs, for example, are shaped by supermarkets and then shape the action of apples and pears NZ as well as fresh product growers. In response to Whittle and Spicer's (2008) critique of "generalised symmetry", I say this, first and foremost ANT network representations are "about how to study things" (Latour, 2005, p.142), they have never professed to be about "the social" (Latour, 1999b, p.22). Therefore, by avoiding the hierarchies that privilege human agency, the CSSF in conjunction with Callon's key principles (1986) has allowed me to acknowledge and describe how things, as well as humans, are embedding neonicotinoids in NZ's arable sector and killing them in NZ's apples and pears sector.

Next, Callon's principle of "agnosticism" has been criticised for ignoring social demographics and power structures like racism and patriarchy (Bloor, 1999; Harding, 1992, 1998, 2008; Restivo, 2010). This mainly comes down to "agnosticism" rejecting the existence of any a priori attribution of scale to social entities, which therefore assumes that "networks are immersed in nothing" (Latour, 1999b, p. 128). Essentially Callon's principle of "agnosticism" is critiqued because it ignores social demographics and power structures like racism and patriarchy at the outset. This radical position suggests that society does not exist only actors and their relations exist (Latour, 1996; Munir and Jones, 2004). Subsequently, aspects which might have characterised the arable and apples and pears sectors were not appropriately illustrated through the CSSF (Latour, 1999a) when informed by Callon's key principles (1986). Although the CSSF helped me identify the relevant actors of each agricultural sector, it became very clear through the data collection and thematic analysis (Chapter 5) that more than just those actors identified, and their relationships influenced the use of neonicotinoids. For example farmers individual taste for crop protection practices (see section 5.2.5), reps desire to sell chemicals (see section 5.2.4), the resources required for the NZEPA to complete a reassessment and the resources required for product groups to find an alternative (see section 5.2.2), also influenced neonicotinoid use. Therefore, the strict use of the CSSF to analyse the data in sections 6.2 and 6.3 meant that these aspects were obscured and not discussed because the CSSF's loops are not sensitive to aspects such as individual preferences and the availability of resources. However, I was able to avoid a complete faux pas by employing a thematic analysis of the data.

6.5.2 Avoiding the objective and un-reflexive approach when following the actors

In his book "After Method" John Law (2004) states that researchers need to ask if they are able or willing to recognise that their methods also craft realities. Essentially, we should act in self-reference which

...requires an awareness of the researcher's contribution to the construction of meanings throughout the research process, and an acknowledgement of the impossibility of remaining 'outside of one's subject matter while conducting research. Reflexivity then urges us "to explore the ways in which a researcher's involvement with a particular study influences, acts upon and informs such research. (Nightingale and Cromby, 1999, p. 228).

Law (2004) continues by saying that this act is not self-indulgent, but possible and necessary to do, especially in a world which has multiple 'goods'. These 'goods' are the truths and non-truths, realities and non-realities, presences and absences, but also arrangements with political implications which a collection of methods might create (p. 143).

[research methods] have effects; they make differences; they enact realities; and they can help to bring into being what they also discover (Law and Urry, 2004, pp. 392-393)

Therefore Law (2004) challenges researchers to ask the questions: what does standard method assemblages silence? Which possible realities does it refuse to enact in its dominant insistence on that which is smooth? And how might it be crafted differently?

Network representations can act as a lens to view the world as a network of relations, distributed and networked equally between human and non-human actors (Latour, 1987, 2004). The way network representations distribute agency has been criticized because it is "never based on the reflexivity of the actor itself but depends wholly on the researcher or one voice in the field that speaks for the network. This voice becomes the only truth" (Jansen, 2016). Therefore, what follows is an account of my attempts to be self-reflexive, in doing this I hope to somewhat make up for this inevitable faux pas of using a network representation to make sense of reality. I also hope to shed light onto how I ascribed agency and followed the actors and finally alarm any future researchers of the traps I fell into.

Another relevant critique argues that network representations like Latour's CSSF (1999a) and research which uses Callon's key principles (1986) adopt an objective stance (Murdoch, 2001). This is because they tend to impose their "own theoretical lexicon" (Whittle and Spicer, 2008). The interpretation of the networks presented in this thesis by no means matches the world of the interview participants. This is largely because the vocabulary being used is vastly different from that used by the interview participants. Which can result in research, utilising the CSSF and Callon's key

principles, offering superior views and assuming that other explanations are wrong (Whittle and Spicer, 2008).

To combat these critiques, I provided a thorough description of the case and the participants. I also presented many quotations which give insight into my translation of the interview participants' responses. Furthermore, to ensure various perspectives were considered and therefore, a fair representation of the network given, interview participants from across NZ's agricultural and horticultural industries were considered. On top of this, interview participants were chosen from across what one described as a bell curve of agricultural practices (Arable Grower 3). This bell curve is regarding the involvement of landowners in the daily agricultural practices occurring on their property. At one end are the landowners who do everything themselves and at the other are those who rely entirely on someone else. There was a conscious effort to avoid using vocabulary from Callon (1986) and Latour (1999a) in the interview questions, so to avoid imposing this same vocabulary on the participants. By using a list of pre-determined questions, a fair representation of each actor was gauged, and the imposition of my views was avoided.

Another danger of using these two tools is the risk of applying them unreflectively in order to verify their universality. Whittle and Spicer (2008), discuss this in terms of Callon's (1986) four-stage model of translation, however, I can see how the same critique may be applied to Latour's (1999a) CSSF. Like how the four steps of translation might be good for interpreting scallop fishing, so might the CSSF be good for interpreting research on the potential of a nuclear chain reaction in uranium (Latour, 1999a). Nevertheless, in both cases, it would be problematic to assume that they can be applied to any setting (Whittle and Spicer, 2008). Essentially there is a danger that new studies using these models "are reduced to a series of deductive tests that confirm or refute" either framework, rather than "being a process of inductive theory generation theory that is grounded in and emergent from the empirical data" (p. 618). What results from this is that users of these network representations present their own interpretations as the truth, while treating those of others as relative (Whittle and Spicer, 2008). Unfortunately, I fell into this trap, by forcing the data into those categories, or loops, described by Latour (1999a). This only occurred after some effort to try and explain the empirical complexities of each case which I observed. However, due to time limitations, I conceded to pursuing the simpler and less accurate analysis of neonicotinoids. Rather than taking the CSSF (Latour, 1999a) and departing from its loops, as Edwards (2014) has done, I squeezed the relationships and actors I observed into the preconceived model.

It is the actors in the network who determine how the phenomena will proceed. Essentially the actors will set the research agenda and the researcher follows the actors (Latour, 1987). Therefore, the topic for a research project may have arisen because of the misinterpretation of the object of

study, e.g. my predisposition as a beekeeper may have meant I saw a controversy around neonicotinoids where one did not exist. In this case, it would make little or no sense to continue with the research. Furthermore, if the participants had not heard of or did not know about neonicotinoids, it would also make little sense to continue with the research project. During this thesis, this situation did arise in some circumstances. During my data collection, I found that the interview participants were either experts on neonicotinoids or knew very little about them. Rather than changing the actors whom I followed or changing the research topic, I focussed more on the broader concept of how agricultural practices become embedded in NZ, using neonicotinoids as a detailed case study throughout the thesis.

I attempted to remain a passive observer outside of the network throughout this research. However, I quickly found out that actors react to being observed and followed. In one situation a NZ based seed and chemical company representative, who had agreed to be interviewed, pulled out due to the sensitivity of the neonicotinoid controversy. Fortunately, this reaction only happened in one circumstance, and it was not significant enough to result in any flow-on effects. This research was fortunate in that the size of the network and the sheer number of potential interview participants meant that I could study the phenomenon without my presence influencing the network itself.

To compensate for this participant pulling out of interviews, I not only found press releases, newspaper articles and other written information related to them, but I also interviewed individuals around that NZ seed and chemical company representative, asking questions about their relationship with them. A lot of the information I uncovered was directed to specific audiences and subsequently had a positive message regarding neonicotinoids. The large variety of information sources which I used had the added benefit of triangulating my interview data. Subsequently, I was able to continue following the actors. However, sometimes this was at a distance rather than through direct interviewing. This has illustrated the idea that there is no divide between subject and object (Latour 2012). Like Edwards (2014), I have highlighted this lack of divide as an important methodological issue that likely affects how research is conducted. Essentially researchers can be followed, enrolled and mobilised in the networks which they study, and therefore they cannot simply follow the actors.

6.5.3 Inclusion and exclusion of actors

When addressing the directive to follow the actors, how does one go about identifying who and what to include, and how does one know when to stop following? Therefore, this section discusses the issues I came across when deciding what actors to include and exclude from this thesis.

A significant issue I came across at the outset of this research was that ANT literature does not really offer strategies for defining research boundaries. Furthermore, it offers little direction on how to

101

decide what to include in the network? Therefore, before I could even begin thinking about my data, I came across two issues, where to set the boundary and when to stop looking for actors. In this research, a small change in one agricultural sector was enough to change the boundaries. It has been suggested by Bloomfield and Vurdubakis (1999) that while researchers using Callon's key principles (1986) must try to remain impartial and consider all actors, ultimately the process of selecting appropriate actors to study is dependent on the suppositions about what actors exist and their relative positions within possible networks. It has also been noted that networks continue to evolve and translate as some actors join, and others leave. Because Callon's principles (1986) require researchers to begin without preconceived notions, the researcher can include actors without predetermined criteria.

Therefore, one of the key issues is that previous works which utilise network representations rarely provide the boundaries of a research project. Subsequently, it is up to the researcher to select the paths he/she wishes to follow (including which actors) and to choose those which will be ignored. Most researchers using network representations likely have a much easier task at setting those boundaries than I had during this research. This is because network representations are normally applied to smaller-scale case-studies, like Edwards (2014) which focussed on a specific research project. Therefore, due to the scale of this case study, I had a much more difficult task.

Subsequently, I used Latour's CSSF (1999a) to set boundaries, and each loop directed me to look in a certain direction for the actors which I eventually found. As explained in Chapter 4, I utilised the loops illustrated by Latour in his CSSF (1999a) as a way of identifying actors. He describes the loops as being "5 types of activities that science studies needs to describe first if it seeks to begin to understand in any sort of realistic way what a given scientific discipline is up to"(p. 99). I set out already knowing that the "links and knots" of both networks were neonicotinoids and initially approached growers/landowners as they are the ones who use this chemical in their crop protection programmes. From here, I asked specific questions targeted at uncovering whom they held relationships with, in terms of neonicotinoid use, and what materials were traded in these relationships. From there, I followed neonicotinoids as they were translated, transformed and mobilised through the network.

Of course, this approach is imperfect. Latour (1993) tasks researchers to follow the actors, which I attempted to do. But this resulted in me continually finding more and more relations that may have contributed to the embedding -and maintenance- or removal of neonicotinoids. Subsequently, I had to be disciplined in deciding which actors and relationships were important. It was relatively easy to know when to stop interviewing participants. I stopped at the point where I did not hear anything new. However, this did not mean that I explored all the relationships mentioned in the interviews.

For example, Regulator 2 mentioned a relationship between the NZEPA and the Cabinet of NZ. Predominantly, I made decisions about what actors and relationships to include based on accessibility to data and how frequently it was referred to by the interview participants. For the most part, I heard about the same themes, relationships and actors during the interviews and the most common of these were discussed in Chapter 5, determining the discussions of sections 6.2 and 6.3. Furthermore, I made some decisions to exclude actors and relationships based on their perceived distance from the controversy at hand. For example, Arable Grower 3 spoke of how click-bait articles on social media are affecting the perception of the public about farming practices and how this is being exacerbated by global urbanisation. Not only were they the only participant to mention this relationship, but if I explored this, I would have to look at all those relationships which affect public perception of neonicotinoids.

This is by no means a perfect way of setting boundaries, particularly when analysing a case study or case studies that might impact most living things on this planet (Chapter 2). Subsequently, I must state that the results produced from this research can still be regarded as preliminary. An unfortunate but inevitable result of this limitation is that I may not have succeeded in realising the network and may, therefore, have produced an incomplete or misleading research story.

6.6 Summary

This chapter and more specifically sections 6.2 and 6.3 have answered the first three research questions: Who and what are the major actors that influence the scale and frequency of neonicotinoid use in NZ and what have been their roles? How have certain relationships formed between actors? How are neonicotinoids as an agricultural practice embedded -and maintained- in NZ, and why has opposing knowledge been unsuccessful in removing it?

The data which was collected and presented in Chapter 5 has been analysed here using Latour's (1999) CSSF. This enhanced my understanding of neonicotinoid use in NZ and more specifically the actors, relationships and materials which have led to them becoming embedded in the agricultural practices of NZ's arable sector and removed from NZ's apples and pears sector. These findings have suggested that the use of neonicotinoids as an agricultural practice in NZ is a very complex phenomenon that cannot be described as being solely shaped by scientific research or by the decisions, motivations and interests of individuals. It should instead be seen as having a collective relational nature. Not only has this analysis revealed that neonicotinoids are not as embedded in NZ's agricultural practices as previously thought, but its use and embedding depends on which agricultural commodity is being produced.

While on one hand neonicotinoids are embedded -and maintained- in the agricultural practices of NZ's arable sector they have effectively been removed from the apples and pears sector. NZ's apples and pears sector have removed neonicotinoids from their pest control programmes due to a change in the "public representation" loop and its circulation through the other loops of the network. Essentially, the apple and pears profession has responded to the international markets unfavourable view of neonicotinoids by restricting residue levels allowed on fruit and subsequently removing them from crop protection programmes.

Sections 6.4 and 6.5 have also answered the fourth question: How can Latour's CSSF (1999) be used to investigate similar controversies in the future, in particular, those which involve hazardous chemicals used in NZ agriculture? This has been achieved by outlining how the CSSF (Latour, 1999a) and Callon's key principles (1986) have contributed to my understanding of neonicotinoid use. Furthermore, I have outlined the issues which I came across while using these tools, as well as their common critiques. In doing this, I have described how I overcame those issues and combatted the critiques. Simultaneously, I have acknowledged where the boundaries and limitations of my research lie and admitted where I went wrong, offering some direction and solutions for future research. The following chapter will offer some examples of how this research has contributed to the literature as well as the ever-growing body of work considering agricultural practices.

Chapter 7

Conclusion

7.1 Summary of research findings

The overall aim of this thesis was to understand how neonicotinoids have become embedded -and remain- in NZ's agricultural practices. Subsequently, four research questions were identified and developed to reach this primary goal. The main findings of this research are presented below and are organised according to those research questions.

My first task was to identify and describe the major actors influencing neonicotinoid use in NZ. This began with an extensive review of publicly available documents which identified actors concerned with the use of neonicotinoids in NZ. The CSSF (Latour, 1999a) also help direct me towards those actors involved in the making and maintenance of agricultural practices. Once I had performed the initial interviews, I also used snowball sampling to try and encourage each participant to direct me towards individuals whom they thought would suitable for this research.

Some of the actors which this process identified were endemic insect pests whose abundance, and affinity to modified pastoral landscapes, have contributed to NZ's reliance on agricultural chemicals like neonicotinoids and are used by other actors to convince one another of the importance of neonicotinoids. Next, there are landowners, they are the ones using seed treatment neonicotinoids on their properties to combat the vagaries of growing outdoors, meet the expectations of their consumers and maximise their yields. There are also beekeepers, who endure the adverse effects of neonicotinoids so that they can maintain relationships with landowners and secure access to valuable bee sites. Therefore, they are unlikely to complain if their hives are poisoned. Next, there are seed and chemical company representatives and product group representatives, these actors form strong relationships with international markets to secure market access for landowners, while simultaneously providing landowners with crop protection programmes, which include neonicotinoids, to ensure that a crop is produced. They also try and convince regulators not to ban neonicotinoids by demonstrating to them that neonicotinoids are vital for the production of key export crops. There are also international and domestic markets, which demand neonicotinoids are used in the production of hybrid seed crops and which self-regulate chemical residues on food. Lastly, there are the consumers, these actors demand that high yielding crops and food, which are free of pests, weeds, diseases and in some cases chemical residues, are available all year round.

My next task was to describe how certain relationships were formed and maintained between those actors. The interview questions attempted to draw out empirical evidence regarding neonicotinoids and encouraged the interview participants to discuss their relationships with the other actors listed above. This identified several relationships centred on the use of neonicotinoids in NZ's agricultural practices. To illustrate, analyse and discuss these relationships, Latour's CSSF (1999) was once again applied to the research. In summary, I found that neonicotinoids are embedded -and remain- in NZ's arable sector because of a series of material relations between arable growers, NZ seed and chemical companies, international seed companies, beekeepers, the NZEPA and consumers.

Firstly, what I have called the "arable profession" (autonomization) is formed by a three-way relationship between international seed companies, NZ seed and chemical companies and arable growers. International seed companies who demand high yielding and 100% seed crops interest NZ seed and chemical companies with payment and continued contracts to grow hybrid seeds in NZ, in return the NZ seed and chemical companies guarantee the production of 100% and high yielding seed crops by offering NZ's more relaxed rules when it comes to some substances deemed hazardous elsewhere (neonicotinoids). The NZ based seed and chemical companies subsequently mobilise crop protection programmes, payment and access to markets to influence the farming practices of NZ's arable growers. To secure their position NZ arable growers then mobilise a commitment to follow those crop management plans and use programs such as Agworld to confirm they have performed crop treatments as directed. Another relationship exists between arable growers and beekeepers (alliance). Essentially landowners convince beekeepers of their agricultural practices (use of neonicotinoids) by mobilising formal contracts, access to stances of highly valued Manuka and a fee for pollination services. Beekeepers similarly pay for access to honey-producing flora to interest arable growers in this relationship. Subsequently, due to the growth of NZ's honeybee population and the Manuka honey market, those beekeepers with access to Manuka are unlikely to complain even in cases where their hives have been poisoned, this is due to the risk it poses their income. Next is the relationship is between the NZEPA, arable product groups and arable growers (alliance). Once again, this relationship relies on the trade of resources; product groups collect levies from growers based on yield, land area or at a fixed rate and in return and growers receive marketing, research and development as well as some practical support in return. Arable product groups also provide lobbying of the NZEPA, convincing them that neonicotinoids are essential for the success of key export crops, by mobilising critical use cases. Subsequently, the NZEPA is convinced of neonicotinoids importance to NZ's agricultural production and are also convinced that biodiversity declines are not caused by neonicotinoids. In return the NZEPA do not changed neonicotinoid regulations. Finally, there is a relationship between the users of NZ grown hybrid (consumers) seeds and international seed companies (public representation). The international consumers interest the international seed companies with payment and those companies provide the consumers with the seed.

106

This demonstrates that each of the actors had a different role in the embedding and maintenance of neonicotinoids in NZ's arable sector. Furthermore, the relationships between the actors are caused by their shared need to achieve their goals. Therefore, the actors are connected in a way that makes neonicotinoids indispensable for each of them to achieve their goals. It is in this regard that the actors make up the network, and their material relations serve to make and maintain neonicotinoids in NZ's agricultural practices.

My third question was to explain why knowledge opposing the use of neonicotinoids has been unsuccessful in removing it from NZ's agricultural practices. Like the above question, this was answered through the application of Latour's CSSF (1999a) in Chapter 6, but its answer began during the data collection process. While performing the interviews, I found that the context surrounding an agricultural practice was present in the material relations I traced and therefore significantly impacted the application of neonicotinoids. Essentially, the embedding of an agricultural practice cannot simply be explained by the scientific resources supporting or opposing its use. To this end, I found that even the commodity being produced, and the agricultural sector was affecting practices. It became apparent very quickly that neonicotinoids were embedded in NZ's arable sector but have been removed from NZ's apples and pears sector. Through the application of Latour's CSSF (1999a), I observed that a slight change in the "public representation" loop of the CSSF (Latour, 1999a, p. 103), circulated through all the other loops of the network and led to neonicotinoids being removed from the apples and pears sector's pest control programmes. In summary, consumers of NZ's apples and pears want to know what practices are contributing to the production of their food. Moreover, they want to make sure that chemicals which they are concerned about and are banned where they live (neonicotinoids) are not being used in the production of the food which they eat. Supermarkets who are directly connected to the consumer, through the transfer of resources, translate this demand into MRLs. Subsequently, to ensure MRLs are met and to maintain market access, apples and pears NZ have removed neonicotinoids from the crop protection plans, introducing an alternative and advising growers about changes in supermarket regulation and consumer demand. Apple and pear growers have therefore "tended to move away from them [neonicotinoids] and look for alternatives" (Product Group Representative 1). Thus, neonicotinoids have been removed from the agricultural practices of NZ's apples and pears sector. This subsequently contradicts my initial assumption that neonicotinoids are embedded -and remain- in NZ's agricultural practices.

Finally, I was charged with understanding how Latour's CSSF (1999) could be used to investigate similar controversies in the future, in particular, those which involve hazardous substance use in NZ's agriculture. To achieve this, in light of my research I critiqued and responded to critiques of Latour's CSSF (1999a) and Callon's key principles (1986). Firstly the principle "generalized symmetry" was vital for this research, by allowing me to highlight that MRLs are first shaped by international

107

supermarkets and then shape the crop protection programs produced by apples and pears NZ and therefore fresh product growers ending neonicotinoid use. Similarly, it enabled me to demonstrate the importance of grass grub in helping shape the crop protection programmes of NZ seed and chemical companies and therefore arable growers use of neonicotinoids. Secondly, by presenting the conflicting viewpoints of all the actors and quoting the exact words of the interview respondents, limiting my biased commentary, I further addressed critiques of "generalized symmetry" as well of those of "agnosticism" (see section 3.5). Essentially through the use of a thematic analysis I have still identified power relations and social hierarchies as they have emerged from the data. Next I addressed critiques of the CSSF's imposition of its "own theoretical lexicon" (Whittle and Spicer, 2008). Essentially the CSSF's interpretation by no means matches the world of the interview participants. This is largely because the vocabulary being used is vastly different from that used by the interview participants. To address this, I provided a thorough description of the case and the participants and presented many quotations which give insight into my translation of the interview participants' responses. I also made a conscious effort to avoid using vocabulary from Callon (1986) and Latour (1999a) in the interview questions, so to avoid imposing this same vocabulary on the participants. Most importantly, Chapter 6 offers some insight into the major limitations and boundaries of myself and the research. By reflecting on my own limitations and the limitations of the tools which I used, I hope that future hazardous substance researchers will do a better job. To fully answer this question, the next section comments on what this research has contributed to CSSF and agricultural practice literature.

7.2 Contributions of this study

One of the main contributions of this study was that by adopting Latour's CSSF (1999a) and Callon's Key principles (1986) I have been able to bring forward the relational nature of agricultural practices in NZ, which has been obscured in previous studies. Furthermore, it helped affirm the work of Ingram (2007) and Warner (2016) who demonstrate that the CSSF can be used to makes sense of agricultural practices rather than its more intended application of understanding how scientific research is done. This thesis has also built on the work of Ingram (2007) and Warner (2016) by taking some direction from Fitzsimmons (2004) and Warner (2008) by applying the CSSF to a phenomenon broader than a specific instance of scientific work or a specific project aimed at understanding the adoption of different agricultural methods (Ingram, 2007). Previous studies have explained agricultural practices as being driven through economic, physical and environmental factors and by the values of individual farmers (Ilbery, 1978: Bangura, 1983; D'Souza, 1993; Lalani et al.,2016). Although these works have located farmer decision making, or the making and maintenance of agricultural practices, in the social, economic and political-ecological context, they tend to preserve the idea that the farmer is the sole, autonomous decision-maker. However, as Ingram (2007) emphasises,

All technology requires institutional and social support; even so-called natural technologies like breast-feeding (and organic agriculture) are unsuccessful without teachers to educate new practitioners, experts to solve problems, and groups to retain information, gather knowledge, and provide social approbation. Even where groups overtly tout motivations beyond some objective will to know (e.g., a religious, economic, or ecological mandate), they engage similar strategies to selectively engage with the material world, to build legitimate scientific arguments, and to disseminate ideas (Ingram, 2007, pp. 301-302).

Therefore, a major contribution of this study is that it adds to the growing literature including Ingram (2007) and Warner (2016) which suggest that farmers are not necessarily free agents or unfettered actors deciding the daily agricultural practices occurring on their property. Indeed, my research resonates with their argument suggesting that various "economic, moral, political, and other relationships between humans and nonhumans" (Ingram, 2007, p. 301) affect the making and maintenance of agricultural practices like neonicotinoid use.

Another contribution of this research is its demonstration that the CSSF (Latour, 1999a) can be used as a means of assessing how an agricultural practice is made and maintained while highlighting what is influencing growers to use a particular hazardous substance. Essentially this thesis has shown how the CSSF can make sense of the network actors' roles in determining on the ground agricultural practices. For example, because of their goal to maximise yield and therefore profit, growers have become consumers of other's expert knowledge (spray programmes, etc.). This satisfies not only the goal of growers but also that of other network members as seen through the supply of seed crops. Therefore, this study has contributed to the body of literature which looks at how agricultural practices are made and maintained. Furthermore, this research offers guidelines for using the CSSF in order to identify research participants, follow the actors, and provide critical insight into a hazardous substance or agricultural practice controversy.

References

Adkins, J. (2016). What is Cooking with Kererū/Kūkupa Management in New Zealand? A historical review using tools from Actor Network Theory. (Unpublished master's thesis). Lincoln University, Lincoln, New Zealand.

AGCARM (2013). Fact sheet on neonicotinoids. Retrieved from http://piat.org.nz/uploads/PIAT_content/pdfs/Indoxacarb%20and%20other%20neurotoxins%20info/ Agcarm%20Neonics%20factsheet.pdf

AGCARM (2018). The AGCARM newsletter: Seed treatment contributes \$1.2 billion to NZ economy. Retrieved from http://agcarm.co.nz/wp-content/uploads/2019/12/Input-201811.pdf

AGCARM (2019a). Welcome to AGCARM. Retrieved from http://agcarm.co.nz/

AGCARM (2019b). Pesticides: Farmers, growers, and pet owners require ways to control pests. Retrieved from http://agcarm.co.nz/pesticides/

Agrichemical Trespass Ministerial Advisory Committee (ATMAC) (2002). *Agrichemical trespass ministerial advisory committee (ATMAC): Final report to the minister for the environment*. Retrieved from https://www.mfe.govt.nz/sites/default/files/agrichemical-trespass-mac-report-nov02_0.pdf

Ahnström, J., Höckert, J., Bergeå, H.L., Francis, C., Skelton, P., & Hallgren, L. (2008). Farmers and nature conservation: What is known about attitudes, context factors and actions affecting conservation? *Renewable Agriculture and Food Systems*, 24(1), 38-47.

Alasuutari, P., Bickman, L., & Brannen, J. (2008). *The SAGE handbook of social research methods*. London: SAGE Publications.

Allen, J. (2003). A question of language. In M. Pryke, G. Rose & S. Whatmore (Ed.), *Using social theory: thinking through research* (pp. 11-27). London: Sage Publications.

Allsopp, M. H., et al. (2008). Valuing Insect Pollination Services with Cost of Replacement (Insect Pollination Replacement). *PLoS ONE*, 3(9), e3128. https://doi.org/10.1371/journal.pone.0003128.

Alvesson, M. (2003). Beyond Neopositivists, Romantics, and Localists: A Reflexive Approach to Interviews in Organizational Research. *The Academy of Management Review, 28*(1), 13-33.

Ardern, S., et al. (2014). *Briefing on the health of bees: Report of the primary production committee*. Retrieved from https://www.parliament.nz/resource/ennz/50DBSCH SCR56864 1/02f9621efb9436bcf27cfeaa7bc1672a4d90293a.

Bangura, A. M. (1983). *Farmer motivation patterns in participating in adaptive crop research trials/demonstrations in Sierra Leone* (Doctoral thesis, Iowa State University, 1983). Retrieved from https://lib.dr.iastate.edu/rtd/8406/

Beard, C. (2015, November). *Honeybees (Apis mellifera) on public conservation lands: a risk analysis*. Retrieved from https://www.doc.govt.nz/Documents/science-and-technical/honeybees-on-public-conservation-lands.pdf

Beaulieu, A., Scharnhorst, A., & Wouters, P. (2007). Not another case study: a middle-range interrogation of ethnographic case studies in the exploration of E-science. Science, Technology & Human Values, 32(6), 672-692.

Becker, H. S. (1998). *Tricks of the trade: How to think about your research while you're doing it.* Chicago: University of Chicago Press.

Bennet, D., Kelly, D. & Clemens, J. (2017). Food plants and foraging distances for the native bee Lasioglossum sordidum in Christchurch Botanic Gardens. *New Zealand Journal of Ecology*, 41. 10.20417/nzjecol.42.1.

Benninghaus, C. (2012). Beyond constructivism?: Gender, medicine and the early history of serm analysis, Germany 1870–1900. *Gender & History*, 24, 647-676. 10.1111/j.1468-0424.2012.01700.x

Beston, A. (2017). *Intensive farming bad news for native bees*. Retrieved from https://www.auckland.ac.nz/en/about/news-events-and-notices/news/news-2017/08/intensive-farming-bad-news-for-native-bees.html.

Bijker, W. E. (1995). *Of bicycles, bakelites, and bulbs: toward a theory of sociotechnical change*. Cambridge, MA: The MIT Press.

Bloomfield, B. P. & Vurdubakis, T. (1999). The Outer Limits: Monsters, Actor Networks and the Writing of Displacement. *Organization*, 6(4), 625-647.

Bloor, D. (1999). Anti-Latour. Studies in History and Philosophy of Science Part A, 30(1), 81-112.

Bonmatin, J. et al. (2014). Environmental fate and exposure; neonicotinoids and fipronil. *Environmental science and pollution research international*, 22. 10.1007/s11356-014-3332-7.

Booth, R. G., et al. (2016). Actor-Network Theory as a sociotechnical lens to explore the relationship of nurses and technology in practice: methodological considerations for nursing research. Nursing Inquiry, 23(2), 109-120.

Bourdôt, G., et al. (2019). *Pastoral sector weeds research strategy (2018 - 2028)*. Retrieved from https://www.agresearch.co.nz/assets/Uploads/agresearch-pastoral-weeds-research-strategy-2018-2031.pdf.

Braun, V. & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative research in Psychology*, 3(2), 77-101.

Briers, M. & Chua, W. F. (2001). The role of actor-networks and boundary objects in management accounting change: a field study of an implementation of activity-based costing. Accounting, Organizations and Society, 26(3), 237-269.

Brockie, B. (2007). Native plants and animals – overview. Retrieved February 18, 2020 from https://teara.govt.nz/en/native-plants-and-animals-overview.

Brooking, T. (2006). Pasture, present and future - a brief history of pastoralism in New Zealand. *RMupdate*, 18. Wellington: Ministry of Agriculture and Forestry.

Bryman, A. & Cassell, C. (2006). The researcher interview: a reflexive perspective. *Qualitative Research in Organizations and Management: An International Journal*, 1(1), 41-55.

Budge, G. E., et al. (2015). Evidence for pollinator cost and farming benefits of neonicotinoid seed coatings on oilseed rape. *Scientific Reports*, 5(1).

Burne, J. (2017). *Experience of self-reflexivity in qualitative research*. Retrieved from https://nzareblog.wordpress.com/2017/11/28/self-reflexivity/.

Callon, M., & Latour, B. (1981). Unscrewing the big leviathan: how actors macro-structure reality and how sociologists help them to do so. In K. Knorr-Cetina & A. Cicourel (Eds.), *Advances in social theory and methodology: toward an integration of micro and macro-sociologies* (pp. 277-303). Boston: Routledge & Kegan Paul.

Callon, M., & Law, J. (1982). On Interests and their Transformation: Enrolment and Counter-Enrolment. *Social Studies of Science*, 12(4), 615–625. https://doi.org/10.1177/030631282012004006 Callon, M. (1986). Some elements of a sociology of translation: domestication of the scallops and the fishermen of St Brieuc Bay. In J. Law (Ed.), *Power, action and belief: a new sociology of knowledge* (pp. 196–233). London: Routledge & Kegan Paul.

Callon, M. (1999). Actor-network theory: The market test. *The Sociological Review*, 47(1_suppl), 181-195.

Callon, M. (2012). Society in the making: the study of technology as a tool for sociological analysis. In W. E. Bijker, T. P. Hughes & T. Pinch (Ed.), *Social construction of technological systems: new directions in the sociology and history of technology* (Anniversary ed., pp. 77-97). Cambridge: MIT Press.

Campbell, C.C. (2004), Journalism and public knowledge. Nat Civic Rev, 93, 3-10. 10.1002/ncr.55

Carrington, D. (2017). UN experts denounce 'myth' pesticides are necessary to feed the world. Retrieved from https://www.theguardian.com/environment/2017/mar/07/un-experts-denouncemyth-pesticides-are-necessary-to-feed-the-world.

Carson, R. (1962). Silent spring. London: Folio Society.

Castree, N., & MacMillan, T. (2001). Dissolving dualisms: actor-networks and the reimagination nature. In N. Castree & B. Braun (Eds.), *Social Nature: Theory, Practice and Politics* (pp. 208-224). London: Blackwell.

Chapman, R. B. & Jackson, T. (2010). *A review of insecticide use on pastures and forage crops in New Zealand*. Retrieved from http://agpest.co.nz/wp-content/uploads/2013/06/A-review-of-insecticide-use-on-pastures-and-forage-crops-in-New-Zealand.pdf.

Chapman, M. (2016). *Submission on: Agricultural Compounds and Veterinary Medicines Amendment Bill.* Retrieved from https://www.hortnz.co.nz/assets/Our-Work-files/HortNZ-submission-dataprotection-for-agricultural-compounds-FINAL-28012016.pdf

Chua, W. F. (1995). Experts, networks and inscriptions in the fabrication of accounting images: A story of the representation of three public hospitals. *Accounting, Organizations and Society*, 20(2), 111-145.

Clarke, G. (2018). *Save the Bees - Ban neonics*. Retrieved from https://organicnz.org.nz/media-releases/save-the-bees-ban-neonics/

Cloke, P., Cook, I., Crang, P., Goodwin, M., Painter, J., & Philo, C. (2004). *Practising human geography*. London: Sage Publications.

Collinge, C. (2006). Flat ontology and the deconstruction of scale: a response to Marston, Jones and Woodward. *Transactions of the Institute of British Geographers*, 31(2), 244-251.

Collins, H. (2010). Humans not Instruments. Spontaneous Generations, 4(1), 138-147.

Collins, H. M. (1992). Epistemological Chicken HM Collins and Steven Yearley. In A. Pickering (ed.), *Science as Practice and Culture (pp. 301)*. Chicago: University of Chicago Press.

Comber, A., Fisher, P., & Wadsworth, R. (2003). Actor-network theory: a suitable framework to understand how land cover mapping projects develop? Land Use Policy, 20(4), 299-309.

COSMOS Corporation. (1983). *Case studies and organizational innovation: Strengthening the connection*. Bethesda, MD: Author.

Cordella, A. & Shaikh, M. (2006). *From Epistemology to Ontology: Challenging the Constructed Truth of ANT*. 10.13140/RG.2.1.1546.5367.

Cressey, D. (2017). The bitter battle over the world's most popular insecticides: As regulators consider a ban on neonicotinoids, debate rages over the harm they cause to bees. *Nature: International weekly journal of science*, 551(7679), 156-158.

Cressman, D. (2009). A brief overview of Actor-Network Theory: Punctualization, Heterogeneous Engineering & Translation. Retrieved from https://summit.sfu.ca/item/13593

Creswell, J. W. (2012). *Qualitative inquiry & research design: Choosing amoung five approaches (2nd ed.)*. Thousand Oaks, CA: Sage

Crowe Horwath (2017, June). Agriculture mega-mergers to create new industry juggernauts. Retrieved from https://www.crowehorwath.co.nz/insights/agriculture-mega-mergers-to-create-newindustry-juggernauts/.

D'Souza, F. (1993). Conscience and courage. Index on Censorship, 22(1), 32.

Bould, G. (2005). Conscience Be My Guide: An Anthology of Prison Writings. London: Zed Books

Dalziel, P., et al. (2018). *The New Zealand Food and Fibre Sector: A Situational Analysis*. Client report prepared for the Primary Sector Council. Lincoln University: Agribusiness and Economics Research Unit.

Dankert, R. (2011). *Using Actor-Network Theory (ANT) doing research*. Retrieved July 8, 2019 from https://ritskedankert.nl/using-actor-network-theory-ant-doing-research/.

Deleuze, G. & Guattari, F. (1988). *A thousand plateaus: Capitalism and schizophrenia*. Minneapolis: University of Minnesota Press.

Didovich, N. (2018). *Other retailers not following Bunnings in banning pesticide linked to bee deaths*. Retrieved from https://www.stuff.co.nz/life-style/homed/garden/100594129/more-pesticides-to-be-pulled-from-shelves

Douglas, M. R., et al. (2015). Neonicotinoid insecticide travels through a soil food chain, disrupting biological control of non-target pests and decreasing soya bean yield. *Journal of Applied Ecology*, 52(1), 250-260.

Durepos, G. & Mills, A. (2017). ANTI-History, relationalism and the historic turn in management and organization studies. *Qualitative Research in Organizations and Management: An International Journal*, 12, 53-67. 10.1108/QROM-07-2016-1393.

Easton, B. (2010). *Economic history, Early Māori economies*. Retrieved from https://teara.govt.nz/en/economic-history/page-2

Edwards, S. (2014). Doing Actor-Network Theory: Integrating network analysis with empirical philosophy in the study of research into Genetically Modified Organisms in New Zealand (Doctoral thesis, Lincoln University 2014). Retrieved from

https://researcharchive.lincoln.ac.nz/bitstream/handle/10182/6744/Edwards_PhD_open.pdf?seque nce=4&isAllowed=y

Ellet, W. (2007). *The case study handbook: How to read, discuss, and write persuasively about cases.* Boston: Harvard Business School Press.

Environmental Protection Authority (2018a). *EPA watching weekend EU vote on neonicotinoids science*. Retrieved June 25, 2019 from https://www.epa.govt.nz/news-and-alerts/latest-news/epa-watching-weekend-eu-vote-on-neonicotinoids-science/

Environmental Protection Authority (2018b). *EPA calls for neonicotinoid information*. Retrieved from https://www.epa.govt.nz/news-and-alerts/latest-news/epa-calls-for-neonicotinoid-information/

Environmental Protection Authority (2018c). EPA ramps up chemical reassessments programme. Retrieved from https://www.epa.govt.nz/news-and-alerts/latest-news/epa-ramps-up-chemicalreassessments-programme/. Environmental Protection Authority (2019a). Bees and other Pollinators. Retrieved June 25, 2019 from https://www.epa.govt.nz/everyday-environment/animals-and-insects/bees/?accordion-anchor=308.

Environmental Protection Authority (2019b). Quick facts about the chemical reassessment programme. Retrieved from https://www.epa.govt.nz/industry-areas/hazardous-substances/chemical-reassessment-programme/faqs/.

Eriksson, L. (2004). The presentation of the scientific self. Social Studies of Science, 34(3), 423-426.

Fenwick, T. (2011). Reading Educational Reform with Actor Network Theory: Fluid spaces, otherings, and ambivalences. *Educational Philosophy and Theory*, 43(s1), 114-134.

Fenwick, T. & Edwards, R. (2012). *Researching Education Through Actor-Network Theory*. Hoboken, United Kingdom: John Wiley & Sons, Incorporated.

Fioravanti, C. & Velho, L. (2010). Let's follow the actors! Does Actor-Network Theory have anything to contribute to science journalism?. *Journal of Science Communication*. 9(4), A09. 10.22323/2.09040202.

Fitzpatrick, R. (2019). *Investigation into Taranaki's Rural Waste Stream*. Retrieved from https://www.wasteminz.org.nz/wp-content/uploads/Richard-Fitzpatrick.pdf

Fitzsimmons, M. (2004). Engaging ecologies. In P. Cloke, P. Crang & M. Goodwin (Eds.), *Envisioning human geographies* (pp. 30-47). London: Arnold.

Fleming, A. & Vanclay, F. (2010). Farmer responses to climate change and sustainable agriculture. A review. *Agronomy for Sustainable Development*, 30(1), 11-19.

Food and Agricultural Organization of the United Nations and Republic of Slovenia Ministry of Agriculture Forestry and Food (2018, May). *Why bees matter: The importance of bees and other pollinators for food and agriculture.* Retrieved from http://www.fao.org/3/I9527EN/i9527en.PDF

Foster, R. J. (2012). Landscaping Boulder Bay, Canterbury, New Zealand: the emergent and contested classification of authentic heritage baches and an endangered species of penguin (Doctorate thesis, Lincoln University 2012). Retrieved from

https://researcharchive.lincoln.ac.nz/bitstream/handle/10182/5885/foster_phd1.pdf?sequence=4&i sAllowed=y Fox, S. (2000). Communities Of Practice, Foucault And Actor-Network Theory. *Journal of Management Studies*, 37(6), 853-868.

Frykberg, E. (2018, August). *Ban of controversial pesticides could cost the country \$1.2 billion*. Retrieved from https://www.radionz.co.nz/news/business/364552/ban-of-controversial-pesticidescould-cost-the-country-1-point-2-billion

Gad, C., & Bruun Jensen, C. (2010). On the Consequences of Post-ANT. *Science, Technology, & Human Values*, 35(1), 55-80. https://doi.org/10.1177/0162243908329567

Garvin, D.A. (2003). Making the case: Professional education for the world of practice. *Harvard Magazine*, 106(1), 56-107.

General Assembly of New Zealand (1959). Agricultural Chemicals Act 1959. 51, 517-530.

General Assembly of New Zealand (1979). Pesticides Act 1979. 26, 402-438.

Giera, N., Meister, A., & Buchan, D. (2006). Bridging the gap between environmental knowledge and research, and desired environmental outcomes to achieve sustainable land management. Retrieved from https://www.agriculture.govt.nz/dmsdocument/6931/direct

Godfray, H., et al. (2014). A restatement of the natural science evidence base concerning neonicotinoid insecticides and insect pollinators. *Proc. R. Soc. B-Biol. Sci.*, 281(1786).

González, G. R. (2013). *The use of Actor-Network Theory and a Practice-Based Approach to understand online community participation* (Doctorate thesis, Sheffield University 2013). Retrieved from http://etheses.whiterose.ac.uk/4000/1/Final_final_final.pdf

Gorman, R. (2019). What's in it for the animals? Symbiotically considering 'therapeutic' humananimal relations within spaces and practices of care farming. *Medical Humanities*, 45(3), 313.

Gottschalk, L. (1968). Understanding history: A primer of historical method. New York: Knopf.

Gray, B. J. & Gibson, J. W. (2013). Actor–Networks, Farmer Decisions, and Identity. *Culture, Agriculture, Food and Environment*, 35(2), 82-101.

Greenpeace European Unit (2018, April). *Three neonicotinoids down, more bee-killing pesticides to go*. Retrieved from https://www.greenpeace.org/eu-unit/issues/nature-food/1080/three-neonicotinoidotinoids-down-more-bee-killing-pesticides-to-go/

Guagnano, G. A., et al. (1995). Influences on Attitude-Behavior Relationships: A Natural Experiment with Curbside Recycling. *Environment and Behavior*, 27(5), 699-718.

Guest, G., Namey, E. & Mitchell, M. (2013). In-depth interviews. In Guest, G., Namey, E., & Mitchell, M. *Collecting qualitative data* (pp. 113-171). 55 City Road, London: SAGE Publications, Ltd. 10.4135/9781506374680

Guy, N. (2017). *Nathan Guy: Farming is the backbone of our economy*. Retrieved from https://www.nzherald.co.nz/hawkes-bay-today/opinion/news/article.cfm?c_id=1503459&objectid=11834189

Hacking, I. (1991). Representation in scientific practice. [Book review]. Science, 252(5008), 979+.

Haggerty, J. and Cambell, H. (2008). *Farming and the environment*. Retrieved from https://teara.govt.nz/en/farming-and-the-environment

Hammersley, M., Foster, P. & Gomm, R. (2000). Case Study Method. London: SAGE Publications.

Hancock, F. (2018). *Mapping New Zealand's chemical romance*. Retrieved from https://www.newsroom.co.nz/2018/04/23/106027/mapping-new-zealands-chemical-romance

Harding, A. (2017). Actor-network theory and micro-learning networks. *Education for Primary Care* 28(5), 295-296.

Harding, S. (1992). After the Neutrality Ideal: Science, Politics, and "Strong Objectivity". Social *Research*, 59(3), 567-587.

Harding, S. (1998). Gender, Development, and Post-Enlightenment Philosophies of Science. *Hypatia*, 13(3), 146-167.

Harding, S. (2008). *Sciences from below. Feminisms, postcolonialities, and modernities*. Durham, NC: Duke University Press.

Harman, G. (2014). Bruno Latour: Reassembling the Political. London, United Kingdom: Pluto Press.

Harris, R. (2004). *Handbook of environmental* law (1st ed.). Wellington, N.Z.: Royal Forest and Bird Protection Society of New Zealand.

Hess, D. J. (1997). Science Studies: An Advanced Introduction. New York: New York University Press.

Hinchliffe, S. (2001). Indeterminacy in-decisions-science, policy and politics in the BSE (Bovine Spongiform Encephalopathy) crisis. *Transactions of the Institute of British Geographers*, 26(2), 182-204.

Hobbs, M. (2002). *The AGCARM (New Zealand association for animal health and crop protection) annual conference: Official opening address*. Retrieved from https://www.beehive.govt.nz/node/14547

Holstein, J. A., & Gubrium, J. F. (1995). *Qualitative Research Methods: The active interview*. Thousand Oaks, CA: SAGE Publications, Inc.

Gubrium, J. F., & Holstein, J. A. (2003). *Postmodern interviewing*. Thousand Oaks, CA: SAGE Publications, Inc.

Hutching, G. (2018). *Roundup, 1080 not on EPA's list as it reassess chemicals for possible bans*. Retrieved from https://www.stuff.co.nz/business/farming/107856448/epa-to-reassess-chemicals-for-possible-bans-but-not-glyphosate-or-neonicotinoids.

Ilbery, B. W. (1978). Agricultural decision-making: a behavioural perspective. *Progress in Geography,* 2(3), 448-466.

Ingram, M. (2007). Biology and Beyond: The Science of "Back to Nature" Farming in the United States. *Annals of the Association of American Geographers*, 97(2), 298-312.

Jakku, E. (2003). Murky waters? Science, politics and environmental decision-making in the Brisbane River dredging dispute (Doctorate Thesis, Griffith University 2003). Retrieved from https://trove.nla.gov.au/work/3758007

Jansen, T. (2016). Beyond ANT: Towards an 'infra-language' of reflexivity. *European Journal of Social Theory*, 20(2), 199-215.

Kahan, D. (2013). Market-oriented farming: an overview. Retrieved January 6, 2019 from http://www.fao.org/3/a-i3227e.pdf.

Kingi, T. (2008). Ahuwhenua - Māori land and agriculture - Changes to Māori agriculture. Retrieved from https://teara.govt.nz/en/ahuwhenua-maori-land-and-agriculture/page-1

Klein, A., et al. (2007). Importance of pollinators in changing landscapes for world crops. *Proc. R. Soc. B-Biol. Sci.*, 274(1608), 303-313.

Kleinschmit, J. & Lilliston, B. (2015, August). Unknown benefits, hidden costs: Neonicotinoid seed coatings, crop yields and pollinators. Retrieved from https://www.iatp.org/sites/default/files/2015_08_06_Neonics_BL_JK.pdf

Kothage, J. (2018). The impact of restrictions on neonicotinoidotinoid and fipronil insecticides on pest management in maize, oilseed rape and sunflower in eight European Union regions. *Pest Management Science*, 74(1), 88-99. 10.1002/ps.4715.

Kvale, S. (2006). Dominance through interviews and dialogues. Qualitative Inquiry, 12 (3), 480-500.

Lalani, B., et al. (2016). Smallholder farmers' motivations for using Conservation Agriculture and the roles of yield, labour and soil fertility in decision making. *Agricultural Systems*, 146, 80-90.

Latour, B. (1987). *Science in action: How to follow scientists and engineers through society*. Cambridge: Harvard University Press.

Latour, B. (1988). The Pasteurization of France. Cambridge: Harvard University Press.

Latour, B. (1991). Technology is society made durable. In J. Law (Ed.), *A sociology of monsters: essays on power, technology and domination* (pp. 103-131). London: Routledge.

Latour, B. (1993). We have never been modern. Cambridge: Harvard University Press.

Latour, B. (1996). *On actor-network theory. A few clarifications plus more than a few complications*. Retrieved Feb 2, 2020 from http://www.cours.fse.ulaval.ca/edc65804/latour-clarifications.pdf

Latour, B. (1999a). Pandora's hope: essays on the reality of science. London: Harvard University Press.

Latour, B. (1999b). On recalling ANT. In J. Law & B. Hassard (Ed.), *Actor Network Theory and after* (pp. 15-25). Oxford: Blackwell.

Latour, B. (2003). *Is re-modernization occurring-And if so, how to prove it? A commentary on Ulrich Beck.* Theory, Culture & Society, 20(2), 35-48.

Latour, B. (2005). *Reassembling the social: An introduction to actor-network theory*. Oxford, United Kingdom: Oxford University Press.

Laurier, E., & Philo, C. (1999). X-Morphising: Review Essay of Bruno Latour's Aramis, or the Love of Technology. *Environment and Planning A: Economy and Space*, 31(6), 1047-1071. https://doi.org/10.1068/a311047 Law, J. (1986). *On Power and its Tactics: A View from the Sociology of Science*. The Sociological Review, 34(1), 1-38.

Law, J. (1986). On the methods of long distance control: vesels, navigation and the Portuguese route to India. In J. Law (Ed.), *Power, Action and Belief A New Sociology of Knowledge* (pp. 234-263). London: Routledge and Kegan Paul.

Law, J. (1986). *Power, action and belief: a new Sociology of Knowledge*? London: Routledge & Kegen Paul.

Law, J. (1987). Technology and heterogeneous engineering: the case of Portuguese expansion. In W. Bijker, T. Hughes and T. Pinch (Ed.), *The social construction of technological systems: New directions in the sociology and history of technology* (pp. 111-134). MIT Press.

Law, J. (1992). Notes on the theory of the actor-network: ordering, strategy, and heterogeneity. *Systems Practice*, 5(4), 379-393.

Law, J. (1999). After ANT: complexity, naming and topology. Sociological Review, 47(1_suppl), 1-14.

Law, J. (2002). *Aircraft Stories: Decentering the Object in Technoscience*. Durham, North Carolina: Duke University Press.

Law, J. (2004). After Method: Mess in Social Science Research. London: Routledge.

Law, J. (2008a). Actor-network theory and material semiotics. In B. S. Turner (Ed.), *The New Blackwell Companion to Social Theory* (pp. 141-158). Oxford: Blackwell.

Law, J. (2008b). The materials of STS. Retrieved February 26, 2020 from http://www.heterogeneities.net/papers.htm

Law, J. (2008c). On sociology and STS. Retrieved February 26, 2020 from http://www.heterogeneities.net/papers.htm

Law, J. and J. Urry (2004). Enacting the social. Economy and Society, 33(3), 390-410.

Lawlor, J. & Kavanagh, D. (2015). Infighting and fitting in: Following innovation in the stent actor– network. *Industrial Marketing Management*, 44, 32-41. https://doi.org/10.1016/j.indmarman.2014.10.005. Lester, P. (2017). Common pesticides can harm bees, but the jury is still out on a global ban. Retrieved June, 2019 from https://www.victoria.ac.nz/news/2017/06/common-pesticides-can-harmbees,-but-the-jury-is-still-out-on-a-global-ban

Lewis-Beck, M. S., Bryman, A., & Futing Liao, T. (2004). *The SAGE encyclopedia of social science research methods* (*Vols. 1-0*). Thousand Oaks, CA: Sage Publications, Inc. 10.4135/9781412950589

Liu, G. Z., Spector, J., Merrill, M., Van Merrienboer, J. G., Driscoll, M. & Erlbaum, L. (2010). *Handbook of Research on Educational Communications and Technology, 3rd Edition*. New York: Springer Science+Business Media.

Lorimer, J. (2010). Moving image methodologies for more-than-human geographies. *Cultural geographies*, 17(2), 237-258.

Lorns, C. (2019). *Time to overhaul New Zealand's pesticide rules*. Retrieved June 25, 2019 from https://www.victoria.ac.nz/law/about/news/1725188-time-to-overhaul-new-zealands-pesticide-rules

Lowe, A. (2001). After ANT: An illustrative discussion of the implications for qualitative accounting case research. *Accounting, Auditing & Accountability Journal*, 14, 327-351.

Mabry, L. (2008). Case study in social research. In P. Alasuutari, L. Bickman & J. Brannen (Ed.), *The SAGE handbook of social research methods* (pp. 214-227). London: Sage Publications.

MacIntyre, A., Allison, N. & Penman, D.(1989). Pesticides : issues and options for New Zealand: a discussion document prepared for the Ministry of the Environment. Retrieved from http://agris.fao.org/agris-search/search.do?recordID=NZ19890127019

Magallanes, C. J. I. (2018). Permitting poison: Pesticide regulation in Aotearoa New Zealand. *Environmental and Planning Law Journal*, 35, 456-490.

Maguire, M., & Delahunt, B. (2017). Doing a Thematic Analysis: A Practical, Step-by-Step Guide for Learning and Teaching Scholars. *AISHE-J*, 8(3), 1-10.

Manktelow, D., Stevens, P., Walker, J., Gurnsey, S., Park, N.M., Zabkiewicz, J., Teulon, D. & Rahman, A. (2005). Trends in Pesticide Use in New Zealand: 2004. Retrieved from https://www.dioxinnz.com/Spray-NZ-Hist/PDF/nz-pesticide-trends.pdf

Mann, R. (1972). *2,4,5-T - Just A Weed Killer?* Retrieved from http://nzetc.victoria.ac.nz/tm/scholarly/tei-Salient35041972-t1-body-d11.html Manuka Health (2018). *Neonicotinoidotinoids in New Zealand Agriculture*. Retrieved from https://www.manukahealth.co.nz/en-au/news/neonicotinoids-in-new-zealand-agriculture/

Marby, L. (2008). Case study in social research. In Alasuutari, P., Bickman, L. and Brannen, J. (Ed.), *The SAGE Handbook of Social Research Methods* (pp. 214-227), London: SAGE Publications.

Martin, N. A. & Workman, P. J. (1994). Confirmation of a pesticide-resistant strain of western flower thrips in New Zealand. In A. J. Popay (Ed.), *The Forty Seventh New Zealand Plant Protection Conference, Waitangi Hotel, 9-11 August 1994* (pp. 144-148), Rotorua, New Zealand: New Zealand Plant Protection Society.

McGrath, P. F. (2014). Politics meets science: The case of neonicotinoid insecticides in Europe. *Sapiens*, 7(1), 1-10.

McLintock, A. H. (2019). HISTORICAL EVOLUTION. Retrieved February 22, 2020 from https://teara.govt.nz/en/1966/farming

McManus, G. & Powe, M. (2007). The Entrepreneurial Process. Melbourne, Australia: Marcus Powe.

McNamara, C., et al. (2004). Making and managing organisational knowledge(s). *Management Accounting Research*, 15(1), 53-76.

McSweeny, J. (2017, June). *Neonicotinoids linked to bee harm by two studies*. Retrieved from https://www.rnz.co.nz/news/national/334158/neonicotinoids-linked-to-bee-harm-by-two-studies

Mellet, M. (2012, June). *Response to national beekeepers association technical committee's comments on quickbayt spray fly bait, APP201268*. Retrieved from https://www.epa.govt.nz/assets/FileAPI/hsno-ar/APP201268/APP201268-APP201268-EnR-Appendix-C-Applicant-response-to-submission.pdf

Miller, P. (1997). The multiplying machine. Accounting, Organizations and Society, 22(3), 355-364.

Ministry for Culture and Heritage (2016). *Feeding Britain*. Retrieved June 25, 2019 from https://nzhistory.govt.nz/war/public-service-at-war/feeding-britain

Ministry for Culture and Heritage (2019). "History of New Zealand, 1769-1914." Retrieved April 2, 2019 from https://nzhistory.govt.nz/culture/history-of-new-zealand-1769-1914

Ministry for Primary Industries (2012). *Data protection for Agricultural Compounds*. Retrieved January 21, 2020 from https://www.biosecurity.govt.nz/dmsdocument/3989/direct

Ministry for Primary Industries (2017). *Apiculture: Ministry for primary industries 2017 apiculture monitoring programme.* Retrieved March 15, 2019 from

https://www.mpi.govt.nz/dmsdocument/27678-apiculture-ministry-for-primary-industries-2017-apiculture-monitoring-programme

Ministry for Primary Industries (2019a, July). *Maximum Residue Levels*. Retrieved September 25, 2019 from https://www.mpi.govt.nz/food-safety/food-safety-for-consumers/whats-in-our-food-2/chemicals-and-food/maximum-residue-levels/

Ministry for Primary Industries (2019b). *Our Story*. Retrieved June 25, 2019 from https://www.mpi.govt.nz/about-us/our-story/

Ministrry for Primary Industries (2020). *Listed Beekeepers*. Retrieved February 26, 2020 from https://www.foodsafety.govt.nz/registers-lists/beekeepers/index.htm?setup_file=beekeepers-ssi.setup.cgi&rows_to_return=20000&submit_search=Search

Ministry for the Environment (1996). *Hazardous substances and new organisms act 1996*. Wellington: Ministry for the Environment.

Ministry for the Environment (2002). *Towards a Pesticide Risk Reduction Policy for New Zealand*. Retrieved June 26, 2019 from https://www.mfe.govt.nz/sites/default/files/towards-pesticide-risk-reduction-apr02.pdf

Ministry for the Environment (2011). *Environmental Protection Authority Act 2011*. Wellington: Ministry for the Environment.

Ministry for the Environment (2017). *New Zealand's environmental reporting series: Our fresh water* 2017. Retrieved from http://www.mfe.govt.nz/sites/default/files/media/Environmental%20reporting/our-fresh-water-2017_1.pdf

Ministry for the Environment (2019a). *Description of the agrichemical sector*. Retrieved April, 2019 from http://www.mfe.govt.nz/publications/waste/study-new-zealand-product-stewardship-scheme-agrichemical-containers/3

Ministry for the Environment (2019b). *Pesticides and other agrichemicals*. Retrieved June 25, 2019 from http://www.mfe.govt.nz/more/hazards/hazardous-substances/pesticides-and-other-agrichemicals

Mol, A. (2002). The body multiple: ontology in medical practice. Durham, NC: Duke University Press.

Morton, J. (2018). *Neonicotinoid ban in NZ could leave no alternatives for farmers - experts*. Retrieved from https://www.nzherald.co.nz/nz/news/article.cfm?c_id=1&objectid=12043526

Munir, K. A. & Jones, M. (2004). Discontinuity and After: the Social Dynamics of Technology Evolution and Dominance. *Organization Studies*, 25(4), 561-581.

Murdoch, J. (1997). Inhuman/Nonhuman/Human: Actor-Network Theory and the Prospects for a Nondualistic and Symmetrical Perspective on Nature and Society. *Environment and Planning D: Society and Space*, 15(6), 731–756. https://doi.org/10.1068/d150731

Murdoch, J. (2001). Ecologising Sociology: Actor-Network Theory, Co-construction and the Problem of Human Exemptionalism. *Sociology*, 35(1), 111-133.

Newstrom-Lloyd, L. E. (2013). Pollination in New Zealand. In J. R. Dymond (Ed.), *Ecosystem Services in New Zealand - conditions and trends* (pp. 408-431). Lincoln: Manaaki Whenua Press.

Nhamo, G. (2005). Environmental Policy Processes Surrounding South Africa's Plastic Bags Regulation: Tensions, Debates and Responses in Waste Product Regulation. Boca Raton, Florida: Dissertation.com.

Nightingale, D. J. & Cromby, J. (1999). *Social Constructionist Psychology a Critical Analysis of Theory and Practice*. Buckingham: Open University Press.

O'Connell, B., Ciccotosto, S. & De Lange, P. (2014). Understanding the application of Actor-Network Theory in the process of accounting change. *In Critical Perspectives on Accounting Conference*, Toronto, 7-9 July 2014.

Orlikowski, W. J. (2000). Using Technology and Constituting Structures: A Practice Lens for Studying Technology in Organizations. *Organization Science*, 11(4), 404-428.

Otago Bulletin Board (2013, July). *Otago study shows legacy of pesticides difficult to avoid*. Retrieved May 28, 2019 from https://www.otago.ac.nz/otagobulletin/research/otago051129.html

Palaganas, E. C., Sanchez, M. C., Molintas, M. P., & Caricativo, R. D. (2017). Reflexivity in Qualitative Research: A Journey of Learning. *The Qualitative Report*, *22*(2), 426-438.

Pannell, D.J. (2008). Public benefits, private benefits, and policy mechanism choice for landuse change for environmental benefits. *Land Economics*, 84(2), 225-240.

Patton, M.Q. (2001). *Qualitative research and evaluation methods: Integrating theory and practice.* Saint Paul, MN: SAGE Publications, INC. Peden, R. (2008). *Farming in the economy - Early farming and the great pastoral era*. Retrieved from https://teara.govt.nz/en/farming-in-the-economy/page-1

Pessoa, A. S. G., Harper, E., Santos, I. S., & Gracino, M. C. (2019). Using Reflexive Interviewing to Foster Deep Understanding of Research Participants' Perspectives. *International Journal of Qualitative Methods.* https://doi.org/10.1177/1609406918825026

Pantumsinchai, P. (2018). Armchair detectives and the social construction of falsehoods: an actornetwork approach. *Information, Communication & Society*, 21(5), 761-778, DOI: 10.1080/1369118X.2018.1428654

Phillips, D. (2000). *Evaluation research: An introduction to principles, methods and practice.* Retrieved from http://www.socresonline.org.uk/5/1/clarke.html

Piddock, G. (2018). *Banning neonicotinoidotinoid agrichemicals likely to lead to greater pesticide use*. Retrieved May 20, 2019 from https://www.stuff.co.nz/business/farming/103527211/banningneonicotinoidotinoid-agrichemicals-likely-to-lead-to-greater-pesticide-use

Prior, L. (2008). Documents and action. In P. Alasuutari, L. Bickman & J. Brannen (Ed.), *The SAGE handbook of social research methods* (pp. 479-492). London: Sage Publications.

Randall, W. & Phoenix, C. (2009). The problem with truth in qualitative interviews: reflections from a narrative perspective. *Qualitative Research in Sport and Exercise*, 1, 125-140. 10.1080/19398440902908993.

Reed, M. I. (1997). In Praise of Duality and Dualism: Rethinking Agency and Structure in Organizational Analysis. *Organization Studies*, 18(1), 21-42. *https://doi.org/10.1177/017084069701800103*

Restivo, S. (2010). Bruno Latour. In G. Ritzer and J. Stepnisky (Ed.), *The Wiley-Blackwell Companion to Major Social Theorists* (PP. 520-540). Boston, Blackwell. 10.1002/9781444396621.ch41

Rice, P.L. & Ezzy, D. (1999). *Qualitative research methods: a health focus*. Melbourne, Australia: Oxford University Press.

Ritzer, G. (Ed.) (2005). *Encyclopedia of social theory*. Thousand Oaks, CA: SAGE Publications, Inc. doi: 10.4135/9781412952552

Rogers, E.M. & Shoemaker, F.F. (1971). *Communication of innovations; a cross-cultural approach*. New York: Free Press. Ross, M. (2019). *Farming, Subsidies and Food Safety in New Zealand*. Retrieved June 25, 2019 from http://agcarm.co.nz/wp-content/uploads/IAHJ-subsidies-Mark-Ross.pdf

Russell Stephens, P. (1966). HISTORICAL EVOLUTION. Retrieved Feb 12, 2020 from https://teara.govt.nz/en/1966/farming

Sanchez-Bayo, F. (2014). The trouble with neonicotinoidotinoids. *Science*, 346(6211), 806-807. doi: 10.1126/science.1259159

Sayes, E. (2013). Actor–Network Theory and methodology: Just what does it mean to say that nonhumans have agency? *Social Studies of Science*, 44(1), 134-149. https://doi.org/10.1177/0306312713511867

Schultze, U. & Boland, R. (2000). Knowledge management technology and the reproduction of work practices. *The Journal of Strategic Information Systems*, 9, 193-212. https://doi.org/10.1016/S0963-8687(00)00043-3

Science Media Centre (2015). *Herbicides and antibiotic resistance – Expert reaction*. Retrieved May 28, 2019 from https://www.sciencemediacentre.co.nz/2015/03/27/herbicides-and-antibiotic-resistance-expert-reaction/.

Scottville Farm (2017). *Seed Crops: New Zealand Grown.* Retrieved December 12, 2019 from https://scottvillefarm.co.nz/seed-crops/.

Shapin, S. (1988). Following Scientists Around. Social Studies of Science, 18(3), 533– 550. https://doi.org/10.1177/030631288018003007

Shavelson, R., & Towne, L. T., (Eds.). (2002) Scientific research in education. Washington, DC: National Academies Press.

Simon-Delso, N., Amaral-Rogers, V., Belzunces, L.P. et al. (2015). Systemic insecticides (neonicotinoids and fipronil): trends, uses, mode of action and metabolites. *Environmental Science and Pollution Research*, 22, 5–3. https://doi.org/10.1007/s11356-014-3470-y

Sørensen, E. (2013). THE PSYCHOLOGIST'S SHORTCUT TO ACTOR-NETWORK THEORY. *The American Journal of Psychology*, 126(3), 369-371. doi: 10.5406/amerjpsyc.126.3.0369

Statistics New Zealand. (2018). *Agricultural and horticultural land use*. Retrieved from http://archive.stats.govt.nz/browse_for_stats/environment/environmental-reporting-series/environmental-indicators/Home/Land/land-use.aspx

Steins, N. A. (2001). New Directions in Natural Resource Management The Offer of Actor-Network Theory. *IDS Bulletin.* 32(4), 18-25.

Stern, P. C. (2000). Toward a coherent theory of environmentally significant behavior. *Journal of Social Issues*, 56(3), 407-424.

Strathern, M. (1996). Cutting the Network. *The Journal of the Royal Anthropological Institute*, 5(3), 517-535.

Sudman, S., & Bradburn, N. M. (1982). *Asking questions: A practical guide to questionnaire design.* San Francisco: Jossey-Bass.

Tabak, E. (2015). *Information cosmopolitics: An actor-network theory approach to information practices*. Burlington: Elsevier Science.

Tatnall, A. (1999). Actor-Network Theory and Information Systems Research. In Khosrow-Pour, M., *Encycolpedia of Information Science and Technology*, 1st edition. (pp. 42-46). IGI Global.

Tatnall, A. (2011). *Actor-Network Theory and Technology Innovation: Advancements and New Concepts.* Hershley, New York: Information Science Reference.

The Reserve Bank of New Zealand (2007). *The Reserve Bank and New Zealand's Economic History*. Retrieved June 13, 2019 from https://www.rbnz.govt.nz/-/media/ReserveBank/Files/Publications/Factsheets%20and%20Guides/factsheet-the-reserve-bankand-nzs-economic-history.pdf

The White House Office of the Press Secretary (2014). *Fact sheet: The economic challenge posed by declining pollinator populations*. Retrieved May 28, 2019 from https://obamawhitehouse.archives.gov/the-press-office/2014/06/20/fact-sheet-economic-challenge-posed-declining-pollinator-populations

Thomas, D. (1987). The Case-study Method in Psychology and Related Disciplines. *Disability, Handicap & Society*, 2(2), 205-207.

Tonelli, F., Silva, S. S., Zambalde, A. L., & Brito, M. J. (2010). The Critical Constructivism of the Actor-Network Theory and the Knowledge-based economy of the Triple Helix: theoretical possibilities and practical implications. Tourism Industry Association New Zealand (2019). *Tourism 2025: Growing value together*. Retrieved May 28, 2019 from http://www.tourism2025.org.nz/assets/Documents/TIA-T2025-Summ-Doc-WEB.pdf

Twaddle, S. and Sanderson, K. (2014). *Economic value of neonicotinoid seed treatment to New Zealand*. Retreieved from http://agcarm.co.nz/wp-content/uploads/2019/09/Economic-value-of-neonicotinoid-seed-treatment-to-New-Zealand-180314_Final.pdf

U.S. Government Accountability Office, Program Evaluation and Methodology Division. (1990). *Case study evaluations*. Washington, DC: Government Printing Office.

United States Environmental Protection Agency (2019a). *Pollinator Potential: Schedule for the Review of Neonicotinoid Pesticides*. Retrieved from https://www.epa.gov/pollinator-protection/schedule-review-neonicotinoid-pesticides

United States Environmental Protection Agency (2019b). *Why we use pesticides*. Retrieved from https://www.epa.gov/safepestcontrol/why-we-use-pesticides

Van House, N. A. (2004). Science and technology studies and information studies. *Annual Review of Information Science and Technology*, 38(1), 1-86.

Vanclay, F. (2004) Social principles for agricultural extension to assist in the promotion for natural resource management. *Australian Journal of Experimental Agriculture*, 44, 213-222.

Vegetables New Zealand (2019). *Research Projects*. Retrieved Jan 6, 2019 from https://www.freshvegetables.co.nz/research-and-development/current-research-projects/.

Walsham, G. (1997). Actor-Network Theory and IS Research: Current Status and Future Prospects. In A. S. Lee, J. Liebenau and J. I. DeGross (Ed.), *Information Systems and Qualitative Research: Proceedings of the IFIP TC8 WG 8.2 International Conference on Information Systems and Qualitative Research, Philedelphia 31st May–3rd June 1997* (pp. 466-480). Boston, MA: Springer US.

Walters, K. (2013). Data, data everywhere but we don't know what to think? Neonicotinoid insecticides and pollinators. *Outlooks on Pest management*, 24(4), 151-155.

Waltz, S. B. (2006). Nonhumans Unbound: Actor-Network Theory and the Reconsideration of "Things" in Educational Foundations. *Educational Foundations*, 20(3-4), 51.

Warner, K. D. (2008). Agroecology as participatory science: Emerging alternatives to technology transfer extension practice. *Science, Technology, & Human Values,* 33(6), 754-777.

Warner, K. D. (2016). Rachel's Dream: Agricultural Policy and Science in the Public Interest. In R. Gottlieb and H. R. Luce (Ed.), *Agroecology in Action: Extending Alternative Agriculture Through Social Networks* (pp. 10-34). Cambridges, Massachusetts: MIT Press.

Warren, C. A. B. (2002). Qualitative interviewing. In J. F. Gubrium & J. A. Holstein (Ed.), *Handbook of interview research: context & method* (pp. 83-101). London: Sage Publications.

Waters, W., Thomson, D., & Nettle, R. (2009). Derived attitudinal farmer segments: A method for understanding and working with the diversity of Australian dairy farmers. *Extension Farming Systems* Journal. 5(2), 47-57.

White, S. S., et al. (2009). Planting Food or Fuel: Developing an Interdisciplinary Approach to Understanding the Role of Culture in Farmers' Decisions to Grow Second-Generation, Biofuel Feedstock Crops. *Comparative Technology Transfer and Society*. 7(3), 25.

Whittle, A. and Spicer, A. (2008). Is Actor Network Theory Critique?. *Organization Studies*, 29(4), 611-629.

Wildblood-Crawford, B. (2006). Grassland utopia and Silent Spring: Rereading the agrichemical revolution in New Zealand. New Zealand Geographer, 62(1), 65-72.

Woodcock, B., Bullock, J., Shore, R., Heard, M., Pereira, M., Redhead, J., Ridding, L., Dean, H., Sleep, D., Henrys, P., Peyton, J., Hulmes, S., Hulmes, L., Sárospataki, M., Saure, C., Edwards, M., Genersch, E., Knaebe, S. & Pywell, R. (2017). Country-specific effects of neonicotinoid pesticides on honey bees and wild bees. *Science*, 356(6345), 1394-1395. 10.1126/science.aaa1190.

Woods, M. (1997). Researching rural conflicts: hunting, local politics and actor-networks. *Journal of rural studies*, 14(3), 321-340.

Woolgar, S. (1991). Configuring the user: the case of usability trials. In J. Law (Ed.), A sociology of monsters: essays on power, technology and domination (pp. 57-99). London: Routledge.

Yin, R. K., Bateman, P. G., & Moore, G. B. (1985). *Case Studies and Organizational Innovation: Strengthening the Connection*. Knowledge, 6(3), 249260. https://doi.org/10.1177/107554708500600303

Yin, R. (1994). Case study research: Design and methods (2nd edition). Thousand Oaks: Sage Publications.

Yin, R. (2003). Case study research: Design and methods (3rd edition). Thousand Oaks, Calif.: Sage Publications.

Yin, R. (2009). Case study research: Design and methods (4th edition). Los Angeles, Calif.: Sage Publications.

Yin, R. (2014). Case study research: Design and methods (Fifth edition). Thousand Oaks, Calif.: Sage Publications

Young, D., et al. (2010). An actor-network theory analysis of policy innovation for smoke-free places: understanding change in complex systems. *American journal of public health*, 100(7), 1208.

Appendix

A.1 Interview Schedules

Thomas Scott

A network analysis of how neonicotinoids have become embedded in New Zealand agriculture

List of open-ended questions.

Introductory questions for all participants

- 1. What is your involvement in New Zealand agriculture, tell me your agricultural story?
- 2. What is your relationship with neonicotinoids?
- 3. What do you think about neonicotinoids, how would you describe them?
- 4. Who do you believe are the major proponents and opponents to neonicotinoids in New Zealand and what are they saying about it?
- 5. From who/where do you get the majority of your information regarding neonicotinoids?
- 6. Why are neonicotinoids so important to New Zealand agriculture and what would be the cost if they were not used?
- 7. Are there any individuals, groups or organisations that you affiliate with who are interested in neonicotinoids? How does your relationship with them function?
- 8. What is the government's role in agriculture in particular in regard to the use of chemicals?

Questions for farmers and Growers

- 1. Who performs your crop monitoring?
- 2. How do you determine which treatments are best for protecting your plants?
- 3. What are the procedures you take prior to treating a crop?
- 4. Is the use of agrichemicals a necessary management activity on your property?
- 5. Could you tell me about a time you have been criticized about using neonicotinoids and by who?
- 6. Could you tell me about a time you have been praised/encouraged to use neonicotinoids and by who?
- 7. What knowledge do you have of current regulation about the use of neonicotinoids in New Zealand?
- 8. What is the importance of pollinators to your farm? How do you receive pollination?
- 9. Do you have a relationship with beekeepers? If so please provide some details.
- 10. Do you take any measures to protect or enhance pollinator populations on the farm?
- 11. Do you hold a relationship with a specific retailer of seed and agri-chemicals?
- 12. Are there any alternatives to chemical treatments for the control of pests, in particular, are there any alternatives to neonicotinoids?

Questions for Regulators

- 1. Does your organisation perform crop monitoring for pests on farms? If yes, how often, why and what does this entail? If no, why not and who does it, if not you?
- 2. Does your organisation monitor the use of hazardous substances on the ground? If yes, how. If not, why not and who does this instead?
- 3. Does your organisation enforce the current regulations surrounding the use of neonicotinoids? If yes, how, if not why not and who does this?
- 4. Is there any legislation in place to regulate the use of neonicotinoids as a seed treatment? If yes, can you please explain it further? If no, why not?

5. Who/which groups do you consult when considering whether or not to change regulations surrounding hazardous chemicals? More specifically who/what groups do you consult in regards to neonicotinoids?

Questions for Seed and Chemical Retailers/Representatives

- 1. How close and how important is the relationship you hold with farmers? Can you tell me why it is so important?
- 2. Do you perform crop monitoring and pest control services for farmers? If yes, what is the procedure, if no then who performs this?
- 3. How do you determine whether a crop requires a treatment?
- 4. How do you decide which form of treatment is most suitable on a farm/crop?
- 5. Do you know what systemic seed treatment is?
- 6. Do you sell foliar spray form of neonicotinoids?
- 7. How often do you meet with an individual farmer and what activities/discussions do you perform?
- 8. How much influence/monitoring does the government, or a regulating body have on your activities?
- 9. Are you directed on which treatments/seeds to be selling/marketing to farmers each season? Who is giving this direction?
- 10. Are there any alternatives to chemical treatments to control pests, in particular, are there any alternatives to neonicotinoids?
- 11. Do you fund any research on neonicotinoids or pollinators? If no, why not, if yes who is performing this research and what is it investigating?

Questions for Beekeepers

- 1. Do you perform pollination services for farmers? Can you provide some detail about your relationship with those farmers?
- 2. Have you any agreements/procedures in place with farmers to protect your bees from exposure to the chemicals they apply?
- 3. Have you ever experienced hive deaths associated with insecticide application on the farm? I yes, can you outline the sequence of events that led to and followed this event?
- 4. Are there any measures in place to protect beekeepers from malpractice on farms?
- 5. Who are you able to turn to if your beehives die from pesticide exposure?

A.2 Research Information Sheet

Lincoln University

Faculty of Environment Society and Design

Research Information Sheet

I would like to invite you to participate in the project entitled "A network analysis of how neonicotinoids have become embedded in New Zealand agricultural practices."

This project is a thesis required for the degree: Master of Natural Resources Management and Ecological Engineering. The aim is to explore how and why certain agrichemicals are used over others. You have been asked to participate in this study due to your connection to the agricultural sector and expertise in either the use, regulation, research or effects of agrichemicals.

Your participation will involve the completion of an informal interview with myself, this is estimated to take up to an hour to complete. The interview will occur at a time and place chosen by you.

The results of this research will be assessed as a part of a master's thesis. They may also be subject to publication in an academic journal and presentation at a conference.

Your participation in this research is voluntary and you may decline to answer any question. You may also withdraw from this project, up to 4 weeks following the agreed upon interview date, by contacting me (Thomas Scott). If you choose to withdraw from the research all the data that you have provided shall be destroyed.

You may choose to keep your identity anonymous and data confidential.

If you have any concerns or questions about your participation or the project itself, please contact me as I am happy to discuss this further.

Researcher: Thomas Scott, Student tom.scott@lincolnuni.ac.nz Ph. +64 272644176

Supervisor: Sarah Edwards sarah.edwards@lincoln.ac.nz

Supervisor: Suzanne Vallance suzanne.Vallance@lincoln.ac.nz

Please note that this research has been reviewed and approved by the Lincoln University Human Ethics Committee.

A.3 Lincoln University Human Ethics Committee Approval

Research Management Office

T 64 3 423 0817 PO Box 85084, Lincoln University Lincoln 7647, Christchurch New Zealand www.lincoln.ac.nz

9 July 2019
Application No: 2019-35
Title: A network analysis of how neonicotinoids have become embedded in New Zealand's agricultural practices.

Applicant: T Scott

The Lincoln University Human Ethics Committee has reviewed the above noted application. Thank you for your response to the questions which were forwarded to you on the Committee's behalf.

I am satisfied on the Committee's behalf that the issues of concern have been satisfactorily addressed. I am pleased to give final approval to your project.

Please note that this approval is valid for three years from today's date at which time you will need to reapply for renewal.

Once your field work has finished can you please advise the Human Ethics Secretary, Alison Hind, and confirm that you have complied with the terms of the ethical approval.

May I, on behalf of the Committee, wish you success in your research.

Yours sincerely

Grant Tavinor Chair, Human Ethics Committee

PLEASE NOTE: The Human Ethics Committee has an audit process in place for applications. Please see 7.3 of the Human Ethics Committee Operating Procedures (ACHE) in the Lincoln University Policies and Procedures Manual for more information.

A.4 Email to participants

Thomas Scott

A network analysis of how neonicotinoids have become embedded in New Zealand agricultural practices.

Dear [Name of recruited participant],

My name is Thomas Scott, I am a postgraduate student at Lincoln University in the faculty of environment-society and design. We spoke on the *[date of the recruitment phone call]* regarding your participation in my research project titled: A network analysis of how neonicotinoids have become embedded in New Zealand agricultural practices.

This email is to again confirm your participation and willingness to be interviewed by me as a part of this research. Attached is a research information sheet that further outlines the project, your role as a participant and the procedures I have taken to protect your identity and minimise risk. Please read through this prior to the interview.

If you are willing to be interviewed could you please reply to this email detailing a date, place and time which you are able to meet for a little over an hour (the interview itself is expected to take 1 hour).

I look forward to hearing from you soon.

Sincerely,

Thomas Scott.

A.5 Consent Form

Lincoln University

Consent Form

A network analysis of how neonicotinoids have become embedded in New Zealand agricultural practices.

I have read and understood the description of the above-named project. On this basis, I agree to participate in the project, and I consent to the publication of the results of the project with the understanding that anonymity will be preserved. I also understand that I may at any time withdraw from the project, including withdrawal of any information I have provided during the interview up until 4 weeks following the agreed upon interview date.

I agree that my participation in this research is voluntary.

I consent to have an audio recording made of my interview.

I consent to notes being made during my interview.

I consent to being identified by my name and profession (please specify)

I consent to being identified by my profession only (please specify)

Name:

Signed:

Date: